UNITED STATES SECURITIES AND EXCHANGE COMMISSION

WASHINGTON, D.C. 20549

FORM 8-K

CURRENT REPORT PURSUANT TO SECTION 13 OR 15(d) OF THE SECURITIES EXCHANGE ACT OF 1934 Date of Report (Date of earliest event reported) February 13, 2025

flyExclusive, Inc.

(Exact name of registrant as specified in its charter)

Delaware001-4044486-1740840(State or other jurisdiction of incorporation)(Commission file Number)(IRS Employer Identification No.)

2860 Jetport Road, Kinston, NC (Address of principal executive offices)

28504 (Zip Code)

252-208-7715 Registrant's telephone number, including area code

Not Applicable (Former name or former address, if changed since last report.)

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Class A Common Stock	FLYX	NYSE American LLC
Redeemable warrants, each whole warrant exercisable for one share of Class A Common Stock at an exercise price of \$11.50 per share	FLYX WS	NYSE American LLC

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

Emerging growth company ⊠

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act. \Box

Item 1.01 Entry into a Material Agreement.

On February 13, 2025, flyExclusive, Inc., a Delaware corporation ("flyExclusive"), entered into an Agreement and Plan of Merger and Reorganization, as may be amended from time to time (the "Merger Agreement"), by and among flyExclusive, FlyX Merger Sub, Inc., a Delaware corporation and wholly owned subsidiary of flyExclusive ("Merger Sub"), Jet.AI Inc., a Delaware corporation ("Jet.AI") and Jet.AI SpinCo, Inc., a Delaware corporation, and a wholly owned subsidiary of Jet.AI ("SpinCo"), pursuant to which (i) as a condition to closing the transaction, Jet.AI will distribute all of the shares of SpinCo, on a pro rata basis, to the stockholders of Jet.AI (the "Distribution") and (ii) the Merger Sub will merge with and into SpinCo (the "Merger" and, together with the Distribution and all other transactions contemplated under the Merger Agreement, the "Transactions") with SpinCo surviving the Merger as a wholly owned subsidiary of flyExclusive.

The Transactions are expected to allow each company to focus on its core business, while providing flyExclusive with increased size and additional cash. The material terms of the Merger Agreement are summarized below. Capitalized terms not used herein and not otherwise defined herein shall have the meaning as set forth in the Merger Agreement.

The Merger

The Merger will become effective at the time the Certificate of Merger has been duly filed with the Secretary of State of Delaware or such other date and time as is agreed upon by Jet.AI and flyExclusive and specified in the Certificate of Merger in accordance with the General Corporation Law of the State of Delaware ("DGCL") (such date, the "Closing Date", and such time, the "Effective Time"). At the Effective Time, the holders of common stock of SpinCo, \$0.001 par value per share ("SpinCo common stock"), will have their SpinCo common stock converted on a pro rata basis into the right to receive shares (the "Merger Shares") of Class A common stock of flyExclusive, \$0.0001 par value per share ("flyExclusive Class A common stock"), based on an assumed exchange ratio which is the quotient of 4,119,403 (the "Merger Consideration Shares") divided by the total outstanding shares of SpinCo common stock (the "Exchange Ratio"), subject to certain adjustments as described below. The assumed Merger Consideration Shares was calculated assuming, among other things, that SpinCo's net cash at closing will be \$12.0 million and that the volume weighted average closing sale price of flyExclusive's Class A common stock for the 30 trading days ending 3 days prior to the Closing Date is \$3.35 per share, which was the closing sale price on February 13, 2025. At the Effective Time, Jet.AI's stockholders will continue to own and hold their then existing shares of Jet.AI common stock.

Conditions to Closing

General Conditions

Consummation of the Merger is conditioned on the approval of the Transactions by an affirmative vote of the number of holders of Jet.AI Common Stock required to approve such Transactions under applicable Law and the Governing Documents of the Company.

In addition, the consummation of the Merger is conditioned upon, among other things, the following mutual conditions to closing, among other things:

- all Requisite Regulatory Approvals shall have been obtained and shall remain in full force and effect and all statutory waiting periods in respect thereof shall have expired:
- the Separation and Distribution and the other transactions contemplated by the Separation and Distribution Agreement shall have been consummated in accordance with the Separation and Distribution Agreement;
- the registration statement on Form S-4 to effect the registration of the Merger Consideration Shares shall have become effective;
- the applicable notice periods required by applicable stock exchange rules or securities laws in connection with the Distribution, if any, shall have expired;
- · each of the Company Shareholder Approval and SpinCo Shareholder Approval shall have been obtained;
- no governmental authority of competent jurisdiction shall have enacted, issued or granted any Law (whether temporary, preliminary or permanent), in each case that is in effect and which has the effect of restraining, enjoining or prohibiting the consummation of the Transaction; and
- the shares of flyExclusive Class A common stock issuable pursuant to the transactions contemplated by the Merger Agreement shall have been approved for listing on NYSE American.

Jet.AI and SpinCo Conditions to Closing

The obligations of Jet.AI and SpinCo to consummate or cause to be consummated the Transactions are subject to, among other things, the satisfaction of the following additional conditions:

- each of flyExclusive and Merger Sub shall have each performed and complied in all material respects with the obligations, covenants and agreements required by the Merger Agreement to be performed or complied with by it at or prior to the Effective Time;
- · the accuracy of all representations and warranties made by flyExclusive and Merger Sub (subject to customary bring-down standards);
- · no flyExclusive Material Adverse Effect shall have occurred between the date of the Merger Agreement and the Closing; and
- flyExclusive shall have delivered to Jet.AI a certificate dated as of the Closing signed by an authorized officer of flyExclusive certifying that (i) the conditions relating to the bring-down of flyExclusive's and Merger Sub's respective representations and warranties and compliance by flyExclusive and Merger Sub with their respective covenants have been satisfied and (ii) no flyExclusive Material Adverse Effect shall have occurred between the date of the Merger Agreement and the Closing Date that is continuing.

flyExclusive and Merger Sub Conditions to Closing

The obligations of flyExclusive and Merger Sub to consummate or cause to be consummated the Transactions is subject to the satisfaction of the following additional conditions, among other things:

- Jet.AI and SpinCo shall each have performed and complied in all material respects with the obligations, covenants and agreements required by the Merger Agreement to be performed or complied with by it at or prior to the Effective Time;
- the accuracy of all representations and warranties made by Jet.AI and SpinCo (subject to customary bring-down standards);
- no SpinCo Material Adverse Effect or Jet.AI Material Adverse Effect shall have occurred between the date of the Merger Agreement and the Closing;
- as of the Closing Date and after payment in full of any Unpaid Transaction Expenses, SpinCo must have Net Cash of at least \$12,000,000;
- each of the Support Agreement delivered concurrently with the execution and delivery of the Merger Agreement shall be in full force and effect as of the Closing and shall not have been rescinded or otherwise terminated;
- · each officer and director of SpinCo shall resign, effective as of the Effective Time, from each of his or her positions as on officer or director of SpinCo;
- the board of directors of flyExclusive must not have made a determination that the Transactions are not in the best interests of the stockholders of flyExclusive; and
- Jet.AI shall have delivered to flyExclusive a certificate dated as of the Closing signed by an authorized officer of Jet.AI certifying that (i)the Separation and Distribution shall have been consummated, (ii) the conditions relating to the bring-down of Jet.AI's and SpinCo's respective representations and warranties and compliance by Jet.AI and SpinCo with their respective covenants have been satisfied, (iii) no Jet.AI Material Adverse Effect or SpinCo Material Adverse Effect shall have occurred between the date of the Merger Agreement and the Closing Date that is continuing and (iv) as of the Closing Date and after payment in full of any Unpaid Transaction Expenses, SpinCo will have Net Cash of at least \$12,000,000.

Covenants

Covenants of flyExclusive and Merger Sub

flyExclusive and Merger Sub made certain covenants under the Merger Agreement. Prior to the Effective Time, flyExclusive will, and will cause its subsidiaries (including Merger Sub) to:

- conduct its operations in the ordinary course of business in all material respects;
- to comply with, and continue performing under, as applicable, the governing documents of flyExclusive;
- not seek any approval from the flyExclusive Shareholders to amend, modify, restate, waive, rescind or otherwise change the governing documents of flyExclusive or Merger Sub;
- not make or declare any dividend or distribution to the flyExclusive Shareholders or make any other distributions in respect of any of flyExclusive's equity interests or Merger Sub capital stock, share capital or equity interests, split, combine, reclassify or otherwise amend any terms of any shares or series of flyExclusive's equity interests or Merger Sub capital stock or equity interests; and
- not enter into any agreement to do any action prohibited under Section 7.1 of the Merger Agreement.

Conduct of Business of Jet.AI and SpinCo

Jet.AI and SpinCo made certain covenants under the Merger Agreement, with respect to SpinCo or the SpinCo Business. Prior to the Effective Time, Jet.AI and SpinCo will:

• conduct the SpinCo Business in the ordinary course of business in all material respects;

- use commercially reasonable efforts to manage the SpinCo Business' working capital and maintain the books and records related to the SpinCo Business consistent with
 past practice;
- use commercially reasonable efforts to maintain their respective relations and goodwill with all material suppliers, material customers and other material commercial counterparties and governmental authorities (in each case, as related to the SpinCo Business);
- not amend, modify, restate, waive, rescind or otherwise change the governing documents of SpinCo, other than an amendment to the certificate of incorporation of SpinCo to increase the number of authorized or outstanding shares of SpinCo Common Stock in connection with the Distribution in accordance with the Merger Agreement and the Transaction Documents;
- other than as required for the Distribution, (A) not declare, set aside or pay any dividends on or make other distributions in respect of any of the Interests of SpinCo (whether in cash, securities or property), except for the declaration and payment of cash dividends or distributions paid on or with respect to a class of Interests of SpinCo that is wholly owned directly or indirectly by SpinCo, (B) not split, combine, subdivide, reduce, or reclassify any of the Interests of SpinCo or issue or authorize or propose the issuance of any other securities in respect of, in lieu of, or in substitution for, Interests of SpinCo and (C) not redeem, repurchase or otherwise acquire, or permit any subsidiary to redeem, repurchase or otherwise acquire, any Interests (including any securities convertible or exchangeable into such Interests) (except for the acquisition by Jet.AI or any of its subsidiaries of any shares of capital stock, membership interests or other equity interests of Jet.AI or its Subsidiaries in connection with the forfeiture of any Company Equity Awards, in each case, in accordance with their respective terms as in effect as of the date of the Merger Agreement);
- other than as contemplated by the Distribution, not issue, sell, pledge, dispose of, grant, transfer or encumber, any shares of capital stock of SpinCo or securities convertible into, or exchangeable or exercisable for, any shares of such capital stock in SpinCo, or any options, warrants, stock units, or other rights of any kind to acquire any shares of capital stock or other equity interests or such convertible or exchangeable securities, or any other ownership interest (including any such interest represented by contract right), or any "phantom" stock, "phantom" stock rights, stock appreciation rights or stock-based performance rights, in each case, of SpinCo:
- represented by contract right), or any "phantom" stock, "phantom" stock rights, stock appreciation rights or stock-based performance rights, in each case, of SpinCo; not sell, assign, transfer, convey, lease, license, abandon, mortgage, pledge or permit any lien or encumbrance on (other than certain permitted liens and encumbrances) or otherwise dispose of any SpinCo assets (excluding certain intellectual property);
- not purchase, sell, license, sublicense, lease, pledge, covenant not to assert, assign, transfer, abandon, cancel, let lapse or expire, or otherwise dispose, transfer or grant any other rights in or with respect to any intellectual property owned by SpinCo or intellectual property licensed to SpinCo (other than with respect to (A) immaterial or obsolete intellectual property owned by SpinCo or (B) the grant of non-exclusive licenses of intellectual property owned by SpinCo in the ordinary course of business consistent with past practice) or disclose any material Trade Secrets of the SpinCo Business to any other Person (other than in the ordinary course of business consistent with past practice to a Person bound by adequate and enforceable use restrictions and confidentiality obligations with respect to such Trade Secrets);
- not receive, collect, use, store, process, share, safeguard, secure (technically, physically or administratively), dispose of, destroy, disclose, or transfer (including cross-border) any personal information (or fail to do any of the foregoing, as applicable) in material violation of any Privacy Requirements;
- not merge, combine or consolidate (pursuant to a plan of merger or otherwise) SpinCo with any Person or adopt a plan of complete or partial liquidation or resolutions providing for a complete or partial liquidation, dissolution, restructuring, recapitalization or other reorganization of SpinCo;
- not acquire (including by merger, consolidation, or acquisition of shares or assets), lease or license, (i) any interest in any Person or (ii) any assets of any Person that would be an asset of SpinCo at the Effective Time, other than, in the case of clause (ii), (A) in the ordinary course of business with respect to assets having a value not exceeding \$10,000, individually, or \$50,000, in the aggregate, the purchase price for which will be paid by the Company or any of its Subsidiaries prior to the Distribution Date:
- not repurchase, repay, prepay, refinance or incur any indebtedness for borrowed money, issue any debt securities, engage in any securitization transactions or similar arrangements or assume, guarantee or endorse, or otherwise as an accommodation become responsible for (whether directly, contingently or otherwise), the obligations of any Person for borrowed money;
- not make any material loans to, material capital contributions or material investments in, or advances of money to, in each case, in excess of \$10,000 individually or \$20,000 in the aggregate, any Person, in each case except for (A) advances to employees or officers of SpinCo for expenses incurred in the ordinary course of the SpinCo Business consistent with past practice or (B) extended payment terms for customers in the ordinary course of the SpinCo Business consistent with past practice;
- not (A) amend or modify in any material respect, terminate (excluding any expiration in accordance with its terms), or waive any material right, benefit or remedy under, any material SpinCo contract or (B) enter into any contract that if entered into prior to the date of the Merger Agreement would be required to be listed on Section 5.6 or Section 5.14(a) of the SpinCo Disclosure Schedule;
- not (A) adopt or enter into any SpinCo Benefit Plan, (B) grant or provide any change-in-control, severance, termination, retention or similar payments or benefits to any employee or service provider of SpinCo, or that would result in any material

- liability to SpinCo, (C) hire or engage, or make an offer to hire or engage, any officer, employee, service provider or individual independent contractor of SpinCo, or (D) terminate (without cause) the employment of any SpinCo employee or engagement of any SpinCo independent contractor;
- except as required by GAAP, not make any material change to any financial accounting principles, methods or practices of SpinCo or with respect to the SpinCo Business;
- · not waive, release, settle, compromise or otherwise resolve any action, litigation or other proceedings;
- not (A) make, change or revoke any material tax election in respect of the SpinCo Business that would bind SpinCo for periods following the Effective Time (other than
 an election reasonably required in connection with the Separation) or (B) settle or compromise any material tax liability for which SpinCo would be responsible under
 any Transaction Document:
- not make or commit to make any capital expenditures, on an annualized basis, in the aggregate, in excess of \$20,000;
- not disclose or agree to disclose to any Person (other than flyExclusive or any of its representative) any material trade secret or any other material confidential or proprietary information, know-how or process of Jet.AI or any of its Subsidiaries, in each case other than in the ordinary course of business consistent with past practice and pursuant to customary contractual obligations or fiduciary or professional duty to maintain the confidentiality thereof;
- not terminate without replacement, or fail to use reasonable best efforts to maintain, any license or permit that is material to the conduct of the business of Jet.AI and its Subsidiaries, taken as a whole:
- not waive the restrictive covenant obligations of any current officer of Jet.AI or SpinCo;
- not (A) limit the right of SpinCo to engage in any line of business or in any geographic area, to develop, market or sell products or services, or to compete with any
 Person or (B) grant any exclusive or similar rights to any Person, in each case, except where such limitation or grant does not, and would not be reasonably likely to,
 individually or in the aggregate, materially and adversely affect, or materially disrupt, the ordinary course operation of the businesses of Jet.AI or SpinCo, taken as a
 whole:
- not terminate without replacement or amend in a manner materially detrimental to Jet.AI and its Subsidiaries, taken as a whole, any insurance policy insuring the business of the Company or SpinCo; and
- · not authorize or enter into any contract to do any of the foregoing or otherwise agree or make any commitment to do any of the foregoing.

Representations and Warranties

The Merger Agreement contains customary representations and warranties by flyExclusive, Merger Sub, Jet.AI and SpinCo. The representations and warranties of the respective parties to the Merger Agreement will not survive the Closing.

Termination

The Merger Agreement may be terminated and the Transactions abandoned, but not later than the Closing, as follows:

- by mutual written agreement of flyExclusive and Jet.AI;
- by flyExclusive or Jet.AI, if the Closing shall not have occurred on or prior to June 30, 2025 (the "Outside Date"); provided, that this right to terminate the Merger Agreement shall not be available to any party whose action or failure to comply with its obligations under the Merger Agreement or any of the other Transaction Documents has been the primary cause of, or has primarily resulted in, the failure of the Closing to occur on or prior to such date:
- by flyExclusive or Jet.AI, if any law shall have been promulgated, entered, enforced, enacted or issued and in effect or shall have been deemed to be applicable to the Transactions, including the Distribution, by any governmental authority of competent jurisdiction which permanently prohibits, restrains or makes illegal the consummation of the Transactions, and such law shall have become final and non-appealable; provided, that this right to terminate the Merger Agreement shall not be available to any party whose action or failure to perform any of its obligations under the Merger Agreement or any of the Transaction Documents is the primary cause of, or primarily resulted in, the enactment or issuance of any such law;
- by flyExclusive upon written notice to Jet.AI, in the event of a breach of any representation, warranty, covenant or agreement on the part of Jet.AI or SpinCo, such that the related bring-down conditions would not be satisfied at the Closing, and which, (i) with respect to any such breach that is capable of being cured, is not cured by Jet.AI or SpinCo by the earlier of: (A) fifteen (15) days after receipt of written notice thereof; or (y) the Outside Date, or (ii) is incapable of being cured prior to the Outside Date; provided, that flyExclusive shall not have this right to terminate the Merger Agreement if flyExclusive or Merger Sub is then in breach of any of its representations, warranties, covenants or agreements set forth in the Merger Agreement to the extent such breach or breaches would give rise to the failure of a condition to Closing;
- by Jet.AI upon written notice to flyExclusive, in the event of a breach of any representation, warranty, covenant or agreement on the part of flyExclusive or Merger Sub, such that the related bring-down conditions would not be satisfied at the Closing, and which, (i) with respect to any such breach that is capable of being cured, is not cured by flyExclusive by the earlier of: (A) fifteen (15) days after receipt of written notice thereof; or (y) the Outside Date, or (ii) is incapable of being cured prior to the Outside Date; provided, that Jet.AI shall not have this right to terminate the Merger Agreement if Jet.AI or SpinCo is then

- in breach of any of its representations, warranties, covenants or agreements set forth in the Merger Agreement to the extent such breach or breaches would give rise to the failure of a condition to Closing;
- by flyExclusive or Jet.AI, if the Company Shareholder Approval shall not have been obtained; and by Jet. AI prior to the Registration Statementbeing declared effective by the SEC, in order to accept a Superior Proposal and enter into, immediately following such termination, a binding and definitive written contract with respect to such Superior Proposal, provided, that (i) Jet.AI has complied with its covenants and agreements under Section 7.10 of the Merger Agreement, (ii) Jet.AI pays an amount to flyExclusive equal to the amount of all reasonable and documented out-of-pocket fees and expenses incurred or paid by flyExclusive and its Affiliates in connection with the Merger Agreement and the Transactions, including fees and expenses of law firms, accounting firms, financial advisors, outside experts, third-party services providers and consultants (the "Parent Expenses Reimbursement Amount"), and (iii) Jet.AI pays a termination fee of \$650,000 (the "Termination Fee") to flyExclusive in accordance with the Merger Agreement.

Certain Related Agreements

Separation and Distribution Agreement

In connection with the execution of the Merger Agreement, flyExclusive, Jet.AI and SpinCo entered into a Separation and Distribution Agreement (the "Separation Agreement") pursuant to which, among other things, (i) Jet.AI will undertake to transfer all of the SpinCo Assets (as defined in the Separation Agreement) to SpinCo and (ii) immediately prior to the Merger and after the Separation (as defined in the Separation Agreement), Jet.AI will distribute 100% of the outstanding shares of SpinCo common stock to Jet.AI's stockholders in the Distribution.

Consummation of the Distribution is subject to a number of conditions, including, among others, (i) the completion of the Contribution (as defined in the Separation Agreement), (ii) execution of the Conveyancing and Assumption Instruments (as defined in the Separation Agreement) and (iii) satisfaction (or, where applicable, waiver) of certain condition in the Merger Agreement to Jet.AI's obligations to consummate the Merger.

Support Agreement

Concurrently with the execution of the Merger Agreement, each officer and director and certain other significant stockholders of Jet.AI, who collectively hold approximately 1.8% of the shares eligible to vote at the Meeting (the "Specified Stockholders"), in their respective capacities as stockholders of Jet.AI, entered into a Support Agreement with flyExclusive, pursuant to which each Specified Stockholder has agreed to, among other things, vote its shares of Jet.AI stock in favor of the adoption of the Merger Agreement not vote those shares for any alternative proposal or any other transaction, proposal, agreement, or action made in opposition to adoption of the Merger Agreement or in competition or inconsistent with the Merger, and not transfer those shares (subject to certain exceptions) to third parties. The Support Agreements provide that they will terminate upon the termination of the Merger Agreement in accordance with its terms.

The foregoing descriptions of the Merger Agreement, the Separation Agreement, the Support Agreement and the transactions contemplated thereby do not purport to be complete and are qualified in their entirety by reference to the Merger Agreement and the Separation Agreement, which are filed as Exhibit 2.1 and 10.1 respectively, and the form of Support Agreement which is filed as Exhibit 10.2, to this Current Report on Form 8-K and are incorporated herein by reference.

Item 7.01 Regulation FD Disclosure.

On February 14, 2025, flyExclusive and Jet.AI issued a joint press release announcing the execution of the Merger Agreement, a copy of which is attached hereto as announcing the execution of the Merger Agreement, a copy of which is attached hereto as announcing the execution of the Merger Agreement, a copy of which is attached hereto as announcing the execution of the Merger Agreement, a copy of which is attached hereto as announcing the execution of the Merger Agreement, a copy of which is attached hereto as announcing the execution of the Merger Agreement, a copy of which is attached hereto as announcing the execution of the Merger Agreement, a copy of which is attached hereto as announcing the execution of the Merger Agreement, a copy of which is attached hereto as announcing the execution of the Merger Agreement, a copy of which is attached hereto as announcing the execution of the Merger Agreement, a copy of which is attached hereto as a copy of the execution of the Merger Agreement, a copy of which is attached hereto as a copy of the execution of the Merger Agreement, and a copy of the execution of the Merger Agreement and a copy of the execution of the Merger Agreement, and a copy of the execution of the Merger Agreement and a copy of the execution of the Merger Agreement and a copy of the execution of the execution of the Merger Agreement and a copy of the execution of

The information in this Item 7.01, including Exhibit 99.1, is furnished and shall not be deemed "filed" for purposes of Section 18 of the Securities Exchange Act of 1934, as amended (the "Exchange Act"), or otherwise subject to liabilities under that section, and shall not be deemed to be incorporated by reference into the filings of flyExclusive under the Securities Act of 1933, as amended, or the Exchange Act, regardless of any general incorporation language in such filings. This Current Report on Form 8-K will not be deemed an admission as to the materiality of any information contained in this Item 7.01, including Exhibit 99.1.

Additional Information and Where to Find It

In connection with the proposed Transactions, flyExclusive has filed relevant materials with the SEC, including a registration statement on Form S-4, which will include a proxy statement/prospectus. After the registration statement is declared effective by the

SEC, the definitive proxy statement/prospectus and other relevant documents will be mailed to the stockholders of Jet.AI as of the record date established for voting on the proposed Transactions and will contain important information about the proposed Transactions and related matters. Stockholders of Jet.AI and other interested persons are advised to read, when available, these materials (including any amendments or supplements thereto) and any other relevant documents in connection with Jet.AI's solicitation of proxies for the meeting of stockholders to be held to approve, among other things, the proposed Transactions because they will contain important information about flyExclusive, Merger Sub, Jet.AI, SpinCo and the proposed Transactions. Stockholders will also be able to obtain copies of the preliminary proxy statement/prospectus, the definitive proxy statement/prospectus and other relevant materials in connection with the transaction without charge, once available, at the SEC's website at www.sec.gov.

Participants in Solicitation

Jet.AI and its respective directors and executive officers may be deemed participants in the solicitation of proxies from Jet.AI's shareholders in connection with the proposed Transactions. Jet.AI's shareholders and other interested persons may obtain, without charge, more detailed information regarding the directors and officers of Jet.AI as reflected in the annual report on Form 10-K for the period ended December 31, 2023, filed with the SEC on April 1, 2024, and amended on April 29, 2024 and August 15, 2024. Information regarding the persons who may, under SEC rules, be deemed participants in the solicitation of proxies to Jet.AI's shareholders in connection with the proposed Transactions will be set forth in the proxy statement/prospectus for the proposed Transactions when available. Additional information regarding the interests of participants in the solicitation of proxies in connection with the proposed Transactions will be included in the proxy statement/prospectus that flyExclusive intends to file with the SEC. You may obtain free copies of these documents as described in the preceding paragraph.

flyExclusive, Merger Sub and their respective directors and executive officers may also be deemed to be participants in the solicitation of proxies from the shareholders of Jet.AI in connection with the proposed Transactions. A list of the names of such directors and executive officers and information regarding their interests in the proposed Transactions will be included in the proxy statement/prospectus for the proposed Transactions when available.

No Solicitation or Offer

This communication shall neither constitute an offer to sell nor the solicitation of an offer to buy any securities, or the solicitation of any proxy, vote, consent or approval in any jurisdiction in connection with the Transactions, nor shall there be any sale of securities in any jurisdiction in which the offer, solicitation or sale would be unlawful prior to any registration or qualification under the securities laws of any such jurisdictions. This communication is restricted by law; it is not intended for distribution to, or use by any person in, any jurisdiction where such distribution or use would be contrary to local law or regulation.

Forward-Looking Statements Legend

This communication contains forward-looking statements. The words "anticipate," "believe," "continue," "could," "estimate," "expect," "intend," "may," "might," "plan," "possible," "potential," "predict," "project," "should," "would" and similar expressions may identify forward-looking statements, but the absence of these words does not mean that a statement is not forward-looking. All statements other than statements of historical facts contained in this communication, including statements regarding the expected timing and structure of the Transactions, the ability of the parties to complete the Transactions, the expected benefits of the Transactions, the tax consequences of the Transactions, and flyExclusive's future results of operations and financial position, business strategy and its expectations regarding the benefits of the Transactions. These forward-looking statements are not guarantees of future performance, conditions or results, and involve a number of known and unknown risks, uncertainties, assumptions and other important factors, many of which are outside the control of flyExclusive and Jet.AI, that could cause actual results or outcomes to differ materially from those discussed in the forward-looking statements. Important factors, among others, that may affect actual results or outcomes include, but are not limited to: the risk that the proposed Transactions may not be completed in a timely manner or at all, which may adversely affect the price of flyExclusive's or Jet.AI's securities; the risk that Jet.AI stockholder approval of the Transactions is not obtained; the inability to recognize the anticipated benefits of the Transactions; the occurrence of any event, change or other circumstance that could give rise to the termination of the Merger Agreement; changes in general economic conditions; the outcome of litigation related to or arising out of the Transactions, or any adverse developments therein or delays or costs resulting therefrom; the effect of the announcement or pendency of the Transa

The foregoing list of factors is not exhaustive. You should carefully consider the foregoing factors and the other risks and uncertainties described in the "Risk Factors" section of flyExclusive's Annual Report on Form 10-K filed with the SEC on May 1, 2024, the "Risk

Factors" section of Jet.AI's Annual Report on Form 10-K filed with the SEC on April 1, 2024 (as amended on April 29 and August 15, 2024), the registration statement on Form S-4, the proxy statement/prospectus and certain other documents filed or that may be filed by flyExclusive or Jet.AI from time to time with the SEC following the date hereof. These filings identify and address other important risks and uncertainties that could cause actual events and results to differ materially from those contained in the forward-looking statements. Forward-looking statements speak only as of the date they are made. Readers are cautioned not to put undue reliance on forward-looking statements, and flyExclusive and Jet.AI assume no obligation and do not intend to update or revise these forward-looking statements, whether as a result of new information, future events, or otherwise.

Neither flyExclusive nor Jet.AI gives any assurance that flyExclusive or Jet.AI will achieve their expectations.

Item 9.01. Financial Statement and Exhibits.

(d) Exhibits.

Exhibit No.	Document
2.1	Agreement and Plan of Merger and Reorganization, dated February 13, 2025, by and among by and among flyExclusive, Inc., FlyX Merger Sub, Inc., Jet.AI Inc., and Jet.AI SpinCo, Inc.
10.1	Separation and Distribution Agreement, dated February 13, 2025, by and among flyExclusive, Inc., Jet.AI Inc., and Jet.AI SpinCo, Inc.
10.2	Form of Support Agreement
99.1	Press Release, Dated February 14, 2025
104	Cover Page Interactive Data File (embedded within the Inline XBRL document)

SIGNATURE

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

Dated: February 18, 2025

FLYEXCLUSIVE, INC.

/s/ Thomas James Segrave, <u>Jr.</u>

Name: Thomas James Segrave, Jr. Chief Executive Officer and Chairman Title:

AGREEMENT AND PLAN OF MERGER AND REORGANIZATION DATED AS OF FEBRUARY 13, 2025

by and among

FLYEXCLUSIVE, INC.,

FLYX MERGER SUB, INC.,

JET.AI INC.

and

JET.AI SPINCO, INC.

1.1 1.2	
Article II -	THE MERGER 7
2.1	
<u>2.2</u>	
2.3	
2.4	
	Corporation 9
<u>2.5</u>	Plan of Reorganization 10
Article III	- CONVERSION OF SHARES 10
<u>3.1</u>	Effect on Capital Stock 10
<u>3.2</u>	
3.3	· · · · · · · · · · · · · · · · · · ·
<u>3.4</u>	Appraisal Rights 16
Article IV	- REPRESENTATIONS AND WARRANTIES OF THE COMPANY RELATING TO THE COMPANY 16
4.1	
4.2	
<u>4.3</u>	Governmental Consents 17
<u>4.4</u>	No Conflict 17
4.5	
<u>4.6</u>	
<u>4.7</u>	* *
4.8	
<u>4.9</u>	Tax Matters 19
Article V	REPRESENTATIONS AND WARRANTIES OF THE COMPANY RELATING TO SPINCO 20
<u>5.1</u>	Organization of SpinCo 20
<u>5.2</u>	Due Authorization 20
<u>5.3</u>	Capitalization of SpinCo 21
<u>5.4</u>	
<u>5.5</u>	
<u>5.6</u>	
<u>5.7</u>	<u> </u>
<u>5.8</u>	
<u>5.9</u>	
<u>5.1</u>	
<u>5.1</u>	
<u>5.1</u>	2 <u>Tax Matters</u> 23

i

	5.13	Absence of Changes 25
	5.14	Material Contracts 25
	5.15	Compliance with Law; Permits 27
	5.16	Intellectual Property 27
	5.17	Environmental Matters 29
	5.18	Affiliate Matters 30
	5.19	Brokers' Fees 30
	5.20	Proxy Statement; Registration Statements 30
	5.21	Board and Shareholder Approval 30
	<u>5.22</u>	Parent Capital Stock 31
	5.23	Compliance with Aviation Laws 31
	<u>5.24</u>	Data Privacy 31
	<u>5.25</u>	Anti-Bribery, Anti-Corruption and Anti-Money Laundering 32
	5.26	Sanctions, Import, and Export Controls 33
	<u>5.27</u>	No Other Representations and Warranties 33
A 1	T.77 1	DEDDEGENITATIONS AND WARDANIES OF BARENT AND MEDGED OUR.
		REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB 33
	<u>6.1</u>	Organization of Parent and Merger Sub 34
	6.2	Due Authorization 34
	6.3	Capital Stock and Other Matters 34
	6.4	Governmental Consents 35
	<u>6.5</u>	No Conflict 35
	<u>6.6</u>	<u>Litigation and Proceedings</u> 35
	<u>6.7</u>	Brokers' Fees 36
	<u>6.8</u>	SEC Filings 36
	<u>6.9</u>	No Other Representations and Warranties 36
Article	VII -	COVENANTS 37
	7.1	Conduct of Business by Parent and Merger Sub Pending the Merger 37
	7.2	Conduct of SpinCo Business Pending the Merger 37
	7.3	Tax Matters 40
	7.4	Preparation of the Registration Statement 42
	7.5	Company Shareholders Meeting 44
	7.6	Reasonable Best Efforts 44
	7.7	Access to Information 44
	7.8	Public Announcements 45
	7.9	Defense of Litigation 45
	7.10	No Solicitation 46
	7.11	Section 16 Matters 48
	7.12	Control of Other Party's Business 48
	7.13	SpinCo Share Issuance 48
	7.14	Transaction Documents 48
	7.15	NYSE American Listing 48
	7.16	Takeover Statutes 49

7.17	Obligations of Merger Sub and SpinCo 49
7.18 7.19	Further Assurances 49 Financial Statements 50
7.19 7.20	D&O Indemnification and Insurance 50
7.20 7.21	
$\frac{7.21}{7.22}$	
<u>7.22</u>	Company I manoing 01
Article VIII	- CONDITIONS TO THE MERGER 51
<u>8.1</u>	Conditions to the Obligations of SpinCo, the Company, Parent and Merger Sub to Effect the Merger 51
<u>8.2</u>	Additional Conditions to the Obligations of the Company and SpinCo 52
<u>8.3</u>	Additional Conditions to the Obligations of Parent and Merger Sub 53
Article IX -	INDEMNIFICATION 54
9.1	Survival of Representations, Warranties and Covenants 54
<u>9.2</u>	<u>Indemnification by Company</u> 55
<u>9.3</u>	<u>Claims for Indemnification</u> 55
<u>9.4</u>	<u>Limitations on Indemnification</u> 57
<u>9.5</u>	Materiality 57
Article X - 7	CERMINATION 57
<u>10.1</u>	Termination 57
<u>10.2</u>	Effect of Termination 59
<u>10.3</u>	<u>Termination Fee</u> 59
<u>10.4</u>	Fees and Expenses 60
Article XI -	MISCELLANEOUS 60
<u>11.1</u>	Governing Law; Jurisdiction 60
<u>11.2</u>	Notices 60
<u>11.3</u>	Headings 61
<u>11.4</u>	Entire Agreement 62
<u>11.5</u>	Amendments and Waivers 62
<u>11.6</u>	Assignment; Parties in Interest; Non-Parties 62
<u>11.7</u>	Specific Performance 62
<u>11.8</u>	WAIVER OF JURY TRIAL 63
<u>11.9</u>	Severability 64
<u>11.1</u> 0	
11.1	<u>Disclosure Schedules</u> 64

EXHIBITS

Exhibit A Separation and Distribution Agreement

AGREEMENT AND PLAN OF MERGER AND REORGANIZATION

This AGREEMENT AND PLAN OF MERGER AND REORGANIZATION, dated as of February 13, 2025, is entered into by and among flyExclusive, Inc., a Delaware corporation ("Parent"), FlyX Merger Sub, Inc., a Delaware corporation and a wholly owned Subsidiary of Parent ("Merger Sub"), Jet.AI Inc., a Delaware corporation (the "Company"), and Jet.AI SpinCo, Inc., a Delaware corporation and, as of the date of this Agreement, wholly owned Subsidiary of the Company ("SpinCo"). Each of the foregoing parties is referred to herein as a "Party" and collectively as the "Parties." Terms with initial capitalized letters used herein (including in the immediately preceding sentence) and not otherwise defined herein have the meanings set forth in Section 1.1 hereof.

RECITALS

- A. As of the date of this Agreement, SpinCo is a wholly owned, direct Subsidiary of the Company.
- B. Contemporaneously with the execution of this Agreement, the Company, SpinCo and Parent are entering into the Separation and Distribution Agreement, pursuant to which the Company will transfer the SpinCo Business to SpinCo (the "Separation").
- C. Upon the terms and subject to the conditions set forth in the Separation and Distribution Agreement the Company will distribute all of the shares of SpinCo Common Stock to the Company stockholders without consideration on a *pro rata* basis as set forth in the Separation and Distribution Agreement (the "*Distribution*"). As of the Distribution Date, SpinCo will be the sole owner and operator of the SpinCo Business.
- D. Contingent on the completion of the Distribution, the Parties to this Agreement intend to effect a merger of Merger Sub, a wholly owned subsidiary of Parent, with and into the SpinCo (the "*Merger*") in accordance with this Agreement and the Delaware General Corporation Law, as amended (the "*DGCL*"). Upon consummation of the Merger, Merger Sub will cease to exist, and SpinCo will continue as the surviving corporation and as a wholly owned subsidiary of Parent.
- E. Promptly following the consummation of the Merger, Parent intends to cause the Surviving Corporation to convert to a limited liability company. At least one day following the conversion of the Surviving Corporation to a limited liability company, Parent intends to contribute all of the equity of the Surviving Corporation to LGM Enterprises, LLC in exchange for the issuance of a number of Common Units of LGM Enterprises, LLC (as defined in the Amended and Restated Operating Agreement of LGM Enterprises, LLC, dated December 27, 2023 (the "LGM Operating Agreement")) equal to the Merger Consideration Shares such that the number of Common Units of LGM Enterprises, LLC held by Parent shall maintain a one-to-one ratio to the number of shares of issued and outstanding Parent Common Stock pursuant to Section 4.12 of the LGM Operating Agreement.

- F. Concurrently with the execution and delivery of this Agreements, and as a condition and inducement to Parent's willingness to enter into this Agreement, the Persons identified on <u>Section 5.3(c)</u> of the SpinCo Disclosure Schedule (the "Specified Stockholders") are executing and delivering to Parent that certain Stockholder Support Agreement, dated as of even date herewith (the "Stockholder Support Agreements"), by and among the Specified Stockholders, Parent and Merger Sub.
- G. The board of directors of the Company (the "Company Board") has unanimously: (a) determined that this Agreement and the transactions contemplated hereby, including the Distribution and the Merger, upon the terms and subject to the conditions set forth in the Separation and Distribution Agreement and herein, are fair to, and in the best interests of, the Company and the Company's stockholders; (b) approved and declared advisable this Agreement, including the execution, delivery, and performance thereof, and the consummation of the transactions contemplated by this Agreement, including the Separation and Distribution and the Merger, upon the terms and subject to the conditions set forth in the Separation and Distribution Agreement and herein; (c) directed that the Separation and Distribution Agreement and this Agreement be submitted to a vote of the Company's stockholders for adoption; and (d) resolved to recommend that Company stockholders vote in favor of adoption of this Agreement in accordance with the DGCL.
- H. The respective boards of directors of Parent and Merger Sub have each unanimously: (a) determined that it is in the best interests of Parent or Merger Sub, as applicable, and their respective stockholders, and declared it advisable, to enter into this Agreement; and (b) approved the execution, delivery, and performance of this Agreement and the consummation of the transactions contemplated hereby, including the Merger, in each case, in accordance with the DGCL.
- I. The Parties desire to make certain representations, warranties, covenants, and agreements in connection with the Merger and the other transactions contemplated by this Agreement and also to prescribe certain terms and conditions to the Merger.

AGREEMENT

Therefore, the Parties agree as follows:

Article I

DEFINITIONS

- 1.1 <u>Definitions</u>. Unless otherwise provided herein, terms with initial capital letters used in this Agreement will have the meanings ascribed to such terms as set forth in in Annex A attached hereto, which is incorporated herein and made a part hereof.
 - 1.2 Interpretation.
 - (a) Unless the context of this Agreement otherwise requires:

- (i) (a) words of any gender include each other gender and neuter form; (b) words using the singular or plural number also include the plural or singular number, respectively; (c) derivative forms of defined terms will have correlative meanings; (d) the terms "hereof," "herein," "hereby," "hereto," "herewith," "hereunder" and derivative or similar words refer to this entire Agreement; (e) the terms "Article," "Section," "Annex," "Exhibit," "Schedule," and "Disclosure Schedule" refer to the specified Article, Section, Annex, Exhibit, Schedule or Disclosure Schedule of this Agreement and references to "paragraphs" or "clauses" shall be to separate paragraphs or clauses of the Section or subsection in which the reference occurs; (f) the words "include," "includes" and "including" shall be deemed to be followed by the phrase "without limitation"; and (g) the word "or" shall be disjunctive but not exclusive;
- (ii) any Law defined or referred to in this Agreement or in any agreement or instrument that is referred to herein means such Law as from time to time amended, modified or supplemented, including (in the case of statutes) by succession of comparable successor Laws and the related regulations thereunder and published interpretations thereof, and references to any Contract or instrument are to that Contract or instrument as from time to time amended, modified or supplemented; <u>provided</u>, that for purposes of any representations and warranties contained in this Agreement that are made as of a specific date or dates, references to any Law shall be deemed to refer to such Law, as amended, and the related regulations thereunder and published interpretations thereof, in each case, as of such date or dates.
- (iii) references to any federal, state, local, or foreign statute or Law shall include all regulations promulgated thereunder; and
- (iv) references to any Person include references to such Person's successors and permitted assigns, and in the case of any Governmental Authority, to any Person succeeding to its functions and capacities.
- (b) The language used in this Agreement shall be deemed to be the language chosen by the Parties to express their mutual intent. The Parties acknowledge that each Party and its attorney have reviewed and participated in the drafting of this Agreement and that any rule of construction to the effect that any ambiguities are to be resolved against the drafting Party, or any similar rule operating against the drafter of an agreement, shall not be applicable to the construction or interpretation of this Agreement.
- (c) Nothing herein (including the SpinCo Disclosure Schedule, Company Disclosure Schedule and the Parent Disclosure Schedule) shall be deemed an admission by any Party or any of its Affiliates, in any Action, that such Party or any such Affiliate, or any third party, is or is not in breach or violation of, or in default in, the performance or observance of any term or provisions of any Contract or any Law.
- (d) Whenever this Agreement refers to a number of days, such number shall refer to calendar days unless Business Days are specified. If any action is to be taken or given on or by a particular calendar day, and such calendar day is not a Business Day, then such action may be deferred until the next Business Day.

- (e) When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded and if the last day of such period is not a Business Day, the period shall end on the next succeeding Business Day.
- (f) The phrase "to the extent" shall mean the degree to which a subject or other thing extends, and such phrase shall not mean simply "if."
- (g) The term "writing," "written" and comparable terms refer to printing, typing and other means of reproducing words (including electronic media) in a visible form.
- (h) All accounting terms used herein and not expressly defined herein shall have the meanings given to them under GAAP, unless the context otherwise requires.
 - (i) All monetary figures shall be in United States dollars unless otherwise specified.
- (j) No reference in this Agreement to dollar amount thresholds shall be deemed to be evidence of a SpinCo Material Adverse Effect, Company Material Adverse Effect or Parent Material Adverse Effect, as applicable, or materiality.
- (k) The phrases "furnished," "provided," "delivered" or "made available" when used with respect to information or documents means that such information or documents have been (i) physically or electronically delivered to the relevant Party (and includes such information or documents that have been furnished to its Representatives acting on its behalf or posted to the SpinCo Datasite) or (ii) are otherwise Parent SEC Reports or Company SEC Reports and made publicly available on the SEC's EDGAR website by Parent or the Company, as applicable, in each case, not later than forty-eight hours prior to the execution of this Agreement.

Article II

THE MERGER

2.1 The Merger. At the Effective Time and upon the terms and subject to the conditions of this Agreement, Merger Sub shall be merged with and into SpinCo in accordance with the applicable provisions of the DGCL, the separate existence of Merger Sub shall cease and SpinCo shall continue as the surviving corporation of the Merger (sometimes referred to herein as the "Surviving Corporation") and shall succeed to and assume all the rights, powers and privileges and be subject to all of the obligations of Merger Sub in accordance with the DGCL. As a result of the Merger, SpinCo shall become a direct, wholly owned Subsidiary of Parent. References herein to "SpinCo" with respect to the period from and after the Effective Time shall be deemed to be references to the Surviving Corporation. At the Effective Time, the effects of the Merger shall be as provided in this Agreement, the Certificate of Merger and the applicable provisions of the DGCL.

2.2 <u>Closing</u>. Unless the Transactions have been abandoned and this Agreement terminated pursuant to <u>Section 10.1</u>, the closing of the Merger (the "*Closing*") shall take place at 10:00 a.m., Eastern time, on the third Business Day after the conditions set forth in <u>Article VIII</u> (other than those, including the Distribution, that are to be satisfied at or immediately prior to the Closing, but subject to the satisfaction or, to the extent permitted by applicable Law, waiver of such conditions at the Closing) have been satisfied or, to the extent permitted by applicable Law, waived, by electronic exchange of documents and signatures, unless another date, time or place is agreed to in writing by the Company and Parent. The date on which the Closing actually occurs is hereinafter referred to as the "*Closing Date*."

2.3 Closing Deliverables; Effective Time.

- (a) At the Closing, the Company or SpinCo, as applicable, will deliver or cause to be delivered:
- (i) to Parent, a certificate signed by an authorized officer of the Company, dated as of the Closing Date, certifying that, to the knowledge and belief of such authorized officer, the conditions specified in Section 8.3(a), (b), (c), and (e) have been satisfied;
- (ii) to Parent, a certificate on behalf of SpinCo, prepared in a manner consistent and in accordance with the requirements of Treasury Regulations Sections 1.897-2(g), 1.897-2(h) and 1.1445-2(c)(3), certifying that no interest in the Company is, or has been during the relevant period specified in Section 897(c)(1)(A)(ii) of the Code, a "U.S. real property interest" within the meaning of Section 897(c) of the Code, and a form of notice to the IRS prepared in accordance with the provisions of Treasury Regulations Section 1.897-2(h)(2), together with written authorization for Parent to deliver such documentation to the IRS on behalf of SpinCo after the Closing;
- (iii) to Parent, copies of resolutions and actions taken by the Company's and SpinCo's board of directors and stockholders in connection with the approval of this Agreement and the Transactions;
- (iv) to Parent, all other documents, instruments or certificates required to be delivered by the Company at or prior to the Closing pursuant to Section 8.3; and
 - (v) each of the consents set forth on <u>Section 2.3(a)(v)</u> of the Company Disclosure Schedule.
 - (b) At the Closing, Parent will deliver or cause to be delivered:
- (i) to the Exchange Agent, the shares of Parent Common Stock to be paid in respect of shares of SpinCo Common Stock in accordance with Section 3.1(a), for further distribution to such holders pursuant to Section 3.2;

- (ii) to the Company and SpinCo, a certificate signed by an authorized officer of Parent, dated the Closing Date, certifying that, to the knowledge and belief of such authorized officer, the conditions specified in <u>Section 8.2(a)</u>, (b) and (d) have been satisfied;
- (iii) to the Company and SpinCo, copies of resolutions and actions taken by Parent's and Merger Sub's board of directors and Merger Sub's stockholder in connection with the approval of this Agreement and the Transactions; and
- (iv) to the Company and Spinco, all other documents, instruments or certificates required to be delivered by Parent at or prior to the Closing pursuant to <u>Section 8.2</u>.
- (c) At least three (3), but no more than ten (10), Business Days prior to the Closing, the Company and SpinCo will prepare and deliver to Parent a statement (the "Estimated Net Cash Statement") that contains an estimated balance sheet of SpinCo as of 11:59 p.m. Eastern Time on the Closing Date (the "Measurement Time") and that sets forth in reasonable detail and accompanied by reasonably detailed backup documentation the Company's and SpinCo's good-faith estimate of Net Cash as of the Measurement Time ("Estimated Net Cash"). The Estimated Net Cash Statement and the components thereof will be prepared from the Company's and SpinCo's books and records and calculated in accordance with the terms of this Agreement. The Company and SpinCo shall provide Parent and its Representatives with reasonable access to such books and records as may be reasonably requested by Parent to review and verify the information contained in the Estimated Net Cash Statement. The Parent and its Representatives shall be entitled to comment on the Estimated net Cash Statement and the Company will make any comments or suggested modifications to the Estimated Net Cash Statement suggested by the Parent or its Representatives. Notwithstanding any input Parent may have with respect to the amounts reflected in such Estimated Net Cash Statement, no position, comment or agreement made or taken by Parent with respect thereto shall preclude or prevent or otherwise limit Parent from taking any other position or making any other argument or pursuing any right or remedy with respect to the Closing Net Cash Statement or any other representation, warranty, covenant, agreement or obligation set forth herein.
- (d) On the Closing Date and following the Effective Time, Parent shall pay or cause to be paid by wire transfer of immediately available funds, all unpaid SpinCo Transaction Expenses as set forth on a written statement to be delivered to Parent not less than two (2) Business Days prior to the Closing Date ("Unpaid Transaction Expenses"), which shall include the respective amounts and wire transfer instructions for the payment thereof, together with corresponding invoices for the foregoing. Notwithstanding anything in this Agreement to the contrary, Parent may use all or any portion of the Cash of SpinCo at the Closing to pay any portion of the Unpaid Transaction Expenses.
- (e) On the Closing Date, SpinCo and Merger Sub shall file a certificate of merger relating to the Merger (the "Certificate of Merger") with the Secretary of State of the State of Delaware in accordance with the relevant provisions of the DGCL and shall make all other filings or recordings required under the DGCL. The Merger shall become effective at the time the Certificate of Merger has been duly filed with the Secretary of State of the State of

Delaware, or such later time as Parent and SpinCo may agree and specify in the Certificate of Merger (such time as the Merger becomes effective being the "Effective Time").

- 2.4 Certificate of Incorporation and Bylaws of the Surviving Corporation; Directors and Officers of the Surviving Corporation
- (a) At the Effective Time, the Certificate of Incorporation of the Surviving Corporation will be amended in its entirety to read as set forth in the Certificate of Merger until thereafter amended as provided therein or by applicable Law.
- (b) At the Effective Time, the bylaws of the Surviving Corporation will be amended in their entirety to read as the bylaws of Merger Sub, until thereafter amended as provided by applicable Law, the Certificate of Incorporation of the Surviving Corporation and such bylaws.
- (c) From and after the Effective Time, until successors are duly elected or appointed and qualified in accordance with applicable Law, (i) the directors of Merger Sub as of immediately prior to the Effective Time shall be the directors of the Surviving Corporation and (ii) the officers of Merger Sub as of immediately prior to the Effective Time shall be the officers of the Surviving Corporation.
- 2.5 <u>Plan of Reorganization</u>. The Parties hereby adopt this Agreement as a "plan of reorganization" withing the meaning of Sections 1.368-2(g) and 1.368-3(a) of the Treasury Regulations.

Article III

CONVERSION OF SHARES

- 3.1 <u>Effect on Capital Stock</u>. At the Effective Time, by virtue of the Merger and without any action on the part of any party to this Agreement or any holder of the capital stock of the Company, SpinCo, Merger Sub or Parent:
 - (a) SpinCo Common Stock and Merger Sub Common Stock.
- (i) Each share of SpinCo Common Stock issued and outstanding as of immediately prior to the Effective Time (other than shares canceled in accordance with <u>Section 3.1(a)(ii)</u>) shall be cancelled and extinguished and automatically converted into the right to receive a number of fully paid and non-assessable shares of Parent Common Stock equal to the Exchange Ratio, with cash paid in lieu of fractional shares of Parent Common Stock in accordance with <u>Section 3.2(h)</u> (the "*Per Share Merger Consideration*"), subject to adjustment in accordance with <u>Section 3.1(a)(iv)</u> and the purchase price adjustment provisions set forth in <u>Section 3.3</u>.
- (ii) Each share of SpinCo Common Stock held by SpinCo as treasury stock or by Parent or Merger Sub, in each case, as of immediately prior to the Effective Time

shall be canceled automatically and shall cease to exist, and no stock or other consideration shall be issued or delivered in exchange therefor or in respect thereof.

- (iii) Each share of SpinCo Common Stock issued and outstanding as of immediately prior to the Effective Time, when converted in accordance with this <u>Section 3.1</u>, shall no longer be outstanding and shall automatically be canceled and shall cease to exist, and each holder of such shares shall cease to have any rights with respect thereto, except the right to receive the Per Share Merger Consideration as provided in <u>Section 3.1(a)(i)</u> and any dividends or distributions and other amounts payable in accordance with <u>Section 3.2(g)</u>.
- (iv) The Exchange Ratio shall be adjusted to the extent appropriate to reflect the effect of any stock split, split-up, reverse stock split, stock dividend or distribution of Parent Common Stock, or securities convertible into any such securities, reorganization, recapitalization, reclassification or other like change (any such change, an "*Equity Adjustment*") with respect to Parent Common Stock having a record date occurring on or after the date of this Agreement and prior to the Effective Time; <u>provided</u>, that nothing in this <u>Section 3.1(a)(iv)</u> shall be construed to permit Parent to take or to permit any of its Subsidiaries to take any action with respect to its securities that is prohibited by the terms of this Agreement.
- (v) At the Effective Time, all of the shares of common stock, par value \$0.0001 per share, of Merger Sub ("Merger Sub Common Stock") issued and outstanding immediately prior to the Effective Time shall be automatically converted into one fully paid and nonassessable share of common stock, par value \$0.0001 per share, of the Surviving Corporation.
- (b) Parent Securities. Each share of Parent Common Stock that is issued and outstanding immediately prior to and at the Effective Time shall remain outstanding immediately following the Effective Time.

3.2 Surrender and Payment.

- (a) Pursuant to Section 3.2 of the Separation and Distribution Agreement, the Exchange Agent shall hold, for the account of the relevant SpinCo stockholders, book-entry shares representing all of the outstanding shares of SpinCo Common Stock distributed in the Distribution.
- (b) Prior to the Effective Time, Parent shall enter into an exchange agreement in customary form (the "Exchange Agent Agreement") with the transfer agent of Parent to act as agent (the "Exchange Agent") for the payment of the Merger Consideration Shares. Substantially concurrently with the Effective Time, Parent shall deposit, or shall cause to be deposited, with the Exchange Agent, for the benefit of the SpinCo Holders, for exchange in accordance with this Section 3.2(b) and Section 3.3 promptly after the Effective Time, bookentry shares representing the Merger Consideration Shares issuable to the SpinCo Holders pursuant to Section 3.1(a)(i) (such bookentry shares of Parent Stock, together with any cash received by the Exchange Agent in respect of dividends or distributions with respect thereto pursuant to Section 3.2(g), as applicable, the "Exchange Fund"). Following the Effective Time,

the Exchange Agent shall deliver, pursuant to instructions from Parent (the "*Parent Instructions*"), the Merger Consideration Shares to be issued pursuant to this <u>Article III</u> out of the Exchange Fund. The Exchange Fund shall not be used for any purpose other than as specified in this Section 3.2(b).

- (c) As promptly as practicable after the Effective Time, Parent shall cause the Exchange Agent to deliver to each SpinCo Holder, from the Exchange Fund, the shares of Parent Common Stock issuable in respect of such shares of SpinCo Common Stock pursuant to the Merger. Each SpinCo Holder shall be entitled to receive in respect of such shares of SpinCo Common Stock held by such Person a bookentry authorization representing the number of whole shares of Parent Common Stock that such holder has the right to receive pursuant to Section 3.1(a)(i) (and any dividends or distributions and other amounts pursuant to Section 3.2(g)).
- (d) Each SpinCo Holder shall be entitled to receive the Per Share Merger Consideration in respect of the SpinCo Common Stock upon receipt of an "agent's message" by the Exchange Agent (or such other evidence, if any, of transfer as the Exchange Agent may reasonably request). Until so transferred each such share of SpinCo Common Stock shall represent after the Effective Time for all purposes only the right to receive the Per Share Merger Consideration payable in respect thereof. Upon payment of the Per Share Merger Consideration pursuant to the provisions of this Section 3.2, each share of SpinCo Common Stock so transferred shall immediately be cancelled.
- (e) The Parent Instructions will instruct the Exchange Agent to designate as Restricted Shares a number of shares of each SpinCo Holder's aggregate portion of the Merger Consideration Shares equal to the Restricted Portion, which shares will include a restrictive legend on its book-entry position describing the Transfer Restriction. For so long as the Restricted Shares remain subject to the forfeiture conditions specified in Section 3.3, the holder thereof will be entitled to (i) exercise the voting rights of such Restricted Shares and (ii) receive any dividends or other distributions payable in respect of such Restricted Shares. Holders of the Restricted Shares will not be entitled to transfer such shares for so long as such shares are subject to forfeiture pursuant to Section 3.3. The book-entry positions of each Restricted Share will include a restrictive legend that such shares may not be transferred until such time as the forfeiture conditions with respect to such shares lapse (the "Transfer Restriction"). Any attempted transfer in violation of the Transfer Restriction will be void *ab initio*, and the Exchange Agent will not process any transfers of the Restricted Shares (other than the forfeiture of such shares pursuant to Section 3.3) for so long as such shares remain Restricted Shares.
- (f) For each SpinCo Holder, the Exchange Agent will include in the book-entry position a restrictive legend as set forth below in respect of the Restricted Shares. The Exchange Agent shall not be entitled to vote or exercise any rights of ownership with respect to Parent Common Stock held by it from time to time hereunder or under the Exchange Agent Agreement.

THE SECURITIES REPRESENTED BY THIS BOOK-ENTRY POSITION ARE SUBJECT FORFEITURE PURSUANT TO AN AGREEMENT THAT ALSO

CONTAINS RESTRICTIONS ON TRANSFER. A COPY OF THE AGREEMENT WILL BE FURNISHED TO ANY INTERESTED PARTY BY THE COMPANY UPON WRITTEN REQUEST, WITHOUT CHARGE."

- (g) Distributions After the Effective Time. Subject to the following sentence, no dividends or other distributions declared after the Effective Time with respect to Parent Common Stock shall be paid with respect to any shares of Parent Common Stock that are not able to be delivered by the Exchange Agent promptly after the Effective Time, whether due to a legal impediment to such delivery or otherwise. Subject to the effect of abandoned property, escheat, Tax or other applicable Laws, following the delivery of any such previously undelivered shares of Parent Common Stock, there shall be paid to the record holder of such shares of Parent Common Stock, without interest, at the time of delivery, to the extent not previously paid, the amount of dividends or other distributions with a record date after the Effective Time theretofore paid with respect to such whole shares of Parent Common Stock.
- (h) No Fractional Shares. No certificates or scrip representing fractional shares of Parent Common Stock or book-entry credit of the same shall be issued on conversion of SpinCo Common Stock, and such fractional share interests will not entitle the owner thereof to vote, or to any other rights of a stockholder of Parent. All fractional shares of Parent Common Stock that a holder of shares of SpinCo Common Stock would otherwise be entitled to receive as a result of the Merger shall be aggregated by the Exchange Agent. The Exchange Agent shall cause the whole shares obtained thereby to be sold on behalf of such holders that would otherwise have been entitled to receive a fractional share of Parent Common Stock pursuant to the Merger in the open market (or otherwise as reasonably directed by Parent), in each case at then-prevailing market prices as soon as reasonably practicable and in no case later than ten (10) Business Days after the Effective Time. The Exchange Agent shall make available the net proceeds thereof, subject to the deduction of the amount of any withholding Taxes as contemplated in Section 3.2(m) and brokerage charges, commissions and conveyance and similar Taxes, to the holders of SpinCo Common Stock that would otherwise have been entitled to receive a fractional share of Parent Common Stock pursuant to the Merger (after the aggregation contemplated by this Section 3.2(h)) on a pro rata basis based on such fractional interest, without interest, as soon as practicable thereafter. The payment of cash in lieu of fractional shares is solely for the purpose of avoiding the expense and inconvenience of issuing fractional shares and does not represent separately bargained-for consideration.
- (i) No Further Ownership Rights in SpinCo Common Stock. All shares of Parent Common Stock issued in respect of shares of SpinCo Common Stock in accordance with the terms of this Section 3.2 (including any cash paid pursuant to Section 3.2(g)) shall be deemed to have been issued in full satisfaction of all rights pertaining to such shares of SpinCo Common Stock.
- (j) Termination of Exchange Fund. Any portion of the Exchange Fund made available to the Exchange Agent that remains undistributed to the SpinCo Holders on the one-year anniversary of the Effective Time shall be delivered to Parent, and any SpinCo Holders who have not received shares of Parent Common Stock in accordance with this Article III shall

thereafter look only to Parent for the Per Share Merger Consideration to which they are entitled pursuant to <u>Section 3.1(a)(i)</u> and any dividends or other distributions with respect to the Parent Common Stock to which they are entitled pursuant to <u>Section 3.2(g)</u> (subject to any applicable abandoned property, escheat or similar Law).

- (k) No Liability. None of the Company, the Surviving Corporation, Parent, Merger Sub, the Exchange Agent or any other Person shall be liable to any SpinCo Holder or holder of Company Common Stock for shares of Parent Common Stock (or dividends or distributions with respect thereto or with respect to SpinCo Common Stock) or cash properly delivered to a public official pursuant to any applicable abandoned property, escheat or similar Law.
- (l) Closing of Transfer Books. From and after the Effective Time, the stock transfer books of SpinCo shall be closed and no transfer shall be made of any shares of capital stock of SpinCo that were outstanding as of immediately prior to the Effective Time.
- (m) Tax Withholding. Parent, the Company, SpinCo, Merger Sub, and the Exchange Agent shall each be entitled to deduct and withhold from the consideration otherwise payable pursuant to this Agreement to any holder of SpinCo Common Stock such amounts as are required to be deducted and withheld with respect to the making of such payment under the Code, or under any provision of state, local or foreign Tax Law. To the extent that amounts are so deducted or withheld and paid over to the appropriate Governmental Authority, such deducted or withheld amounts will be treated for all purposes of this Agreement as having been paid to the Person in respect of which such deduction or withholding was made.
 - 3.3 Net Cash Adjustment; Forfeiture of Restricted Shares; Issuance of Additional Merger Consideration Shares.
 - (a) The Restricted Shares shall be restricted securities subject to the forfeiture provisions set forth in <u>Section 3.3(b)</u>.
- (the "Closing Net Cash Statement") setting forth in reasonable detail and accompanied by reasonably detailed backup documentation, Parent's calculation of (i) Net Cash and (ii) the Purchase Price resulting therefrom (the "Proposed Amounts"), taking into account and giving effect to the payment of any Unpaid Transaction Expenses in connection with the Closing in accordance with the terms of this Agreement. The Closing Net Cash Statement and the components thereof will be prepared in accordance with the terms of this Agreement. Parent shall provide the Company with reasonable access to such books and records as may be reasonably requested by the Company to review and verify the information contained in the Closing Net Cash Statement (subject to customary confidentiality obligations as requested by Parent and execution of customary access letters if requested by accountants). The Parties agree that the purpose of preparing the Closing Net Cash Statement and determining the final amounts of Net Cash and the related purchase price adjustment contemplated by this Section 3.3 is to measure differences between the amount of Estimated Net Cash used in the calculation of the Initial Purchase Price and the adjustment procedures set forth

in this <u>Section 3.3</u> are not intended to permit the introduction of different judgments, accounting methods, policies, principles, practices, procedures, classifications or estimation methodologies for the purpose of determining the final amount of Net Cash, other than in accordance with the terms of this Agreement.

- (c) Within thirty (30) days following the Company's receipt of the Closing Net Cash Statement, if the Company disputes in good faith the calculation of specific line items set forth in the Closing Net Cash Statement, the Company will deliver a written notice to Parent that sets forth the nature of each individual disputed item and the basis therefor in reasonable detail (including the determinations and calculations thereof) (a "Closing Net Cash Statement Dispute"). If the Company does not notify the Parent of any Closing Net Cash Statement Dispute within such thirty (30)-day period in accordance with the immediately foregoing sentence, then the Closing Net Cash Statement and the determination and calculation of the Purchase Price set forth therein will be final, conclusive and binding on the Parties and each SpinCo Holder that received Restricted Shares for all purposes of this Agreement. If the Company timely notifies Parent of a Closing Net Cash Statement Dispute in accordance with this Section3.3(c), then Parent and the Company will negotiate in good faith to resolve all disputed matters set forth therein and, if Parent and the Company so resolve all disputes, the Closing Net Cash Statement (and the computations of the Proposed Amounts indicated thereon), as amended to the extent necessary to reflect the resolution of the dispute, shall be deemed to be final, conclusive and binding on the Parties and each SpinCo Holder that received Restricted Shares for all purposes of this Agreement.
- (d) If the Company and Parent, notwithstanding such good faith effort, fail to resolve the Closing Net Cash Statement Dispute within thirty (30) days (or longer, as mutually agreed to by such parties in writing) after the Company delivers to Parent notice of the Closing Net Cash Statement Dispute, then Parent and the Company jointly will engage BDO (or if BDO is not able to serve in such capacity, another nationally or regionally recognized independent accounting firm as is mutually agreed to by Parent and the Company) (the "Independent Auditor"), to promptly resolve any and all unresolved matters of the Closing Net Cash Statement Dispute. As promptly as practicable thereafter, Parent and the Company will each prepare and submit a written presentation to the Independent Auditor (each of which shall be shared with the other party, but not until such time that both parties have submitted their presentations) and will use commercially reasonable efforts to cause the Independent Auditor to make a final determination in accordance with this Agreement with respect to only those items and amounts set forth in the Closing Net Cash Statement Dispute and the Closing Net Cash Statement that are identified by either Parent or the Company as being items that Parent and the Company are unable to resolve. The scope of the disputes to be resolved by the Independent Auditor is limited to those matters that remain in dispute and that were included in the Closing Net Cash Statement Dispute, and the Independent Auditor is not to make any other determination (including the interpretation of provisions of this Agreement unrelated to the Closing Net Cash Statement Dispute). In resolving any disputed item, the Independent Auditor will be bound by the terms of this Agreement, will serve as an expert and not an arbitrator, and will not assign a value to any item greater than the greatest value for such item claimed by either party or less than the smallest value for such item claimed by either party in the Closing Net Cash Statement or

Cash Statement Dispute (as applicable). Except as Parent and the Company may otherwise mutually agree, all communications between any party or its respective Representatives, on the one hand, and the Independent Auditor, on the other hand, will be in writing with copies simultaneously delivered to the non-communicating party (except in such cases where both parties are submitting a presentation). The fees, costs and expenses of the Independent Auditor will be borne by Parent and the Company in inverse proportion, as determined by the Independent Auditor, as they may prevail on the matter resolved by the Independent Auditor. For example, should the items in dispute total in amount to \$1,000 and the Independent Auditor awards \$600 in favor of the Company's position, sixty percent (60%) of the costs of its review would be borne by Parent and forty percent (40%) of the costs would be borne by the Company. Absent fraud or manifest error, all determinations made by the Independent Auditor and the resulting computation of Net Cash and the Purchase Price (based thereon) will be final, conclusive and binding on the Parties and each SpinCo Holder that received Restricted Shares for all purposes of this Agreement.

- (e) If the Purchase Price (as finally determined pursuant to Section 3.3(c)) is equal to or greater than an amount equal to (x) the Initial Purchase Price, plus (y) the Purchase Price Adjustment Threshold, then the Parent shall promptly issue the SpinCo Holders a number of shares of Parent Common Stock equal to the lesser of (i) the quotient of (A) the Surplus divided by the Parent Trading Price, rounded up to the nearest whole share, and (ii) 20% of the of the Merger Consideration Shares, rounded up to the nearest whole share (such shares of Parent Common Stock, the "Additional Merger Consideration Shares"). Parent will have no further liability for the Surplus other than the issuance of the Additional Merger Consideration Shares.
- (f) If the Purchase Price (as finally determined pursuant to <u>Section 3.3(c)</u>) is equal to or greater than the Initial Purchase Price, then the Parent will deliver written instructions to the Exchange Agent directing the Exchange Agent to remove the restrictive legend on 100% of the Restricted Shares, and such shares will no longer be subject to the Transfer Restriction or forfeiture provisions of this <u>Section 3.3</u>.
- (g) If the Purchase Price (as finally determined pursuant to Section 3.3(c)) is less than the difference of (x) the Initial Purchase Price, minus (y) the Purchase Price Adjustment Threshold (the amount by which the Purchase Price is less than the Initial Purchase Price, the "Deficit") and the Deficit exceeds the Restricted Shares Value, then 100% of the Restricted Shares will be forfeited and cancelled for no consideration, and Parent will deliver written instructions to the Exchange Agent directing the Exchange Agent to cancel the book-entry position of (and cancel the Restricted Shares represented thereby) 100% of the Restricted Shares of each SpinCo Holder. Absent Fraud, the SpinCo Holders will have no further liability for the Deficit other than the forfeiture of the Restricted Shares.
- (h) If the Purchase Price (as finally determined pursuant to <u>Section 3.3(c)</u>) is less than the difference of (x) the Initial Purchase Price, minus (y) the Purchase Price Adjustment Threshold and the Restricted Shares Value exceeds the Deficit, then Parent will deliver written instructions to the Exchange Agent directing the Exchange Agent to (i) cancel the book-entry position of (and cancel the Restricted Shares represented thereby) a number of Restricted Shares

of each SpinCo Holder, rounded up to the nearest whole share, equal in value (based on the Parent Trading Price) to such SpinCo Holder's pro rata share of the Deficit, and (ii) remove the restrictive legend on any remaining balance of Restricted Shares that are not so cancelled and forfeited, and such shares will no longer be subject to the Transfer Restriction or forfeiture provisions of this Section 3.3.

- (i) For so long as the Restricted Shares remain subject to the forfeiture conditions specified in this Section 3.3, the holder thereof shall be entitled to (i) exercise the voting rights carried by such Restricted Shares and (ii) receive any dividends or other distributions in respect of such Restricted Shares. For the avoidance of doubt, prior to the Restricted Shares being released from restriction in accordance with the terms of this Section 3.3, holders of the Restricted Shares shall not be entitled to transfer such shares, and such shares shall bear a legend prohibiting transfer until such time as the forfeiture conditions with respect to such shares have been satisfied (the "Transfer Restriction"). Any attempted transfer in violation of the Transfer Restriction shall be void *ab initio*.
- 3.4 <u>Appraisal Rights</u>. In accordance with Section 262 of the DGCL, no appraisal rights shall be available to the SpinCo Holders in connection with the Merger.

Article IV

REPRESENTATIONS AND WARRANTIES OF THE COMPANY RELATING TO THE COMPANY

Except as otherwise disclosed or identified in (a) the Company SEC Documents filed and publicly available on the SEC's EDGAR database at least two (2) Business Days prior to the date hereof (excluding any disclosures of factors or risks contained or references therein under the captions "Risk Factors" or "Forward-Looking Statements" to the extent they are forward-looking statements and any other similar general, predictive or cautionary statements) or (b) in the Company Disclosure Schedule (to the extent that it is reasonably apparent on the face of such disclosure that it is relevant to or applies to such representation or warranty of the Company under this <u>Article IV</u>), the Company hereby represents and warrants to Parent and Merger Sub as follows:

4.1 <u>Organization of the Company</u>. The Company has been duly incorporated and is validly existing and in good standing as a Delaware corporation. The Company has all requisite corporate power and authority to own, lease and operate its properties and assets in the manner in which such assets and properties are now owned, leased and operated and to conduct its business as it is now being conducted. The Company has made available to Parent and Merger Sub true and complete copies of the Governing Documents of the Company as in effect on the date hereof. The Company is duly licensed or qualified and in good standing (or equivalent status as applicable) in each jurisdiction in which the assets owned or leased by it or the character of its activities require it to be so licensed or qualified or in good standing (or equivalent status as applicable), except as would not, individually or in the aggregate, have a Company Material Adverse Effect.

xvii

- Due Authorization. Subject to the receipt of the Company Shareholder Approval, the Company has all requisite corporate power and authority to execute and deliver this Agreement and the Transaction Documents to which it is or will be a party and to consummate the Transactions, except for such further action of the Company Board required, if applicable, to establish the Record Date and the Distribution Date, and the declaration of the Distribution by the Company Board (the effectiveness of which is subject to the satisfaction or, to the extent permitted by applicable Law, waiver, of the conditions set forth in the Separation and Distribution Agreement). The execution and delivery by the Company of this Agreement and the Transaction Documents to which it is or will be a party as of the Effective Time and the consummation of the Transactions have been duly authorized by all necessary and proper corporate action on its part, and no other corporate action on the part of the Company is necessary to authorize this Agreement or the Transaction Documents to which it is or will be a party as of the Effective Time or, subject to such further action of the Company Board required, if applicable, to establish the Record Date and the Distribution Date, and the declaration of the Distribution by the Company Board (the effectiveness of which is subject to the satisfaction or, to the extent permitted by applicable Law, waiver of the conditions set forth in the Separation and Distribution Agreement), consummate the Transactions. Each of this Agreement and the Transaction Documents to which the Company is or will be a party as of the Effective Time has been or will be duly and validly executed and delivered by it and (assuming that this Agreement or such other applicable Transaction Documents to which each of Parent and Merger Sub is or will be a party as of the Effective Time constitutes a legal, valid and binding obligation of each of Parent and Merger Sub (as applicable)), constitutes or, when executed and delivered, will constitute the legal, valid and binding obligation of the Company, enforceable against it in accordance with its terms, subject to applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and similar Laws affecting creditors' rights generally and subject, as to enforceability, to general principles of equity (collectively, the "Remedies Exception").
- 4.3 <u>Governmental Consents</u>. No Consent of, with or to any Governmental Authority is required to be obtained or made by the Company in connection with the execution or delivery by the Company of this Agreement or the Transaction Documents to which it is or will be a party or the consummation by the Company of the Transactions, except for or in compliance with (a) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware pursuant to the provisions of the DGCL; (b) the rules and regulations of NASDAQ; (c) applicable requirements of the Securities Act and the Exchange Act; and (d) Consents described in <u>Section 5.5</u> and Consents set forth on Section 4.3 of the Company Disclosure Schedule.
- 4.4 No Conflict. Subject to the receipt of the Consents set forth in Section 4.3, the execution and delivery by the Company of this Agreement and the Transaction Documents to which it is or will be a party as of the Effective Time and the consummation by the Company of the Transactions (for the avoidance of doubt, including performance of the Transaction Documents following the Closing) do not and will not as of the Effective Time, (a) violate any provision of, or result in the breach of, any Law applicable to the Company or by which any of its assets or properties is bound; (b) with or without lapse of time or the giving of notice or both, require a consent or approval under, conflict with, result in a violation or breach of, or constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate

or cancel any Contract to which the Company is a party or by which its assets or properties are bound; or (c) breach or violate any provision of the Governing Documents of the Company, except, in the case of each of clauses (a) and (b), as would not, individually or in the aggregate, reasonably be expected to have a Company Material Adverse Effect.

- 4.5 <u>Litigation and Proceedings</u>. (a) There are no Actions pending or, to the Knowledge of the Company, threatened before or by any Governmental Authority against the Company that, individually or in the aggregate, would reasonably be expected to result in a Company Material Adverse Effect, and (b) the Company is not subject to any judgment, decree, injunction or order of any Governmental Authority that, in each case, would reasonably be expected to result, individually or in the aggregate, in a Company Material Adverse Effect.
- 4.6 <u>Brokers' Fees</u>. Except as set forth on Section 4.6 of the SpinCo Disclosure Schedule, no broker, investment banker, or other Person is entitled to any brokerage fee, finders' fee or other similar commission for which Parent or any of its Subsidiaries, including Merger Sub, the Surviving Corporation or the SpinCo, would be liable in connection with the transactions contemplated by this Agreement based on arrangements made on behalf of the Company or any of its Affiliates.

4.7 <u>Sufficiency of the SpinCo Assets.</u>

- (a) As of the Effective Time (assuming receipt of all consents, approvals and authorizations relating to the matters set forth in Section 4.3 and Section 5.5), after giving effect to the Separation, the SpinCo Assets owned by SpinCo will constitute all of the assets, properties and rights reasonably necessary for Parent and SpinCo to conduct the SpinCo Business immediately following the Closing in all material respects in the same manner as it is conducted on the date hereof.
- (b) Except for Permitted Liens or Liens created by or through Parent or Merger Sub, SpinCo and the Company collectively have, and at the Closing (after giving effect to the Separation and the other Transactions) the SpinCo shall have, good, valid and marketable title to, or a valid leasehold in, license to or other legal right to use, as the case may be, all of the SpinCo Assets, free and clear of any Liens.
- (c) The SpinCo Assets include a deposit of not less than \$3,500,000 held by Textron Aviation (the "*Textron Deposit*"). Following the Effective Time, the Parent and SpinCo will be able to use the Textron Deposit in a substantially similar manner and for substantially similar purposes as the Company prior to the Separation.
- (d) That certain securities purchase agreement by an between the Company and Ionic Ventures, LLC ("Ionic"), dated March 28, 2024 (the "Ionic Agreement") is valid and binding on the Company and, to the Knowledge of the Company, Ionic, and is in full force and effect and enforceable in accordance with its terms, subject to the Remedies Exception. Neither the Company nor, to the Knowledge of the Company, Ionic is in breach of, or default under, the Ionic Agreement in any material respect. There remains at least \$8,500,000 of available funding to the Company under the Ionic Agreement.

4.8 Internal Controls. The Company has established and maintains a system of internal controls over financial reporting (as defined in Rule 13a-15(f) or 15d-15(f), as applicable, under the Exchange Act (collectively, "Internal Controls")) that are designed to provide reasonable assurances that (i) transactions are executed in accordance with management's general or specific authorizations, (ii) transactions are recorded as necessary to permit preparation of financial statements in conformity with GAAP and to maintain accountability for assets, (iii) access to assets is permitted only in accordance with management's general or specific authorization and (iv) the recorded accountability for assets is compared with the existing assets at reasonable intervals and appropriate action is taken with respect to any differences. The Internal Controls are overseen by the audit committee of the Company Board (the "Company Audit Committee"). Since January 1, 2023, the Company's principal executive officer and its principal financial officer have disclosed to the Company's independent auditor and the Company Audit Committee (a) any significant deficiency or material weakness in the Company's Internal Controls and (b) any fraud involving management or other employees who have a significant role in the Company's Internal Controls. Since January 1, 2023, neither the Company nor any Company Subsidiary has received any material, unresolved complaint, allegation, assertion or claim regarding the impropriety of any accounting or auditing practices, procedures, methodologies or methods of the Company or any Company Subsidiary or their respective internal accounting controls.

4.9 Tax Matters.

- (a) (i) All Tax Returns required to be filed by the Company or its Subsidiaries for all periods before and through the date of the Effective Time with respect to the SpinCo Assets or the SpinCo Business have been, or will be, timely filed (taking into account applicable extensions), (ii) all such Tax Returns are, or will be when filed, true, correct, and complete, and (iii) all Taxes, whether or not shown as due on the such Tax Returns, in respect of the SpinCo Assets and the SpinCo Business have been paid or, in the case of any such Taxes not yet due and payable, will be timely paid.
- (b) (i) No Governmental Authority has asserted any written claim, assessment, or deficiency for Taxes against any of the SpinCo Assets or the SpinCo Business (and, to the Knowledge of the Company, no such claim, assessment, or deficiency has been threatened or proposed), except for deficiencies that have been fully satisfied by payment, settled or withdrawn and (ii) no claim, audit, or other proceeding by any Governmental Authority is pending or threatened in writing with respect to any Taxes of or with respect to the SpinCo Assets or the SpinCo Business.

Article V

REPRESENTATIONS AND WARRANTIES OF THE COMPANY RELATING TO SPINCO

Except as otherwise disclosed or identified in (a) the Company SEC Documents filed and publicly available on the SEC's EDGAR database at least two (2) Business Days prior to the date hereof (excluding any disclosures of factors or risks contained or references therein under the

captions "Risk Factors" or "Forward-Looking Statements" to the extent they are forward-looking statements and any other similar general, predictive or cautionary statements) or (b) in the SpinCo Disclosure Schedule (to the extent that it is reasonably apparent on the face of such disclosure that it is relevant to or applies to such representation or warranty of the Company under this Article V), the Company hereby represents and warrants to Parent and Merger Sub as follows:

- 5.1 <u>Organization of SpinCo</u>. SpinCo has been duly incorporated and is validly existing and in good standing as a Delaware corporation and has all requisite corporate power and authority to own, lease and operate its assets in the manner in which such assets are now (or as of the Effective Time will be) owned, leased or operated and to conduct its business as it is now being conducted. SpinCo has made available to Parent and Merger Sub true and complete copies of the Governing Documents of SpinCo. SpinCo is duly licensed or qualified and in good standing (or equivalent status as applicable) in each jurisdiction in which the assets owned or leased by it or the character of its activities require it to be so licensed or qualified or in good standing (or equivalent status as applicable), except as would not, individually or in the aggregate, reasonably be expected to have a SpinCo Material Adverse Effect.
- Due Authorization. SpinCo has all requisite corporate power and authority to execute and deliver this Agreement and the Transaction Documents to which it is or will be a party at the Effective Time and to consummate the Transactions (subject to the SpinCo Shareholder Approval, which will occur immediately after the execution of this Agreement, and the Company Shareholder Approval, and except for such further action of the Company Board required, if applicable, to establish the Record Date and the Distribution Date, and the effectiveness of the declaration of the Distribution by the Company (which is subject to the satisfaction or, to the extent permitted by applicable Law, waiver of the conditions set forth in the Separation and Distribution Agreement). The execution and delivery by SpinCo of this Agreement and the Transaction Documents to which it is or will be a party at the Effective Time and the consummation by SpinCo of the SpinCo Shareholder Approval, no other corporate action on the part of SpinCo is necessary to authorize this Agreement or the Transaction Documents to which it is or will be a party at the Effective Time has been, or when executed and delivered will be, duly and validly executed and delivered by SpinCo and (assuming that this Agreement or such other applicable Transaction Document to which Parent or Merger Sub is or will be a party at the Effective Time constitutes a legal, valid and binding obligation of Parent or Merger Sub (as applicable)) constitutes or will constitute a legal, valid and binding obligation of SpinCo in accordance with its terms, subject to the Remedies Exception.

5.3 <u>Capitalization of SpinCo</u>.

(a) As of the date hereof, (i) the authorized capital stock of SpinCo consists of 1,000 shares of SpinCo Common Stock, (ii) the issued and outstanding shares of capital stock of SpinCo consists of 1,000 shares of SpinCo Common Stock, and (iii) no shares of SpinCo

xxi

Common Stock are being held by SpinCo in its treasury. All of the issued and outstanding shares of SpinCo Common Stock are, as of the date hereof (and as of immediately prior to the Distribution will be), owned, of record and beneficially, by the Company and have been duly authorized and validly issued, are fully paid and nonassessable and have not been issued in violation of any preemptive or similar rights. Immediately prior to the Effective Time, there will be outstanding a number of shares of SpinCo Common Stock determined in accordance with Section 7.13.

- (b) There are no (i) outstanding options, warrants, rights or other securities convertible into or exchangeable or exercisable for shares of capital stock of SpinCo, or any other commitments or agreements providing for the issuance, sale, repurchase or redemption of shares of capital stock of SpinCo, (ii) agreements of any kind which may obligate SpinCo to issue, purchase, redeem or otherwise acquire any of its shares of capital stock or (iii) voting trusts, proxies or other agreements or understandings with respect to the voting shares of capital stock of SpinCo.
 - (c) The Specified Stockholders are set forth on Section 5.3(c) of the SpinCo Disclosure Schedule.
- 5.4 <u>Subsidiaries</u>. SpinCo does not (and will not after giving effect to the Separation) (a) have any Subsidiaries or (b) own or hold, directly or indirectly, any Interest in any other Person.
- 5.5 Governmental Consents. No Consent of, with or to any Governmental Authority is required to be obtained or made by SpinCo in connection with the execution or delivery by SpinCo of this Agreement or the Transaction Documents to which SpinCo is or will be a party at the Effective Time or the consummation by SpinCo of the Transactions, except for: (a) the filing of the Certificate of Merger with the Secretary of State of the State of Delaware pursuant to the provisions of the DGCL; (b) applicable requirements of state securities or "blue sky" Laws, the Securities Act and the Exchange Act; and (c) Consents described in Section 4.3 and Consents set forth on Section 5.5 of the SpinCo Disclosure Schedule.
- 5.6 No Conflict. Subject to the receipt of the Consents described in Section 5.5, the execution and delivery by SpinCo of this Agreement and the Transaction Documents to which SpinCo is or will be a party at the Effective Time and the consummation by SpinCo of the Transactions (for the avoidance of doubt, including performance of the Transaction Documents following the Closing by SpinCo) do not and will not: (a) violate any provision of, or result in the material breach of, any Law applicable to SpinCo or by which any of its assets or properties is bound; (b) with or without lapse of time or the giving of notice or both, require a consent or approval under, conflict with, result in a violation or breach of, or constitute a default under, result in the acceleration of, or create in any party the right to accelerate, terminate or cancel any Contract to which SpinCo is a party or by which its assets or properties is (or will at the Closing be) bound; (c) result in the creation of any Lien upon any of the properties or assets of SpinCo; or (d) violate any provision of the Governing Documents of SpinCo, except, in the case of clauses (a) and (b), as would not, individually or in the aggregate, reasonably be expected to have a SpinCo Material Adverse Effect.

xxii

5.7 <u>Sufficiency of the SpinCo Assets</u>.

- (a) As of the Effective Time (assuming receipt of all consents, approvals and authorizations relating to the matters set forth in Section 4.3 and Section 5.5), after giving effect to the Separation, the SpinCo Assets owned by SpinCo will constitute all of the assets, properties and rights reasonably necessary for Parent and SpinCo to conduct the SpinCo Business immediately following the Closing in all material respects in the same manner as it is conducted on the date hereof.
- (b) Except for Permitted Liens or Liens created by or through Parent or Merger Sub, SpinCo and the Company collectively have, and at the Closing (after giving effect to the Separation and the other Transactions) the SpinCo shall have, good, valid and marketable title to, or a valid leasehold in, license to or other legal right to use, as the case may be, all of the material SpinCo Assets, free and clear of any Liens.

5.8 Net Cash; No SEC Filing Requirements.

- (a) Set forth on Section 5.8 of the SpinCo Disclosure Schedule is a statement (the "Financial Information Schedule") that (i) contains an estimated balance sheet of SpinCo as of 11:59 p.m. Eastern Time on December 31, 2024 (the "Balance Sheet," and such date, the "Balance Sheet Date") as if the Separation and Distribution Agreement were consummated as of the Balance Sheet Date, and (ii) sets forth in reasonable detail and accompanied by reasonably detailed backup documentation the Company's and SpinCo's good-faith estimate of the Net Cash as if the Separation and Distribution Agreement were consummated as of the Balance Sheet Date ("Balance Sheet Date Net Cash"). The Financial Information Schedule and the components thereof, including the Balance Sheet Date Net Cash, were prepared in good faith from the Company's and SpinCo's books and records, were derived from the financial reporting systems and the consolidated financial statements of the Company (which financial statements were prepared in accordance with GAAP), and were calculated in accordance with the terms of this Agreement. The Financial Information Statement fairly presents, in all material respects the assets, liabilities, and stockholders' equity of the SpinCo Business as of the Balance Sheet Date.
- (b) As of the date hereof, SpinCo is not required to file or furnish any form, report, registration statement, prospectus or other document with the SEC.
- 5.9 No Undisclosed Liabilities. There is no Liability of SpinCo or related to the SpinCo Business whether or not of a type required to be reflected or reserved for on a consolidated balance sheet of the SpinCo Business or in the notes thereto prepared in accordance with GAAP, except for (a) Liabilities reflected or reserved for in the Balance Sheet; (b) Liabilities that have arisen since the Balance Sheet Date in the ordinary course of the operation of the SpinCo Business that are not, individually or in the aggregate, material; (c) Liabilities arising out of or in connection with this Agreement, the Transaction Documents and the Transactions; or (d) Liabilities that, individually or in the aggregate, would not reasonably be expected to be material.

xxiii

- 5.10 <u>Litigation and Proceedings</u>. As of the date hereof, there are no Actions pending or, to the Knowledge of the Company or SpinCo, threatened before or by any Governmental Authority against SpinCo or with respect to the SpinCo Business, and neither the Company nor any of its Subsidiaries (solely as they relate to the SpinCo Business or SpinCo) is subject to any judgment, decree, injunction or order of or investigation or inquiry by any Governmental Authority, except, in each case, as would not reasonably be expected, individually or in the aggregate, to be material to SpinCo.
- 5.11 <u>Real Property</u>. SpinCo does not own or lease, at no time previously has owned or leased and after giving effect to the Separation, will not own or lease, any real property.

5.12 <u>Tax Matters</u>.

- (a) (i) All Tax Returns required to be filed by or with respect to SpinCo or the SpinCo Business have been timely filed (taking into account applicable extensions), (ii) all such Tax Returns are true, correct and complete, and (iii) all Taxes, whether or not shown as due on such Tax Returns, in respect of SpinCo and the SpinCo Business have been paid, except to the extent adequate reserves therefor in accordance with GAAP have been provided on the SpinCo Financial Statements;
- (b) (i) No Governmental Authority has asserted any written claim, assessment or deficiency for Taxes against SpinCo or the SpinCo Business (and, to the Knowledge of the Company and SpinCo, no such claim, assessment or deficiency has been threatened or proposed in writing), except for deficiencies that have been fully satisfied by payment, settled or withdrawn and (ii) no claim, audit or other proceeding by any Governmental Authority is pending or threatened in writing with respect to any Taxes of SpinCo or the SpinCo Business;
- (c) SpinCo does not, and as of the Effective Time will not, have any Liability for Taxes of any other Person (other than the Company or any of its subsidiaries) under Treasury Regulations section 1.1502-6 (or any similar provision of state, local or non-U.S. Law), as a transferee or successor or by contract or operation of Law or otherwise;
- (d) Other than in connection with the Separation, SpinCo has not constituted either a "distributing corporation" or a "controlled corporation" (within the meaning of Section 355(a)(1)(A) of the Code) in a distribution of stock qualifying for tax-free treatment under Section 355 of the Code;
- (e) SpinCo has not participated in, or is currently participating in, and none of the SpinCo Assets or SpinCo Business has been used or is currently being used in, a "reportable transaction" within the meaning of Treasury Regulations Section 1.6011-4(b);
- (f) There are no Liens for Taxes upon the assets of SpinCo or the SpinCo Business. No Governmental Authority has threatened in writing that it is considering, or in the process of, imposing any Lien for Taxes on the assets of the Company or any of its Subsidiaries with respect to any of the SpinCo Assets or the SpinCo Business;

X	X	1	v	

- (g) None of the SpinCo Assets is, and as of the Effective Time, none of the assets owned by SpinCo will be, any stock, partnership interest, limited liability company interest, joint venture ownership interest, or other equity ownership interest in any other Person, and as of the Effective Time, SpinCo will not be subject or a party to any arrangement or contract that is treated as a partnership for income Tax purposes;
- (h) Section 5.12(h) of the SpinCo Disclosure Schedule sets froth each jurisdiction (other than U.S. federal Taxes) in which (i) the Company or any of its Subsidiaries is or has ever been registered for Taxes with respect to the SpinCo Assets or SpinCo Business, or (ii) the Company, any of its Subsidiaries, or any business of any of them (A) files, is required to file, or has been required to file any Tax Return with respect to the SpinCo Assets or SpinCo Business, or (B) is liable for any Taxes with respect to the SpinCo Assets or SpinCo Business on a "nexus" basis, identifying, in each such case, the type of Tax Return or Tax, as applicable; and neither the Company nor any of if Subsidiaries (including SpinCo) is or has ever been subject to Tax in any country other than the United States with respect to the SpinCo Assets or SpinCo Business;
- (i) No claim has been made by a taxing authority in a jurisdiction where the Company or any Subsidiary does not currently file a particular type of Tax Return or pay a particular type of Tax with respect to the SpinCo Assets or SpinCo Business that the Company or such Subsidiary is or may be required to file such a Tax Return or pay such type of Tax (including obligations to withhold amounts with respect to Taxes) in that jurisdiction with respect to the SpinCo Assets or SpinCo Business;
- (j) SpinCo is not an has never been a party to any Tax indemnity agreement, Tax sharing agreement, Tax allocation agreement, or similar contract or arrangement, whether written or unwritten; and
- (k) No payment, amount, or benefit that could be, or has been received (whether in cash or property or the vesting of cash or property or the cancellation of indebtedness) by any current or former employee, officer, stockholder, director or other individual service provider with respect to SpinCo as a result of or in connection with the execution and delivery of this Agreement or the consummation of the transactions contemplated by this Agreement could be characterized as an "excess parachute payment" (as defined in Section 280G(b)(1) of the Code or any corresponding provision of state, local or foreign Tax Law) or otherwise could have a deduction therefor disallowed by reason of the Code Section 280G or will be subject to an excise tax under Code Section 4999. SpinCo has no liability or obligation to "gross up" any Person for or with respect to any liability under Sections 409A, 457A, or 4999 of the Code (o any corresponding provision of foreign Tax Law);
- (1) SpinCo does not employ or engage any Person as an employee or contractor, has never employed or engaged any Person an employee or contractor, and, after giving effect to the Separation and as of the Effective Time, will not employ or engage any Person as an employee or contractor; and

- (m) SpinCo does not have or maintain, has never had or maintained, and after giving effect to the Separation and as of the Effective Time, will not have or maintain any Benefit Plan.
- 5.13 <u>Absence of Changes</u>. Except as set forth in Section 5.13 of the SpinCo Disclosure Schedule, since the Balance Sheet Date until the date of this Agreement, (a) there has not been any SpinCo Material Adverse Effect and (b) except in connection with the Transaction Process or as contemplated by this Agreement and the other Transaction Documents, since the Balance Sheet Date and through the date hereof, the Company and its Subsidiaries, including SpinCo, have, in all material respects, conducted the SpinCo Business in the ordinary course of business.

5.14 Material Contracts.

- (a) Except as set forth in Section 5.14(a) of the SpinCo Disclosure Schedule and except for Contracts that do not constitute SpinCo Assets, as of the date hereof, neither the Company nor any of its Subsidiaries (other than SpinCo) (solely with respect to the SpinCo Business), or SpinCo, is party to or otherwise bound by or subject to Contracts of the following types (together with any SpinCo Affiliate Contracts, the "SpinCo Material Contracts"):
- (i) any Contract that relates to the purchase or sale of goods or services pursuant to which the SpinCo Business has received more than \$50,000 or paid more than \$50,000 in the past twelve (12) months;
- (ii) any Contract that limits or purports to limit in any material respect the ability of the SpinCo Business to compete with any Person or in any line of business or in any geographic region in the world;
- (iii) any Contract that grants exclusive rights to a customer or a supplier or (to the extent material to the SpinCo Business) any other commercial counterparty that will relate to or affect the SpinCo Business after the Closing;
- (iv) any Contract that requires any future capital expenditures by the SpinCo Business in excess of \$50,000 that will not be paid prior to the Closing;
- (v) any Contract that requires any milestone, earn out or similar payments to be made by the SpinCo Business in excess of \$50,000 that will not be paid prior to the Closing;
- (vi) any Contract that relates to the creation, incurrence, assumption or guarantee of any indebtedness for borrowed money or any bonds, debentures, notes or similar instruments, in each case, in excess of \$50,000;
- (vii) any Contract pursuant to which (A) any Person grants to SpinCo or, with respect to the SpinCo Business, to the Company or any of its Subsidiaries other than SpinCo, any license, right, permission, consent, non-assertion or release with respect to any Intellectual Property that is material to the SpinCo Business, other than (1) non-exclusive click-

wrap, shrink-wrap or off-the-shelf Software licenses that are commercially available on standard and reasonable terms to the public generally with license, maintenance, support and other fees of less than \$10,000 in any twelve (12)-month period, (2) non-disclosure agreements entered into in the ordinary course of business consistent with past practice and (3) non-exclusive licenses granted by any suppliers or service providers to SpinCo in the ordinary course of business consistent with past practice solely for the receipt of services from such supplier or service provider, and solely where such licenses are ancillary to the primary purpose of such Contract, or (B) SpinCo or, with respect to the SpinCo Business, the Company or any of its Subsidiaries other than SpinCo, grants any license, right, permission, consent, non-assertion or release with respect to any Intellectual Property that is material to the SpinCo Business, other than (1) non-exclusive licenses granted to customers SpinCo in the ordinary course of business consistent with past practice, (2) non-exclusive licenses granted to any suppliers or service providers by SpinCo in the ordinary course of business consistent with past practice solely for the benefit of SpinCo and (3) non-disclosure agreements entered into in the ordinary course of business consistent with past practice;

- (viii) any Contract to which SpinCo or, with respect to the SpinCo Business, the Company or any of its Subsidiaries other than SpinCo, is a party with any Governmental Authority;
 - (ix) any lease, sublease, occupancy agreement or license for real property;
- (x) any Contract that is a settlement, conciliation or similar agreement with any Governmental Authority or that otherwise involves any settled or threatened claim, action, suit or proceeding pursuant to which the SpinCo Business has (or will have after the Closing) any monetary or other material outstanding obligation;
- (xi) any Contract that contains "most favored nation" pricing provisions for the benefit of the relevant counterparty that will relate to or affect the SpinCo Business after the Closing;
 - (xii) any joint venture, strategic alliance, joint development, partnership or similar arrangement;
- (xiii) any Contract relating to the acquisition or disposal or divestiture of, or investment in, any joint venture, partnership or similar arrangement or any material assets or businesses;
- (xiv) any prime contract, subcontract, purchase order, task order, delivery order, teaming agreement, joint venture agreement, strategic alliance agreement, basic ordering agreement, pricing agreement, letter contract or other similar arrangement of any kind where the counterparty or the ultimate customer is, or the work performed under such contract was funded by, a Governmental Authority;

XXV	11	

- (xv) any Contract related to the Company's "Jet Card Program", including any Contracts with participants of the "Jet Card Program"; and
- (xvi) any Contract not otherwise described in any other subsection of this <u>Section 5.14(a)</u> that would be required to be filed by SpinCo as a "material contract" (as such term is defined in Item 601(b)(10) of Regulation S-K of the SEC) if SpinCo were subject to the reporting requirements of the Exchange Act as of the date hereof.

The Company has made available to Parent copies of each SpinCo Material Contract that are correct and complete. Each SpinCo Material Contract is valid and binding on the Company or its applicable Subsidiary, including SpinCo and, to the Knowledge of the Company or SpinCo, the counterparty thereto, and is in full force and effect and enforceable in accordance with its terms, subject to the Remedies Exception. Neither the Company nor its applicable Subsidiary, including SpinCo is, and to the Knowledge of the Company or SpinCo, no counterparty thereto is, in breach of, or default under, any SpinCo Material Contract in any material respect.

5.15 Compliance with Law; Permits.

- (a) The Company and the Company's Subsidiaries (in each case, solely with respect to the SpinCo Business) and SpinCo are, and, during the past two (2) years SpinCo and, solely with respect to their conduct and operation of the SpinCo Business, the Company and its other Subsidiaries (i) have been in compliance with all applicable Laws, except where noncompliance has not been and would not reasonably be expected to be material to the SpinCo Business, taken as a whole, and (ii) have not received notice from any Governmental Authority alleging any material non-compliance with or possible material violation of any applicable Law or that the Company or any of its Subsidiaries (with respect to the SpinCo Business) or SpinCo, is subject to any inspection, investigation, survey, audit or other review.
- (b) Except, in each case, as would not reasonably be expected to have, individually or in the aggregate, a SpinCo Material Adverse Effect, (i) the Company and its Subsidiaries (with respect to the SpinCo Business) and SpinCo hold all Permits necessary to conduct the SpinCo Business substantially as conducted as of the date hereof and in compliance with applicable Law and (ii) such Permits are valid and in full force and effect and the Company or its applicable Subsidiary or SpinCo is in compliance with the terms thereof.

5.16 <u>Intellectual Property</u>.

(a) Section 5.18(a) of the SpinCo Disclosure Schedule sets forth a list of, as of the date hereof, all SpinCo Owned Intellectual Property that is the subject of any registration, issuance, or application for registration or issuance, with any Governmental Authority or Internet domain name registrar (specifying for each such item (i) the record owner (and, if different from the record owner, the beneficial owner), (ii) the jurisdiction in which such item has been issued, registered or filed, (iii) the issuance, registration or application date and (iv) the issuance, registration or application number) (any Intellectual Property set forth or required to be set forth on Section 5.18(a) of the SpinCo Disclosure Schedule, collectively, the "SpinCo Registered Intellectual Property").

xxviii

- (b) All SpinCo Registered Intellectual Property that is material to the SpinCo Business is subsisting, valid and enforceable. Except as scheduled in Section 5.18(b) of the SpinCo Disclosure Schedule, to the Knowledge of the Company or SpinCo, none of the SpinCo Registered Intellectual Property has been or is subject to any interference, derivation, reexamination, cancellation or opposition proceeding.
- (c) As of the date hereof, to the Knowledge of the Company or SpinCo, the Company and each of its Subsidiaries, and, as of the Separation, SpinCo, have prepared and are preparing to file patent applications for all potentially patentable inventions included in the SpinCo Owned Intellectual Property that are material to the SpinCo Business as currently conducted and as currently proposed to be conducted, except, where in the exercise of reasonable business judgment, the Company, such Subsidiary or SpinCo has decided not to file or has decided to defer filing a patent application on a potentially patentable invention. The Company, each of its Subsidiaries, and SpinCo has complied and does comply in all material respects with all Laws regarding the duty of disclosure, candor and good faith in connection with each Patent included in the SpinCo Registered Intellectual Property. To the Knowledge of the Company or SpinCo, no public disclosure bar by the Company, any of its Subsidiaries, or SpinCo arisen, which has rendered any Patent included in the SpinCo Registered Intellectual Property that is material to the SpinCo Business, invalid, unenforceable or unpatentable.
- (d) As of the date hereof, (i) the Company and its Subsidiaries solely and exclusively own, and as of the Separation, SpinCo solely and exclusively owns, all rights, title and interest in and to the SpinCo Owned Intellectual Property, in each case, free and clear of all Liens (other than Permitted Liens) and (ii) to the Knowledge of the Company or SpinCo the Company or one of its Subsidiaries has, and, as of the Separation, SpinCo has, valid and enforceable rights to use and exploit, pursuant to a written SpinCo Contract, all other Intellectual Property (except for such other Intellectual Property in the public domain for which no license is necessary) used or practiced by the SpinCo Business that is material to the SpinCo Business. The SpinCo Owned Intellectual Property and such SpinCo Licensed Intellectual Property collectively constitute all Intellectual Property used in, and necessary and sufficient for, the conduct and operation of the SpinCo Business, as currently conducted.
- (e) In the past six (6) years with respect to Patents or in the past three (3) years with respect to all other Intellectual Property, neither SpinCo or, with respect to the SpinCo Business, the Company or any of its Subsidiaries other than SpinCo, the conduct of the SpinCo Business, or any SpinCo Owned Intellectual Property has infringed, misappropriated (or constituted or resulted from a misappropriation of), diluted or otherwise violated, or is infringing, misappropriating (or constitutes or results from the misappropriation of), diluting or otherwise violating any Intellectual Property of any Person. To the Knowledge of the Company or SpinCo, neither SpinCo or, with respect to the SpinCo Business, the Company or any of its Subsidiaries other than SpinCo, has received from any Person in the past three (3) years any written notice, charge, complaint, claim or other assertion: (i) of any infringement, misappropriation, dilution or other violation of any Intellectual Property of any Person or (ii) challenging the ownership, use.

validity or enforceability of any SpinCo Owned Intellectual Property that is material to the SpinCo Business as currently conducted and as currently proposed to be conducted.

- (f) To the Knowledge of the Company or SpinCo, no other Person has infringed, misappropriated, diluted or violated, or is infringing, misappropriating, diluting or violating, any SpinCo Owned Intellectual Property or any SpinCo Licensed Intellectual Property exclusively licensed to the Company, any of its Subsidiaries, or SpinCo, in each case, that is material to the SpinCo Business. No such claims have been made in writing or, to the Knowledge of the Company or SpinCo, otherwise made against any Person by SpinCo or, with respect to the SpinCo Business, the Company or any of its Subsidiaries other than SpinCo in the past three (3) years.
- (g) To the Knowledge of the Company or SpinCo, SpinCo and, with respect to the SpinCo Business, the Company and each of its Subsidiaries other than SpinCo, have taken and currently takes adequate and commercially reasonable steps to maintain the secrecy and confidentiality of all Trade Secrets included in the SpinCo Owned Intellectual Property and all Trade Secrets of any Person to whom SpinCo or, with respect to the SpinCo Business, the Company or any of its Subsidiaries other than SpinCo, has a confidentiality obligation with respect to such Trade Secrets. No Trade Secret material to the SpinCo Business has been authorized by the Company, any of its Subsidiaries, or SpinCo to be disclosed (or, to the Knowledge of the Company or SpinCo, has been disclosed) to any Person other than (i) pursuant to a written agreement adequately restricting the disclosure and use of such Trade Secret or (ii) to a Person who otherwise has a duty to protect such Trade Secret.
- (h) To the Knowledge of the Company or SpinCo, (i) as of the date hereof, the Company or one of its Subsidiaries, and (ii) as of the Separation, SpinCo, owns or has a valid right to access and use pursuant to a written SpinCo Contract, all SpinCo IT Systems in the manner in which they are currently accessed or used in the conduct of the SpinCo Business. To the Knowledge of the Company or SpinCo, the SpinCo IT Systems are adequate in all material respects for the operation and conduct of the SpinCo Business as currently conducted. To the Knowledge of the Company or SpinCo, the SpinCo IT Systems have not materially malfunctioned or failed for the past three (3) years. To the Knowledge of the Company or SpinCo, during the past three (3) years, there has been no unauthorized access to or use of, or breach or other violation of, any SpinCo IT Systems, in each case, that has had, individually or in the aggregate, a material adverse effect on the SpinCo Business or the SpinCo IT Systems.
- (i) Neither the execution of this Agreement or any of the other Transaction Documents nor the consummation of the Transaction Process will result in (i) the loss or impairment of, or any Lien on, any material SpinCo Owned Intellectual Property or material SpinCo Licensed Intellectual Property; (ii) the release, disclosure or delivery of any source code included in the SpinCo Owned Intellectual Property to any Person; or (iii) the payment of any additional consideration to, or the reduction of any amount(s) payable from, any Person, in each case, with respect to any SpinCo Owned Intellectual Property or SpinCo Licensed Intellectual Property, and where the applicable consideration or amount(s) exceeds \$10,000.

5.17 Environmental Matters.

(a)	Except as would not otherwise reasonably	be expected to incu	ur or result in, indiv	idually or in the	aggregate, a	materia
liability of SpinCo:						

- (i) With respect to their operation of the SpinCo Business, the Company and its Subsidiaries are, and during the past three (3) years the Company and its Subsidiaries have been, in compliance with applicable Environmental Laws, which compliance includes obtaining, maintaining, and complying with all Permits required under Environmental Laws for the operation of the SpinCo Business, all of which Permits are in full force and effect;
- (ii) With respect to their operation of the SpinCo Business, the Company and its Subsidiaries have not received notice from any Governmental Authority or Person alleging any non-compliance with or Liability under any applicable Environmental Law by the Company or any of its Subsidiaries;
- (iii) No Actions pursuant to any Environmental Law to the extent affecting the SpinCo Business or any SpinCo Assets are pending or threatened in writing or, to the Knowledge of the Company or SpinCo, threatened orally against the Company or any of its Subsidiaries; and
- (iv) Neither the Company nor any of its Subsidiaries nor, to the Knowledge of the Company or SpinCo, any other Person has Released Hazardous Materials at, on, upon, into or from any real property at concentrations or under conditions that would result in the Company or any Subsidiary incurring Liability under Environmental Laws.
- 5.18 <u>Affiliate Matters</u>. Except for Contracts for employment, compensation or benefit agreements or arrangements with directors, officers and employees made in the ordinary course of business or as set forth in Section 5.18 of the SpinCo Disclosure Schedule, SpinCo is not party to any SpinCo Affiliate Contract.
- 5.19 <u>Brokers' Fees.</u> Except as set forth in Section 5.19 of the SpinCo Disclosure Schedule, no broker, finder, investment banker or other Person is entitled to any brokerage fee, finders' fee or other similar commission, for which Parent, Merger Sub or SpinCo would be liable in connection with the transactions contemplated by this Agreement or any other Transaction Document based upon arrangements made by or on behalf of SpinCo.
- 5.20 Proxy Statement; Registration Statements. None of the information regarding any of the Company or any of its Subsidiaries (including SpinCo), the SpinCo Business, or the Transactions to be provided by the Company or SpinCo or any of their respective Subsidiaries specifically for inclusion in, or incorporation by reference into, the Registration Statement, or the documents relating to the Distribution that are filed with the SEC and/or distributed to Company stockholders (the "Distribution Documents") will, in the case of the Distribution Documents or any amendment or supplement thereto, at the time of the first mailing of the Distribution Documents and of any amendment or supplement thereto, or, in the case of the Registration Statement, at the time such registration statement becomes effective, at the Distribution Time or at the Effective Time, contain an untrue or false statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in

the light of the circumstances under which they are made, not false or misleading. The Registration Statement will comply as to form in all material respects with the provisions of the Securities Act, except that no representation is made by the Company or SpinCo with respect to information provided by Parent specifically for inclusion in, or incorporation by reference into, the Registration Statement.

- 5.21 <u>Board and Shareholder Approval</u>. Each of the Company Board and the SpinCo Board, at a meeting duly called and held or by written consent, has by unanimous vote of all directors present or unanimous consent, (a) approved this Agreement, the Separation and Distribution Agreement and the other Transaction Documents and authorized and approved the execution, delivery and performance hereof and thereof and the consummation of the Transactions, including the Merger and the Separation, and (b) declared each of them advisable, fair to and in the best interests of the Company, SpinCo and their respective stockholders. As of the date hereof, the sole shareholder of SpinCo is (and as of immediately prior to the Distribution the sole shareholder of SpinCo will be) the Company. Immediately after the execution of this Agreement, the Company will approve and adopt, as SpinCo's sole shareholder, this Agreement and the Transaction Documents and the Transactions, including the Merger (the "SpinCo Shareholder Approval"). The SpinCo Shareholder Approval and Company Shareholder Approval is the only approval of the shareholders of the Company or any of its Subsidiaries required to approve the Transactions.
- 5.22 <u>Parent Capital Stock</u>. Neither the Company nor any of its Subsidiaries, including SpinCo, owns (directly or indirectly, beneficially or of record) or will own on the Closing Date, nor is a party to any Contract for the purpose of acquiring, holding, voting or disposing of, in each case, any shares of capital stock of Parent (other than as contemplated by this Agreement).

5.23 <u>Compliance with Aviation Laws</u>.

- (a) Except as set forth on Section 5.23(a) of the SpinCo Disclosure Schedule, and except as has not been, or would not reasonably be expected to be, individually or in the aggregate, material to the Company, taken as a whole, the Company and each of its Subsidiaries, including SpinCo and the SpinCo Business, (i) is in compliance with all applicable Laws, including all applicable Laws prescribed or administered by the U.S. Federal Aviation Administration ("FAA") and Department of Transportation ("DOT") under Title 14 of the Code of Federal Regulations and Title 49 of the U.S. Code (such Laws, collectively, the "Aviation Laws"), (ii) has not violated, been subject to an investigation with respect to or made voluntary disclosures with respect to potential violations of any Aviation Laws since January 1, 2023, and (iii) has not been cited by the FAA, DOT or other Governmental Authority for any material discrepancies or violations during inspections or audits since January 1, 2023.
- (b) <u>Section 5.23(b)</u> of the SpinCo Disclosure Schedule sets forth a true and complete list of each aircraft owned or leased by the Company and its Subsidiaries as of the date of this Agreement (each, an "*Aircraft*"), including the manufacturer, model, aircraft registration number and manufacturing year of each such Aircraft.

5.24 <u>Data Privacy</u>.

xxxii

- (a) (i) SpinCo (and, with respect to the SpinCo Business, the Company and each of its Subsidiaries other than SpinCo) are, and have for the past three (3) years been, in material compliance with all Privacy Requirements. During the past three (3) years, SpinCo (and, with respect to the SpinCo Business, neither the Company nor any of its Subsidiaries other than SpinCo) has not received any written notice of any claims, charges, investigations, or regulatory inquiries related to or alleging the violation of any Privacy Requirements.
- (b) SpinCo (and, with respect to the SpinCo Business, the Company and each of its Subsidiaries other than SpinCo), has (i) implemented, and for the past three (3) years has maintained, commercially reasonable technical and organizational safeguards to protect Personal Information and other confidential data in its possession or under its control, and (ii) taken commercially reasonable steps to ensure that any third party with access to any Personal Information collected by or on behalf of SpinCo (or, with respect to the SpinCo Business, the Company or any of its Subsidiaries other than SpinCo) has implemented and maintained commercially reasonable safeguards.
- (c) During the past three (3) years: (i) there have been no material breaches, security incidents, misuse of or unauthorized access to or disclosure of any Personal Information ("Security Incident") maintained by SpinCo (or, with respect to the SpinCo Business, the Company or any of its Subsidiaries other than SpinCo); nor (ii) to the Knowledge of the Company or SpinCo, has any Person processing Personal Information on behalf of SpinCo (and, with respect to the SpinCo Business, the Company or any of its Subsidiaries other than SpinCo) experienced any Security Incidents with respect to such Personal Information. SpinCo (and, with respect to the SpinCo Business, the Company and its Subsidiaries other than SpinCo) have implemented reasonable disaster recovery and business continuity plans.
- (d) To the Knowledge of the Company or SpinCo, the transfer of Personal Information in connection with the Transactions will not violate in any material respect any Privacy Requirements.
- 5.25 Anti-Bribery, Anti-Corruption and Anti-Money Laundering. None of the Company or the Company's Subsidiaries (in each case, with respect to the SpinCo Business), SpinCo, or any of their respective directors, officers, employees, agents, or any other Person acting for or on behalf of the SpinCo Business or SpinCo has, directly or indirectly, for the previous five (5) years, (a) made, offered, or promised to make or offer any payment, loan, or transfer of anything of value, including any reward, advantage, or benefit of any kind, to or for the benefit of any Government Official, candidate for public office, political party, or political campaign, for the purpose of (i) influencing any act or decision of such Government Official, candidate, party or campaign to do or omit to do any act in violation of a lawful duty, (iii) obtaining or retaining business for or with any Person, (iv) expediting or securing the performance of official acts of a routine nature, or (v) otherwise securing any improper advantage; (b) paid, offered, or promised to pay or offer any bribe, payoff, influence payment, kickback, unlawful rebate, or other similar unlawful payment of any nature; (c) made, offered or promised to make or offer any unlawful contributions, gifts, entertainment, or other unlawful expenditures; (d) established or maintained

xxxiii

any unlawful fund of corporate monies or other properties; (e) created or caused the creation of any false or inaccurate books and records of the Company or the Company's Subsidiaries (in each case, with respect to the SpinCo Business) or SpinCo; or (f) otherwise violated any provision of the Foreign Corrupt Practices Act of 1977, 15 U.S.C. §§ 78dd-1, et seq., the Money Laundering Control Act, the Currency and Foreign Transactions Reporting Act, The Uniting and Strengthening America by Providing Appropriate Tools Required to Intercept and Obstruct Terrorism Act of 2001, or any other Laws relating to corruption, bribery, or money laundering, in each case of clauses (a)-(e), in a manner that would result in a violation of any of the Laws described in clause (f). Within the past five (5) years, none of the Company, the Company's Subsidiaries, or SpinCo has made any voluntary disclosure to any Governmental Authority relating to corruption, bribery, or money laundering Laws; been the subject of any investigation or inquiry regarding compliance with such Laws; or been assessed any fine or penalty under such Laws.

- 5.26 Sanctions, Import, and Export Controls. None of the Company or the Company's Subsidiaries (in each case, with respect to the SpinCo Business), SpinCo, or any of their respective directors, officers, employees, agents, or any other Person acting for or on behalf of the SpinCo Business or SpinCo (a) is a Person with whom transactions are prohibited or limited under any economic sanctions Laws, including those administered by the U.S. government (including, without limitation, the Department of the Treasury's Office of Foreign Assets Control, the Department of State, or the Department of Commerce), the United Nations Security Council, the European Union, or Her Majesty's Treasury, or (b) has violated any Laws relating to economic sanctions within the last five (5) years. The Company and the Company's Subsidiaries (in each case, with respect to the SpinCo Business) and SpinCo are and for the past five (5) years have been in possession of and in compliance with any and all licenses, registrations, and permits that may be required for their lawful conduct under economic sanctions, import, and export control Laws. Within the past five (5) years, none of the Company, the Company's Subsidiaries, or SpinCo has made any voluntary disclosure to any Governmental Authority relating to sanctions, import, or export control Laws; been the subject of any investigation or inquiry regarding compliance with such Laws; or been assessed any fine or penalty under such Laws.
- 5.27 No Other Representations and Warranties. Except as expressly set forth in Article VI or in any Transaction Document, the Company and SpinCo each acknowledges and agrees that neither Parent, Merger Sub nor any of their Affiliates, nor any of their respective Representatives has made, or is making, any express or implied representation or warranty whatsoever with respect to Parent, Merger Sub or any of its Affiliates, or any of their respective businesses, operations, assets, liabilities, conditions (financial or otherwise) or prospects. Without limiting the generality of the foregoing, except as expressly set forth in Article VI or in any Transaction Document, each of the Company and SpinCo expressly disclaim reliance on any other representations, warranties, statements, information or inducements, oral or written, express or implied, or as to the accuracy or completeness of any statements or other information, made to, or made available to, itself or any of its Representatives, in each case with respect to, or in connection with, the negotiation, execution or delivery of this Agreement, any instrument or other document delivered pursuant to this Agreement or the transactions contemplated by this

xxxiv

Agreement, and notwithstanding the distribution, disclosure or other delivery to the Company, SpinCo or any of their Representatives of any document or other information with respect to any one or more of the foregoing.

Article VI

REPRESENTATIONS AND WARRANTIES OF PARENT AND MERGER SUB

Except as otherwise disclosed or identified in (a) the Parent SEC Documents filed and publicly available on the SEC's EDGAR database at least two (2) Business Days prior to the date hereof (excluding any disclosures of factors or risks contained or references therein under the captions "Risk Factors" or "Forward-Looking Statements" to the extent they are forward-looking statements and any other similar general, predictive or cautionary statements) or (b) the corresponding section or subsection of the Parent Disclosure Schedule (it being understood that each such disclosure shall also apply to each other representation and warranty contained in this <u>Article VI</u> to the extent that it is reasonably apparent on the face of such disclosure that it is relevant to or applies to such representation or warranty), Parent and Merger Sub hereby represent and warrant to the Company and SpinCo as follows:

6.1 Organization of Parent and Merger Sub.

- (a) Parent is a corporation duly incorporated, validly existing and in good standing under the Laws of Delaware and has all requisite corporate power and authority to own, lease and operate its assets in the manner in which such assets are now owned, leased and operated and to conduct its business as it is now being conducted. Parent has made available to the Company true and complete copies of the Governing Documents of Parent. Parent is duly licensed or qualified and in good standing (or equivalent status as applicable) in each jurisdiction in which the assets owned or leased by it or the character of its activities require it to be so licensed or qualified or in good standing (or equivalent status as applicable), except as would not, individually or in the aggregate, reasonably be expected to have a Parent Material Adverse Effect.
- (b) Merger Sub is a corporation duly incorporated, validly existing and in good standing under the Laws of Delaware. Merger Sub is a wholly owned Subsidiary of Parent. The copies of the Governing Documents of Merger Sub which were previously furnished or made available to the Company are true and complete copies of such documents as in effect on the date of this Agreement.

6.2 <u>Due Authorization</u>.

(a) Each of Parent and Merger Sub has all requisite corporate power and authority to execute and deliver this Agreement and the Transaction Documents to which it is or will be a party at the Effective Time and (subject to the receipt of the Consents described in Section 6.4) to consummate the Transactions. The execution and delivery by each of Parent and Merger Sub of this Agreement and the Transaction Documents to which it is or will be a party at the Effective Time and the consummation by each of Parent and Merger Sub of the Transactions

XXXV

have been duly and validly authorized and approved by all necessary and proper corporate action on its part, and, except as disclosed in Section 6.2 of the Parent Disclosure Schedule, no other corporate action on the part of Parent or Merger Sub is necessary to authorize this Agreement or the Transaction Documents to which it is or will be a party at the Effective Time. Each of this Agreement and the Transaction Documents to which it is or will be a party at the Effective Time has been, or when executed and delivered will be, duly and validly executed and delivered by Parent and (assuming that this Agreement or such other applicable Transaction Documents to which each of the Company or SpinCo is or will be a party at the Effective Time constitutes a legal, valid and binding obligation of each of the Company and SpinCo (as applicable)) constitutes or will constitute a legal, valid and binding obligation of Parent and Merger Sub (as applicable), enforceable against Parent and Merger Sub (as applicable) in accordance with its terms, subject to the Remedies Exception.

6.3 <u>Capital Stock and Other Matters.</u>

- (a) As of February 11, 2025, the authorized Common Stock of Parent was 300,000,000 shares, divided into (i) 200,000,000 shares of Parent Class A Common Stock, 18,199,586 of which were issued and outstanding and (ii) 100,000,000 shares of Parent Class B Common Stock, of which 59,930,000 shares were issued and outstanding ((i) and (ii) collectively, the "*Parent Securities*"). The foregoing represents all of the issued and outstanding Parent Securities as of February 11, 2025. All issued and outstanding Parent Securities (i) have been duly authorized and validly issued and are fully paid and non-assessable; (ii) have been offered, sold and issued in compliance with applicable Law, including federal and state securities Laws, and all requirements set forth in (1) Parent's Governing Documents, and (2) any other applicable Contracts governing the issuance of such securities; and (iii) are not subject to, nor have they been issued in violation of, any purchase option, call option, right of first refusal, preemptive right, subscription right or any similar right under any provision of any applicable Law, Parent's Governing Documents or any Contract to which Parent is a party or otherwise bound.
- (b) Subject to successful filing and acceptance of the Registration Statement, the shares of Parent Common Stock comprising the Merger Consideration Shares, when issued in accordance with the terms hereof, shall be duly authorized and validly issued, fully paid and non-assessable and issued in compliance with all applicable state and federal securities Laws and not subject to, and not issued in violation of, any Lien (other than applicable state and federal securities Laws), purchase, option, call option, right of first refusal, preemptive right, subscription right or any similar right under any provision of applicable Law, Parent's Governing Documents, or any Contract to which Parent is a party or otherwise.
- 6.4 <u>Governmental Consents.</u> No consent, waiver, approval or authorization of, or designation, declaration or filing with, or notification to, any Governmental Authority or other Person is required on the part of Parent or Merger Sub with respect to Parent's or Merger Sub's execution or delivery of this Agreement or the consummation of the Transactions, except as otherwise disclosed on Section 6.4 of the Parent Disclosure Schedule.

xxxvi

- 6.5 No Conflict. Subject to the receipt of the Consents described in Section 6.4, the execution and delivery by each of Parent and Merger Sub of this Agreement and the other Transaction Documents to which it is or will be a party at the Effective Time and the consummation by Parent and Merger Sub of the Transactions do not and will not as of the Effective Time: (a) violate any provision of, or result in the material breach of, any Law applicable to Parent and the Parent Subsidiaries or by which any of its assets or properties is bound; (b) with or without lapse of time or the giving of notice or both, require a consent or approval under, conflict with, result in a violation or breach of, or constitute a default under, result in the acceleration of, or create in any party the right to accelerate, terminate or cancel any Parent Material Contract; or (c) violate any provision of the Governing Documents of Parent, or Merger Sub, except, in the case of clauses (a) and (b), as would not, individually or in the aggregate, reasonably be expected to have a Parent Material Adverse Effect.
- 6.6 <u>Litigation and Proceedings.</u> As of the date of this Agreement, there are no pending or, to the Knowledge of Parent, threatened Actions against Parent or Merger Sub, their respective properties or assets, or, to the Knowledge of Parent, any of their respective directors, managers, officers or employees (in their capacity as such), that would, individually or in the aggregate, reasonably be expected to have a Parent Material Adverse Effect. As of the date of this Agreement, there are no investigations or other inquiries pending or, to the Knowledge of Parent, threatened by any Governmental Authority, against Parent or Merger Sub, their respective properties or assets, or, to the Knowledge of Parent, any of their respective directors, managers, officers or employees (in their capacity as such). As of the date of this Agreement, there is no outstanding Governmental Order imposed upon Parent or Merger Sub, nor are any assets of Parent or Merger Sub's respective businesses bound or subject to any Governmental Order the violation of which would, individually or in the aggregate, reasonably be expected to have a Parent Material Adverse Effect. From their respective dates of inception to the date of this Agreement, Parent and Merger Sub have not received any written notice of or been charged with the violation of any Laws, except where such violation has not been, individually or in the aggregate, that would reasonably be expected to have a Parent Material Adverse Effect.
- 6.7 <u>Brokers' Fees</u>. No broker, finder, investment banker or other Person is entitled to any brokerage fee, finders' fee or other similar commission, for which Parent, Merger Sub or SpinCo would be liable in connection with the transactions contemplated by this Agreement based upon arrangements made by Parent or any Parent Subsidiary.
- 6.8 SEC Filings. Parent has timely filed or furnished all statements, prospectuses, registration statements, forms, reports and documents required to be filed by it with the SEC since December 27, 2023, pursuant to the Exchange Act or the Securities Act (collectively, as they have been amended since the time of their filing through the date hereof, the "Parent SEC Filings"). Each of the Parent SEC Filings, as of the respective date of its filing (or if amended or superseded by a filing prior to the date of this Agreement or the Closing Date, then on the date of such filing), complied in all material respects with the applicable requirements of the Securities Act, the Exchange Act, the Sarbanes-Oxley Act and any rules and regulations promulgated thereunder applicable to the Parent SEC Filings. As of the respective date of its filing (or if amended or superseded by a filing prior to the date of this Agreement or the Closing Date, then

XXXX	711	

on the date of such filing), the Parent SEC Filings did not contain any untrue statement of a material fact or omit to state a material fact required to be stated therein or necessary to make the statements made therein, in light of the circumstances under which they were made, not misleading. As of the date hereof, there are no outstanding or unresolved comments in comment letters received from the SEC with respect to the Parent SEC Filings. To the Knowledge of Parent, none of the Parent SEC Filings filed on or prior to the date hereof is subject to ongoing SEC review or investigation as of the date hereof.

6.9 No Other Representations and Warranties. Except as expressly set forth in Article IV and Article V or in any Transaction Document, each of Parent and Merger Sub acknowledges and agrees that neither the Company nor any of its Affiliates (including SpinCo), nor any of their respective Representatives has made, or is making, any representation or warranty whatsoever to with respect to the Company or any of its Affiliates (including SpinCo), or any of their respective businesses, operations, assets, liabilities, conditions (financial or otherwise). Without limiting the generality of the foregoing, except as expressly set forth in Article IV or Article V or in any Transaction Document, each of Parent and Merger Sub disclaim reliance on any other representations, warranties, statements, information or inducements, oral or written, express or implied, or as to the accuracy or completeness of any statements or other information, made to, or made available to, itself or any of its Representatives, in each case with respect to, or in connection with, the negotiation, execution or delivery of this Agreement.

Article VII

COVENANTS

- 7.1 Conduct of Business by Parent and Merger Sub Pending the Merger. From the date hereof and prior to the Effective Time (or the earlier termination of this Agreement) (the "Interim Period"), except as (i) required or otherwise contemplated by this Agreement (including as set forth in Section 7.1 of the Parent Disclosure Schedule) or the Transaction Documents, (ii) as consented to by the Company in writing (which consent shall not be unreasonably withheld, conditioned, delayed or denied) or (iii) as required by applicable Law, Parent shall, and shall cause its Subsidiaries, including Merger Sub, as applicable, to, conduct its operations in the ordinary course of business in all material respects. During the Interim Period, Parent shall, and shall cause its Subsidiaries, including Merger Sub, to comply with, and continue performing under, as applicable, the Governing Documents of Parent. Without limiting the generality of the foregoing, during the Interim Period, except as (i) required or otherwise contemplated by this Agreement (including as set forth in Section 7.1 of the Parent Disclosure Schedule) or the Transaction Documents, (ii) as consented to by the Company in writing (which consent shall not be unreasonably withheld, conditioned, delayed or denied) or (iii) as required by applicable Law, Parent shall not, and shall cause its Subsidiaries, including Merger Sub, as applicable, not to:
- (a) seek any approval from the Parent Stockholders to amend, modify, restate, waive, rescind or otherwise change the Governing Documents of Parent or Merger Sub;

XXXV	111

- (b) (i) make or declare any dividend or distribution to the Parent Stockholders or make any other distributions in respect of any of Parent's equity interests or Merger Sub capital stock, share capital or equity interests or (ii) split, combine, reclassify or otherwise amend any terms of any shares or series of Parent's equity interests or Merger Sub capital stock or equity interests; or
 - enter into any agreement to do any action prohibited under this <u>Section 7.1</u>.
- 7.2 Conduct of SpinCo Business Pending the Merger. During the Interim Period (solely with respect to SpinCo or the SpinCo Business), except as (i) required or otherwise contemplated by this Agreement (including as set forth in Section 7.2 of the SpinCo Disclosure Schedule) or the Transaction Documents (including the Separation and Distribution), (ii) as consented to by Parent in writing or (iii) as required by applicable Law, the Company and SpinCo shall, and shall cause and ensure that their respective Subsidiaries, (x) conduct the SpinCo Business in the ordinary course of business in all material respects, (y) use commercially reasonable efforts to manage the SpinCo Business' working capital and maintain the books and records related to the SpinCo Business consistent with past practice and (z) use commercially reasonable efforts to maintain their respective relations and goodwill with all material suppliers, material customers and other material commercial counterparties and Governmental Authorities (in each case, as related to the SpinCo Business). Without limiting the generality of Section 7.2(a), during the Interim Period (solely with respect to SpinCo or the SpinCo Business), except as (A) required or contemplated by this Agreement (including as set forth in Section 7.2 of the SpinCo Disclosure Schedule), or the Transaction Documents, (B) as consented to by Parent in writing or (C) as required by applicable Law, the Company and SpinCo shall not, and each shall cause its respective Subsidiaries not to (in each case solely with respect to SpinCo or the SpinCo Business):
- (a) amend, modify, restate, waive, rescind or otherwise change the Governing Documents of SpinCo, other than an amendment to the certificate of incorporation of SpinCo to increase the number of authorized or outstanding shares of SpinCo Common Stock in connection with the Distribution in accordance with this Agreement and the Transaction Documents;
- (b) other than as required for the Distribution, (i) declare, set aside or pay any dividends on or make other distributions in respect of any of the Interests of SpinCo (whether in cash, securities or property), except for the declaration and payment of cash dividends or distributions paid on or with respect to a class of Interests of SpinCo that is wholly owned directly or indirectly by SpinCo, (ii) split, combine, subdivide, reduce, or reclassify any of the Interests of SpinCo or issue or authorize or propose the issuance of any other securities in respect of, in lieu of, or in substitution for, Interests of SpinCo or (iii) redeem, repurchase or otherwise acquire, or permit any Subsidiary to redeem, repurchase or otherwise acquire, any Interests (including any securities convertible or exchangeable into such Interests) (except for the acquisition by the Company or any of its Subsidiaries of any shares of capital stock, membership interests or other equity interests of the Company or its Subsidiaries in connection with the

xxxix

forfeiture of any Company Equity Awards, in each case, in accordance with their respective terms as in effect as of the date hereof);

- (c) other than as contemplated by the Distribution, issue, sell, pledge, dispose of, grant, transfer or encumber, any shares of capital stock of SpinCo of any class, or securities convertible into, or exchangeable or exercisable for, any shares of such capital stock or other Interests in SpinCo, or any options, warrants, stock units, or other rights of any kind to acquire any shares of capital stock or other Interests or such convertible or exchangeable securities, or any other ownership interest (including any such interest represented by Contract right), or any "phantom" stock, "phantom" stock rights, stock appreciation rights or stock based performance rights, in each case, of SpinCo;
- (d) sell, assign, transfer, convey, lease, license, abandon, mortgage, pledge or permit any Lien on (other than a Permitted Lien) or otherwise dispose of any SpinCo Assets (excluding Intellectual Property, which is the subject of Section 7.2(e) below);
- (e) (i) purchase, sell, license, sublicense, lease, pledge, covenant not to assert, assign, transfer, abandon, cancel, let lapse or expire, or otherwise dispose, transfer or grant any other rights in or with respect to any Intellectual Property (other than with respect to (A) immaterial or obsolete SpinCo Owned Intellectual Property or (B) the grant of non-exclusive licenses of SpinCo Owned Intellectual Property in the ordinary course of business consistent with past practice); or (ii) disclose any material Trade Secrets of the SpinCo Business to any other Person (other than in the ordinary course of business consistent with past practice to a Person bound by adequate and enforceable use restrictions and confidentiality obligations with respect to such Trade Secrets);
- (f) receive, collect, use, store, process, share, safeguard, secure (technically, physically or administratively), dispose of, destroy, disclose, or transfer (including cross-border) any Personal Information (or fail to do any of the foregoing, as applicable) in material violation of any Privacy Requirements;
- (g) merge, combine or consolidate (pursuant to a plan of merger or otherwise) SpinCo with any Person or adopt a plan of complete or partial liquidation or resolutions providing for a complete or partial liquidation, dissolution, restructuring, recapitalization or other reorganization of SpinCo;
- (h) acquire (including by merger, consolidation, or acquisition of shares or assets), lease or license, (i) any interest in any Person or (ii) any assets of any Person that would be an asset of SpinCo at the Effective Time, other than, in the case of clause (ii), (A) in the ordinary course of business with respect to assets having a value not exceeding \$10,000, individually, or \$50,000, in the aggregate, the purchase price for which will be paid by the Company or any of its Subsidiaries prior to the Distribution Date;
- (i) repurchase, repay, prepay, refinance or incur any indebtedness for borrowed money, issue any debt securities, engage in any securitization transactions or similar arrangements or assume, guarantee or endorse, or otherwise as an accommodation become

responsible for (whether directly, contingently or otherwise), the obligations of any Person for borrowed money;

- (j) make any material loans, material capital contributions or material investments in, or advances of money to, in each case, in excess of \$10,000 individually or \$20,000 in the aggregate, any Person (other than SpinCo), in each case, except for (i) advances to employees or officers of SpinCo for expenses incurred in the ordinary course of the SpinCo Business consistent with past practice and in accordance with the Company's and its Subsidiaries' policies in respect thereof or (ii) extended payment terms for customers in the ordinary course of the SpinCo Business consistent with past practice;
- (k) (A) amend or modify in any material respect, terminate (excluding any expiration in accordance with its terms), or waive any material right, benefit or remedy under, any SpinCo Material Contract or (B) enter into any Contract that if entered into prior to the date hereof would be required to be listed on Section 5.14(a) of the SpinCo Disclosure Schedule;
- (l) (i) adopt or enter into any Benefit Plan, (ii) grant or provide any change-in-control, severance, termination, retention or similar payments or benefits to any employee or service provider of SpinCo, or that would result in any material Liability to SpinCo, (iii) hire or engage, or make an offer to hire or engage, any officer, employee, service provider or individual independent contractor of SpinCo, or (iv) terminate (without cause) the employment of any SpinCo Employee or engagement of any SpinCo Independent Contractor;
- (m) except as required or permitted by GAAP, make any material change to any financial accounting principles, methods or practices of SpinCo or with respect to the SpinCo Business;
 - (n) waive, release, settle, compromise or otherwise resolve any Action, litigation or other proceedings;
- (o) (i) make, change or revoke any material Tax election in respect of the SpinCo Assets or the SpinCo Business that would bind SpinCo for periods following the Effective Time (other than an election reasonably required in connection with the Separation) or (ii) settle or compromise any material Tax liability for which SpinCo would be responsible under any Transaction Document;
- (p) make or commit to make any capital expenditures, on an annualized basis, in the aggregate, in excess of \$20,000, in the aggregate;
- (q) disclose or agree to disclose to any Person (other than Parent or any of its representatives) any material Trade Secret or any other material confidential or proprietary information, know-how or process of the Company or any of its Subsidiaries, in each case other than in the ordinary course of business consistent with past practice and pursuant to customary contractual obligations or fiduciary or professional duty to maintain the confidentiality thereof;

- (r) terminate without replacement or fail to use reasonable efforts to maintain any license or permit that is material to the conduct of the business of the Company and its Subsidiaries, taken as a whole;
 - (s) waive the restrictive covenant obligations of any current officer of the Company or SpinCo;
- (t) (i) limit the right of the Company or SpinCo to engage in any line of business or in any geographic area, to develop, market or sell products or services, or to compete with any Person or (ii) grant any exclusive or similar rights to any Person, in each case, except where such limitation or grant does not, and would not be reasonably likely to, individually or in the aggregate, materially and adversely affect, or materially disrupt, the ordinary course operation of the businesses of the Company or SpinCo, taken as a whole;
- (u) terminate without replacement or amend in a manner materially detrimental to the Company and its Subsidiaries, taken as a whole, any insurance policy insuring the business of the Company or SpinCo; or
- (v) authorize or enter into any Contract to do any of the foregoing or otherwise agree or make any commitment to do any of the foregoing.

7.3 Tax Matters.

- (a) The Company and SpinCo, on the one hand, and Parent, on the other hand, shall cooperate with each other in obtaining, and shall use their respective reasonable best efforts to obtain, any Tax opinions required to be filed with the SEC in connection with the filing of the Registration Statement and shall each use its respective reasonable efforts to cause such opinions to be timely filed.
- (b) At and after the Effective Time, each of the Parties covenants and agrees that it will maintain all books and records and file all federal, state, and local income Tax Returns and schedules thereto of the Parent, SpinCo, and Merger Sub in a manner consistent with the Merger's being qualified as a reorganization and nontaxable exchange under Section 368(a)(1)(A) of the Code (and comparable provisions of any applicable state or local income Tax Laws); provided, that Parent and Merger Sub make no representations or covenants to the Company, SpinCo or any of their stockholders as to whether the Merger ultimately will be determined to be as such a reorganization.
- (c) No later than sixty (60) days after the Effective Time, the Company shall provide Parent with a schedule setting forth SpinCo's holding period and adjusted basis for U.S. federal income Tax purposes, as well as any differences thereof for any state or local Tax purposes, as of the Effective Time, of each of the SpinCo Assets and any other assets held by SpinCo as of the Effective Time.
- (d) From and after the Closing Date, the Company will include SpinCo in its consolidated federal income Tax Return, as well as all state or local income Tax Returns for

which consolidation is permitted, for all periods through end of the day of the Effective Time. The Company will prepare the initial draft of all income Tax Returns of the Company and its Subsidiaries for each Pre-Closing Tax Period that are required to be filed after the Closing Date. All such income Tax Returns for any Pre-Closing Tax Period must be prepared (i) in accordance with applicable Law and (ii) consistent with the past practices of the Company except as otherwise required by applicable Law. The Company will give the Parent a reasonable opportunity, and in no event less than thirty (30) days, to review, comment on and approve any income Tax Returns that relate to a Pre-Closing Tax Period of the Company or applicable Subsidiary. The Company will make such revisions to those income Tax Returns as are reasonably requested by the Parent that (i) are consistent with the terms of this Agreement with respect to Tax liabilities related to SpinCo, the SpinCo Assets, or the SpinCo Business and (ii) are in accordance with applicable Tax Law. The parties will cooperate fully and promptly in connection with the preparation and filing of such income Tax Returns, and the Company will timely file all income Tax Returns for each Pre-Closing Tax Period that are required to be filed after the Closing Date. Parent will be responsible for preparing and filing, or causing to be timely prepared and filed, all Tax Returns for each Straddle Period and Post-Closing Tax Period of SpinCo.

- (e) Whenever it is necessary to determine the liability for Taxes of SpinCo for a portion of a taxable year or period that begins on or before and ends after the Closing Date, in the case of any Taxes that are imposed on a periodic basis and are payable for a Straddle Period, the portion of such Tax that constitutes a Pre-Closing Tax will be as follows: (i) in the case of any Taxes other than Taxes based upon or related to income, payroll, or receipts, the portion of such Tax constituting a Pre-Closing Tax will equal the amount of such Tax for the entire Tax Period multiplied by a fraction the numerator of which is the number of days in the Tax Period; and (ii) in the case of any Tax based upon or related to income or receipts, the portion of such Tax constituting a Pre-Closing Tax will be determined and apportioned using an interim closing of the books as of the close of business on the day of the Effective Time except that exemptions, allowances or deductions that are calculated on an annual basis will be allocated to the period ending at the end of the day of the Effective Time in the same proportion as the number of calendar days during the Taxable Period through such date bears to the number of calendar days in the entire Taxable Period.
- (f) Parent and the Company will cooperate, as and to the extent reasonably requested by the other of the parties, in connection with the preparation and filing of Tax Returns pursuant to this Agreement. Such cooperation will include the retention and (upon any other party's request) the provision of records and information that are reasonably relevant to the preparation and filing of any such Tax Return. Parent and the Company will retain all books and records with respect to Tax matters pertinent to SpinCo relating to any Taxable Period beginning on or before the Closing Date until the expiration of the applicable statute of limitations (and, to the extent notified by the Parent or the Company, any extensions thereof) of the respective Taxable Periods, and to abide by all record retention agreements entered into with any Tax authority.

- (g) So long as taxable periods of SpinCo ending on or before, or including, the day of the Effective Time remain open for an assessment of Tax, Parent and the Company will notify the other in writing within ten (10) Business Days after receipt by Parent or the Company of written or oral notice of any pending or threatened audit or assessment with respect to Taxes of SpinCo, the SpinCo Assets or the SpinCo Business relating to any Pre-Closing Tax Period or Straddle Period.
- (h) The Company will not agree to any adjustment, or make any Tax election, that will or may create an increase in Taxes for SpinCo or the Parent in respect of any Tax Period ending after the Closing Date, except with Parent's prior written consent.
- (i) Notwithstanding <u>Section 10.4</u>, all federal, state, local, foreign and other transfer, sales, use or similar Taxes applicable to, imposed upon or arising out of the Transactions will be paid by the Company.

7.4 Preparation of the Registration Statement.

- (a) Within fourteen (14) Business Days after the execution of this Agreement, Parent, the Company and SpinCo shall jointly prepare and Parent shall file with the SEC the Registration Statement.
- (b) Each of Parent, the Company and SpinCo shall use its reasonable best efforts to have the Registration Statement declared effective as promptly as practicable after such filing (including by responding to comments of the SEC) and to keep such registration statements effective for as long as is necessary to consummate the Transactions, and, prior to the effective date of the Registration Statement, each of Parent, the Company and SpinCo shall take all action reasonably required (other than qualifying to do business in any jurisdiction in which it is not now so qualified or filing a general consent to service of process in any such jurisdiction) to be taken under any applicable securities Laws in connection with the Parent Share Issuance and the Distribution. As promptly as practicable after the Registration Statement shall have become effective, each of the Company and Parent, as applicable, shall cause the Distribution Documents or the applicable portions of the Registration Statement (including the prospectus included therein), as applicable, to be mailed or made available to the Company's stockholders, in each case pursuant to applicable Law. No filing of, or amendment or supplement to, the Registration Statement will be made by Parent without providing the Company and SpinCo with a reasonable opportunity to review and comment thereon (and such comments shall be reasonably considered by Parent in good faith). Each Party (as applicable) will use its reasonable best efforts to cause the Distribution Documents and the Registration Statement to comply in all material respects with the applicable requirements of U.S. federal securities Laws.
- (c) Each of Parent, the Company and SpinCo shall ensure that none of the information supplied by or on its behalf for inclusion or incorporation by reference in (A) the Registration Statement will, at the time filed with the SEC, at each time at which it is amended and at the time it becomes effective under the Securities Act, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary to make the statements therein, not misleading or (B) the Distribution Documents will, at the date they are

xliv

mailed or made available to the Company's stockholders, contain any untrue statement of a material fact or omit to state any material fact required to be stated therein or necessary in order to make the statements therein, in light of the circumstances under which they are made, not misleading.

- If, at any time prior to the Effective Time, any information relating to Parent, the Company or SpinCo, or any of their respective Affiliates, directors or officers, should be discovered by Parent, the Company or SpinCo, which should be set forth in an amendment or supplement to the Registration Statement, so that the Registration Statement would not include any misstatement of a material fact or omit to state any material fact necessary to make the statements therein, in light of the circumstances under which they were made, not misleading, the Party that discovers such information shall promptly notify the other Party and an appropriate amendment or supplement describing such information shall be promptly filed with the SEC, and, to the extent required by Law, disseminated to the shareholders of Parent or the Company, as applicable. Each Party shall notify the other Party promptly of the time when the Registration Statement has become effective and of the issuance of any stop order or suspension of the qualification of the shares of Parent Common Stock issuable pursuant to the Merger or shares of SpinCo Common Stock issuable in the Distribution for offering or sale in any jurisdiction. In addition, each Party agrees to provide the other Party and their respective counsel promptly with copies of any written comments or requests for amendments or supplements, and shall promptly inform the other Party of any oral comments or requests for amendments or supplements, that such Party or its counsel may receive from time to time from the SEC with respect to the Registration Statement, promptly after receipt of such comments, and shall provide the other Party with copies of any written or oral responses or correspondence between it or its Affiliates and the SEC related thereto. Each Party and their respective counsel shall be given a reasonable opportunity to review in advance any such written responses and to participate in any discussions or oral material communications with the SEC, and each Party shall reasonably consider in good faith the additions, deletions, comments or changes suggested thereto by the other Party and its counsel.
- (e) Promptly following the final determination, if any, that Parent is required to issue Additional Merger Consideration Shares pursuant to Section 3.3(e), to the extent the Registration Statement does not effect the registration under the Securities Act of the issuance of a sufficient number of shares to include the Additional Merger Consideration Shares, Parent will prepare and file with the SEC a post-effective amendment to the Registration Statement or other registration statement (as determined in Parent's reasonable discretion) in order to effect the registration under the Securities Act of the issuance of the such Additional Merger Consideration Shares.
- 7.5 <u>Company Shareholders Meeting</u>. The Company shall call, give notice of, convene and hold a general meeting (the "*Company Shareholders Meeting*") as promptly as reasonably practicable following the date on which the Registration Statement is declared effective, for the purpose of obtaining the Company Shareholder Approval; <u>provided</u>, that subject to the requirements of any applicable Law, the Company may (and, in the case of clause (C) on up to two (2) occasions upon the reasonable request of the Parent (and for no more than five (5)

Business Days each) shall) postpone or adjourn the Company Shareholders Meeting (A) if a quorum has not been established; (B) after consultation with Parent, to allow reasonable additional time for the filing and mailing of any supplement or amendment to the Proxy Statement as may be required under applicable Law and for such supplement or amendment to be disseminated and reviewed by Company's shareholders sufficiently in advance of the Company Shareholders Meeting; (C) to allow reasonable additional time to solicit additional proxies, if and to the extent the requisite Company Shareholder Approval would not otherwise be obtained; (D) after consultation with Parent, if otherwise required by applicable Law; or (E) with the prior written consent of Parent; provided, that the Company Shareholders' Meeting will not be adjourned to a date that is more than thirty (30) days after the date for which the Company Shareholders' Meeting was originally scheduled (excluding any adjournments required by applicable Law). The Company shall advise Parent upon request on a daily basis during each of the last five (5) Business Days prior to the date of the Company Shareholders Meeting as to the aggregate tally of proxies received by Parent with respect to the Company Shareholder Approval and at additional times upon the reasonable request of Parent.

Reasonable Best Efforts. Each of Parent, the Company and their respective Subsidiaries shall use its reasonable best efforts to promptly take, or cause to be taken, all actions, and to promptly do, or cause to be done, and to assist and cooperate with the other in doing, all things reasonably necessary, proper or advisable under applicable Laws to consummate and make effective the Merger and the other transactions contemplated by this Agreement and the other Transaction Documents, as promptly as practicable and in any event prior to the Outside Date, including (i) the obtaining of all necessary actions or nonactions, waivers, consents, clearances, approvals, and expirations or terminations of waiting periods, from Governmental Authorities and the making of all necessary registrations and filings in connection therewith, and (ii) using its commercially reasonable efforts to obtain all necessary consents, approvals or waivers from third parties.

7.7 Access to Information.

(a) The Company shall, and shall cause its Subsidiaries, on the one hand, and Parent shall, and shall cause its Subsidiaries, on the other hand, to afford to the other Party and to its respective Representatives, reasonable access, during normal business hours and subject to bona fide policies and procedures established by the other Party, during the Interim Period, in such manner as to not interfere with Parent's and its Subsidiaries' business's or the SpinCo Business's (as applicable) normal operations, the properties, books and records and appropriate senior-level officers and employees of Parent and its Subsidiaries or the Company and its Subsidiaries (including SpinCo) that are related to the SpinCo Business (as applicable), and shall furnish such Party and its Representatives with such accounting (including accountants' work papers), financial and operating data and other information concerning the affairs of Parent and its Subsidiaries or the SpinCo Business or SpinCo (as applicable), in each case, as such Party and its Representatives may reasonably request for the purposes of furthering the Transactions or for purposes of preparing for the operation of Parent and the Surviving Corporation post-Closing; provided, that (x) such investigation shall only be upon reasonable notice and at the sole cost and expense of the investigating Party; and (y) nothing in this Agreement shall require any Party to

permit any inspection or disclose any information to any other Party that (i) would unreasonably interfere with the conduct of such Party's business or result in damage to property (other than immaterial damage), except with such other Party's prior written consent (which may be withheld or denied at its sole discretion), (ii) would cause a violation of any Law, privacy policy or any confidentiality obligations and similar restrictions that may be applicable to such information, or (iii) would jeopardize the attorney-client privilege or other disclosure privilege or protection to such Party; provided, that the Party that would otherwise be required to disclose information to the other shall take any and all reasonable action necessary to permit such disclosure without such loss of privilege or violation of agreement, policy, Law or other restriction, including through the use of commercially reasonable efforts to obtain any required consent or waiver to the disclosure of such information from any third party and through the implementation of appropriate and mutually agreeable "clean room" or other similar procedures designed to limit any such adverse effect of sharing such information by each Party).

- 7.8 Public Announcements. Except (a) as otherwise expressly contemplated by this Agreement, (b) in connection with any press release, public statement or filing to be issued or made by Parent with respect to any Modification in Recommendation, and (c) for the separate or joint press releases to be issued by the Parties in the forms agreed by the Parties (or any public statement or disclosure that contains or reflects only such information previously disclosed in press releases or other public disclosures made in accordance with this Section 7.8), neither Parent nor the Company will, and each of Parent and the Company will cause its Subsidiaries not to, issue any press release or otherwise make any public statements or disclosure with respect to the Transactions without the prior written consent of the other Party. Notwithstanding the foregoing, to the extent such disclosure is required by applicable Law or the rules of any stock exchange, the Party seeking to make such disclosure will promptly notify the other Party thereof and the Party making such statement will use efforts reasonable under the circumstances to consult in good faith with the other Party thereto prior to making such disclosure in order to allow a mutually agreeable release or announcement to be issued. Notwithstanding the foregoing, any Party may make statements that are consistent with previous public releases made by such Party in compliance with this Section 7.8.
- 7.9 <u>Defense of Litigation</u>. Parent and the Company shall provide the other Party prompt notice in writing of any Action brought by any stockholder or purported stockholder of such Party against it, any of its Subsidiaries or any of their respective directors and officers (including, with respect to the Company, SpinCo) relating to this Agreement, the Transactions, the Transaction Documents and any of the matters contemplated hereby or thereby, and shall keep the other Party informed on a reasonably current basis with respect to the status thereof and consider any comments or suggestions made by the other Party with respect to the strategy therefor; <u>provided</u>, that prior to the Effective Time, no Party shall compromise, settle, come to an arrangement regarding or agree to compromise, settle or come to an arrangement regarding any Action arising or resulting from the Transactions or consent to the same, without the prior written consent of the other Party to the extent (a) such Action includes the other Party or any of its Subsidiaries, directors or officers as named defendants or (b) such compromise, settlement or arrangement would reasonably be expected to prevent, materially impair, materially delay or

xlvii

otherwise have a material adverse effect on the ability of the Parties to perform their respective obligations hereunder, or to consummate the Transactions in a timely manner.

7.10 No Solicitation.

- (a) The Company shall not, and shall cause its Subsidiaries not to, and shall use reasonable best efforts to cause its and its Subsidiaries' Representatives not to, directly or indirectly, solicit, initiate, or knowingly take any action to facilitate the submission of any Takeover Proposal or the making of any proposal that could reasonably be expected to lead to any Takeover Proposal, or, subject to Section 7.10(b): (i) conduct or engage in any discussions or negotiations with, disclose any non-public information relating to the Company or any of its Subsidiaries to, afford access to the business, properties, assets, books, or records of the Company or any of its Subsidiaries to, or knowingly assist, participate in, facilitate, or encourage any effort by, any third party that is seeking to make, or has made, any Takeover Proposal; or (ii) enter into any agreement in principle, letter of intent, term sheet, acquisition agreement, merger agreement, option agreement, joint venture agreement, partnership agreement, or other contract relating to any Takeover Proposal (each, a "Company Acquisition Agreement"). Except as expressly permitted by this Section 7.10(a), the Company Board shall not effect a Company Adverse Recommendation Change. The Company shall, and shall cause its Subsidiaries to cease immediately and cause to be terminated, and shall not authorize or knowingly permit any of its or their Representatives to continue, any and all existing activities, discussions, or negotiations, if any, with any third party conducted prior to the date hereof with respect to any Takeover Proposal and shall use its commercially reasonable efforts to cause any such third party (or its agents or advisors) in possession of non-public information in respect of the Company or any of its Subsidiaries that was furnished by or on behalf of the Company and its Subsidiaries to return or destroy (and confirm destruction of) all such information. The Company will be liable for any breach of this Section 7.10(a) by its Representative.
- (b) Notwithstanding Section 7.10(a), prior to the receipt of the Company Shareholder Approval, the Company Board, directly or indirectly through any Representative, may, subject to Section 7.10(c) and Section 7.10(d): (i) participate in negotiations or discussions with any third party that has made (and not withdrawn) a bona fide, unsolicited Takeover Proposal in writing that the Company Board believes in good faith, after consultation with outside legal counsel and its other professional advisors (and, if necessary, contact with such third party to clarify the terms and conditions of such Takeover Proposal), constitutes or would reasonably be expected to result in a Superior Proposal; (ii) thereafter furnish to such third party non-public information relating to the Company or any of its Subsidiaries pursuant to an executed confidentiality agreement; (iii) following receipt of and on account of a Superior Proposal, make a Company Adverse Recommendation Change; and (iv) take any action that any court of competent jurisdiction orders the Company to take (which order remains unstayed). Nothing contained herein shall (i) prevent the Company Board from disclosing to the Company's stockholders a position contemplated by Rule 14d-9 and Rule 14e-2(a) promulgated under the Exchange Act; (ii) making any "stop, look and listen" communication to the stockholders of the Company pursuant to Rule 14d-9(f) under the Exchange Act or (iii) making any disclosures to

xlviii

the stockholders of the Company with regard to the transactions contemplated by this Agreement or any Takeover Proposal required by Law.

- (c) The Company shall notify Parent promptly (but in no event later than 48 hours) after receipt by the Company (or any of its Representatives) of any Takeover Proposal, any inquiry that could reasonably be expected to lead to a Takeover Proposal, any request for non-public information relating to the Company or any of its Subsidiaries or for access to the business, properties, assets, books, or records of the Company or any of its Subsidiaries by any third party. In such notice, the Company shall identify the third party making, and details of the material terms and conditions of, any such Takeover Proposal, indication or request. The Company shall keep Parent reasonably informed of material developments affecting the status and material terms of any such Takeover Proposal, indication or request. The Company shall promptly provide Parent with a list of any non-public information concerning the Company's and any of its Subsidiary's business, present or future performance, financial condition, or results of operations, provided to any third party, and, to the extent such information has not been previously provided to Parent, copies of such information.
- (d) Except as expressly permitted by this Section 7.10, the Company Board shall not effect a Company Adverse Recommendation Change or enter into (or permit any Subsidiary to enter into) a Company Acquisition Agreement. Notwithstanding the foregoing, at any time prior to the receipt of the Company Shareholder Approval, the Company Board may effect a Company Adverse Recommendation Change or enter into (or permit any Subsidiary to enter into) a Company Acquisition Agreement, if: (i) the Company promptly notifies Parent, in writing, at least two (2) Business Days (the "Superior Proposal Notice Period") before making a Company Adverse Recommendation Change or entering into (or causing a Subsidiary to enter into) a Company Acquisition Agreement, of its intention to take such action with respect to a Superior Proposal, which notice shall state expressly that the Company has received a Takeover Proposal, that the Company Board intends to declare a Superior Proposal and that the Company Board intends to effect a Company Adverse Recommendation Change and/or the Company intends to enter into a Company Acquisition Agreement; (ii) the Company includes in such notice a description in reasonable detail of such Superior Proposal and the identity of the third party making such Superior Proposal; (iii) the Company shall, and shall cause its Representatives to, during the Superior Proposal Notice Period, negotiate with Parent in good faith to make such adjustments in the terms and conditions of this Agreement so that such Takeover Proposal ceases to constitute a Superior Proposal, if Parent, in its discretion, proposes to make such adjustments; and (iv) the Company Board determines in good faith, after consulting with outside legal counsel and its other professional advisors, that such Takeover Proposal continues to constitute a Superior Proposal after taking into account any adjustments made by Parent during the Superior Proposal Notice Period in the terms and conditions of this Agreement.
- (e) Notwithstanding anything to the contrary in the foregoing, the Company Board may effect a Company Adverse Recommendation Change, after the date of this Agreement but prior to the receipt of the Company Shareholder Approval, if: (i) prior to effecting the Company Adverse Recommendation Change, the Company Board determines in good faith, after consulting with outside legal counsel and its other professional advisors, that the

failure to effect such Company Adverse Recommendation Change, would be reasonably likely to result in a violation of its fiduciary duties under applicable Law, (ii) the Company Board shall notify Parent, in writing, at least five (5) Business Days before taking such action of its intention to take such action and a reasonable description of the event or circumstances giving rise to its determination and (iii) at the end of such notice period, the Company Board takes into account any amendment or modification to this Agreement proposed by Parent and determines in good faith, after consulting with outside legal counsel and its other professional advisors, that the failure to effect such Company Adverse Recommendation Change, would, nevertheless, be reasonably likely to result in a violation of its fiduciary duties under applicable Law.

- 7.11 Section 16 Matters. Prior to the Effective Time, each of Parent, the Company and SpinCo shall take all such steps as may be required (to the extent permitted by applicable Law) to cause any dispositions of SpinCo Common Stock or acquisitions of Parent Common Stock resulting from the Transactions, including the Distribution, directly or indirectly, by each individual, if any, who is subject to Section 16(a) of the Exchange Act with respect to Parent or SpinCo, as applicable, as an officer or director thereof to be exempt under Rule 16b-3 promulgated under the Exchange Act, such steps to be taken in accordance with (and to the extent permitted by) applicable SEC rules and regulations and interpretations of the SEC staff.
- 7.12 <u>Control of Other Party's Business</u>. Nothing contained in this Agreement shall give the Company or SpinCo, directly or indirectly, the right to control or direct Parent or its Subsidiaries' operations prior to the Effective Time. Nothing contained in this Agreement shall give Parent, directly or indirectly, the right to control or direct the operations of the Company, including the SpinCo Business, prior to the Effective Time. Prior to the Effective Time, each of the Company, SpinCo and Parent shall exercise, consistent with the terms and conditions of this Agreement, complete control and supervision over its respective operations.
- 7.13 SpinCo Share Issuance. Prior to the Effective Time, SpinCo will take all actions necessary to authorize the issuance of a number of, or stock split of, shares of SpinCo Common Stock such that the total number of shares of SpinCo Common Stock outstanding immediately prior to the Effective Time will equal the number of shares of Company Common Stock entitled to receive the Distribution outstanding immediately prior to the Effective Time in accordance with the terms of the Separation and Distribution Agreement. Each of the Company and SpinCo shall effect such amendments, filings or other actions with respect to its respective Governing Documents as are necessary to effect the Distribution in accordance with the terms of this Agreement and the Separation and Distribution Agreement.
- 7.14 <u>Transaction Documents</u>. Parent shall, or shall cause its applicable Subsidiaries to, execute and deliver to the Company at or prior to the Closing each of the Transaction Documents to which it or any such Subsidiary is or will be a party at the Effective Time. The Company shall, or shall cause its applicable Subsidiaries to, including SpinCo, execute and deliver to Parent at or prior to the Closing each of the Transaction Documents to which it or any such Subsidiary is or will be a party at the Effective Time.
 - 7.15 NYSE American Listing.

- (a) Parent shall prepare and submit to NYSE American a listing application, if required under NYSE American rules, covering the shares of Parent Common Stock issuable in the Merger, and shall use its reasonable best efforts to cause the shares of Parent Common Stock issuable pursuant to the Merger to be approved for listing on the NYSE American, subject to official notice of issuance, as promptly as practicable after the date of this Agreement, and in any event prior to the Effective Time.
- (b) Parent will use commercially reasonable efforts to (i) maintain the listing and trading of Parent Common Stock on the NYSE American, for so long as Parent qualifies for such listing under the rules and regulations of the NYSE American, and (ii) comply in all material respects with Parent's reporting, filing, and other obligations under the rules an regulations of the NYSE American. In the event that Parent Common Stock is no longer eligible for listing and trading on the NYSE American, Parent will use commercially reasonable efforts to secure the listing or quotation of Parent Common Stock on a Nasdaq market or the American Stock Exchange (if such listing is permitted by the bylaws, rules or regulations of any of the foregoing) and to comply in all material respects with Parent's reporting, filing and other obligations under the bylaws or rules of such exchanges.
- 7.16 <u>Takeover Statutes</u>. If any "fair price," "moratorium," "control share acquisition," "business combination" or other form of antitakeover Law shall become applicable to the Transactions, Parent, Merger Sub and their respective boards of directors shall use all reasonable efforts to grant such approvals and take such actions as are reasonably necessary so that the Transactions may be consummated as promptly as practicable on the terms contemplated hereby and otherwise act to eliminate or minimize the effects of such statute or regulation on the Transactions.
- 7.17 Obligations of Merger Sub and SpinCo. Parent shall take all action necessary to cause Merger Sub to perform its obligations and take any actions contemplated or required under this Agreement or the Transaction Documents or to consummate the Transactions, including the Merger, upon the terms and subject to the conditions set forth in this Agreement. The Company shall take all action necessary prior to the Effective Time to cause SpinCo to perform its obligations and to take any actions contemplated or required to be taken by SpinCo under this Agreement or the Transaction Documents to consummate the Transactions, including the Merger, upon the terms and subject to the conditions set forth in this Agreement and the Separation and Distribution Agreement.

7.18 Further Assurances.

(a) Except as otherwise expressly provided in this Agreement, the Parties shall, and shall cause their respective Affiliates to, use their respective commercially reasonable efforts to take, or cause to be taken, all appropriate action, to do, or cause to be done, and to assist and cooperate with the other Parties in doing, all things necessary, proper or advisable under this Agreement, any other Transaction Document or applicable Law as may be required to carry out the provisions of this Agreement or such other Transaction Document, as applicable, and to consummate and make effective the Merger and the other Transactions (other than with respect to the matters covered in Section 7.5, which shall be governed by the provisions of

Section 7.5 and any consents required in connection with the Separation, which shall solely be governed by the Separation and Distribution Agreement). In furtherance and not in limitation of the foregoing, each Party shall use commercially reasonable efforts to obtain all consents, approvals or waivers from third parties necessary in connection with the Merger (other than with respect to the matters covered in Section 7.5, which shall be governed by the provisions of Section 7.5 and any consents required in connection with the Separation, which shall solely be governed by the Separation and Distribution Agreement); provided, that no Party or any of its Affiliates shall be required to commence any litigation, offer or pay any money, or otherwise grant any accommodation (financial or otherwise) to any third party with respect to the foregoing. The failure to obtain any consents, approvals or waivers from third parties shall not in and of itself constitute a breach of this Agreement.

- (b) Parent will use good faith, commercially reasonable efforts (which efforts does not require any payments from Parent) to assist the Company in obtaining the consent set forth in Section 2.3(a)(v) of the Company Disclosure Schedule.
- (c) During the Interim Period, the Company and Parent will cooperate with each other to identify assets that are necessary or desirable to be included among the SpinCo Assets. Notwithstanding the list of SpinCo Assets set forth in the Separation and Distribution Agreement, if, during the Interim Period, Parent or the Company identify any asset of the Company or its Affiliates that (i) is not listed among the SpinCo Assets in the Separation and Distribution Agreement, and (ii) is necessary or desirable in connection with the operation of the SpinCo Business, then Parent, Company, and SpinCo will take appropriate measures to amend the Separation and Distribution Agreement to include such asset among the SpinCo Assets, but only if Parent agrees, in its absolute discretion, to include such asset among the SpinCo Assets.
- 7.19 <u>Financial Statements</u>. During the Interim Period and from and after the Closing, the Company shall use reasonable best efforts, in connection with the filing of the Registration Statement and any other applicable SEC filings, to (i) cooperate with Parent to prepare pro forma financial statements that comply with the rules and regulations of the SEC to the extent required for SEC filings, including the requirements of Regulation S-X, and (ii) provide and make reasonably available upon reasonable notice, the senior management employees of the Company to discuss the materials prepared and delivered pursuant to this <u>Section 7.19</u>.
- 7.20 <u>D&O Indemnification and Insurance</u>. For a period of six (6) years from and after the Effective Time, the Company shall indemnify and hold harmless each present and former director, officer or employee of SpinCo against any costs or expenses (including reasonable attorneys' fees), judgements, fines, losses, claims, damages or liabilities incurred in connection with any claims, action, suit, proceeding or investigation, whether civil, criminal, administrative or investigative, arising out of or pertaining to any matters existing or occurring at or prior to the Effective Time, whether asserted or claimed prior to, at or after the Effective Time, to the fullest extent that the Company or any of its Subsidiaries (including the SpinCo), as the case may be, would have been permitted under the Governing Documents of SpinCo or the Company as in effect on the date hereof to indemnify such Person (including promptly advancing expenses as incurred to the fullest extent permitted under such Governing Documents). Without limiting the

foregoing, the Company shall (i) maintain for a period of not less than six (6) years from the Effective Time provisions in its Governing Documents concerning the indemnification and exculpation (including provisions relating to expense advancement) of SpinCo's respective former and current officers, directors or employees that are no less favorable to those Persons than the provisions of the Governing Documents of the Company as of the date hereof and (ii) not to amend, repeal, waive or otherwise modify such provisions in any respect that would adversely affect the rights of those Persons thereunder, in each case, except as required by applicable Law.

- 7.21 Contribution of Surviving Corporation to LGM Enterprises. Promptly following the consummation of the Merger, Parent shall (i) (A) cause the Surviving Corporation to convert to a limited liability company, and (B) at least one day following the conversion of the Surviving Corporation to a limited liability company, Parent shall contribute all of the equity of the Surviving Corporation to LGM Enterprises, LLC in exchange for the issuance of a number of Common Units of LGM Enterprises, LLC (as defined in LGM Operating Agreement) equal to the Merger Consideration Shares such that the number of Common Units of LGM Enterprises, LLC held by Parent shall maintain a one-to-one ratio to the number of shares of issued and outstanding Parent Common Units of LGM Enterprises, LLC held by Parent shall maintain a one-to-one ratio to the number of shares of issued and outstanding Parent Common Stock pursuant to Section 4.12 of the LGM Operating Agreement.
- 7.22 <u>Company Financing</u>. The Company will use good faith, commercially reasonable efforts to maximize the amount of Cash included in the SpinCo Assets.

Article VIII

CONDITIONS TO THE MERGER

- 8.1 <u>Conditions to the Obligations of SpinCo, the Company, Parent and Merger Sub to Effect the Merger</u>. The respective obligations of each Party to consummate the Merger shall be subject to the fulfillment (or, to the extent permitted by applicable Law, waiver by the Company and Parent) at or prior to the Closing of the following conditions:
- (a) (i) all Requisite Regulatory Approvals shall have been obtained and shall remain in full force and effect and all statutory waiting periods (and any extensions thereof) in respect thereof shall have expired or been terminated; and (ii) there shall not be in effect any voluntary agreement between the Parent or the Company and any Governmental Authority pursuant to which Parent or the Company has agreed not to consummate the Transactions for any period of time;
- (b) the Separation and the Distribution and the other transactions contemplated by the Separation and Distribution Agreement shall have been consummated in accordance with the Separation and Distribution Agreement;

liii

- (c) (i) the Registration Statement shall have become effective in accordance with the Securities Act or the Exchange Act, as applicable, and shall not be the subject of any stop order by the SEC or actual or threatened proceedings by a Governmental Authority seeking such a stop order; and (ii) the applicable notice periods required by applicable stock exchange rules or securities Laws in connection with the Distribution, if any, shall have expired;
 - (d) each of the Company Shareholder Approval and SpinCo Shareholder Approval shall have been obtained;
- (e) no Governmental Authority of competent jurisdiction shall have enacted, issued or granted any Law (whether temporary, preliminary or permanent), in each case that is in effect and which has the effect of restraining, enjoining or prohibiting the consummation of the Transactions; and
- (f) the shares of Parent Common Stock issuable pursuant to the Transactions shall have been approved for listing on NYSE American, subject to official notice of issuance.
- 8.2 <u>Additional Conditions to the Obligations of the Company and SpinCo</u>. The obligation of the Company and SpinCo to consummate the Merger shall be subject to the fulfillment (or, to the extent permitted by applicable Law, waiver by the Company) at or prior to the Closing of the following additional conditions:
- (a) Parent and Merger Sub shall each have performed and complied in all material respects with the obligations, covenants and agreements required by this Agreement to be performed or complied with by it at or prior to the Effective Time;
- (b) all representations and warranties made by Parent and Merger Sub set forth in Article VI (other than the representations and warranties referenced in the second and third sentences of this Section 8.2(b)), without giving effect to materiality, Parent Material Adverse Effect or similar qualifications, shall be true and correct in all respects at and as of the date hereof and as of the Closing Date as though such representations and warranties were made at and as of the Closing Date (except in the case of any representation or warranty that by its terms addresses matters only as of another specified date, which shall be so true and correct only as of such specified date), except to the extent the failure of such representations and warranties to be true and correct (without giving effect to materiality, Parent Material Adverse Effect or similar qualifications) would not have, individually or in the aggregate, a Parent Material Adverse Effect. The representations and warranties made by Parent set forth in Section 6.1(a), the first and third sentences of Section 6.1(b), the entirety of Section 6.2, Section 6.3 (other than the first two sentences of Section 6.3(a) and Section 6.7 shall be true and correct in all material respects at and as of the date hereof and as of the Closing Date as though such representations and warranties were made at and as of the Closing Date (except in the case of any representation or warranty that by its terms addresses matters only as of another specified date, which shall be so true and correct only as of such specified date). The representations and warranties made by Parent set forth in the first two (2) sentences of Section 6.3(a) shall be true and correct in all respects at and as of the date hereof and as of the Closing Date (other than for de minimis deviations in

the case of the representations and warranties set forth in the first two (2) sentences of <u>Section 6.3(a)</u>, and except in the case of any representation or warranty that by its terms addresses matters only as of another specified date, which shall be so true and correct only as of such specified date);

- (c) Parent shall have delivered to the Company the certificate referenced in <u>Section 2.3(b)(ii)</u> dated as of the Closing Date signed by an authorized officer of Parent certifying that each of the conditions set forth in <u>Section 8.2(a)</u>, (b) and (d) have been satisfied;
- (d) Parent and Merger Sub shall have executed and delivered the applicable Transaction Documents, and to the extent applicable, performed and complied with the obligations, covenants and agreements thereunder required to be performed by them prior to the Effective Time in all material respects, and each such agreement shall be in full force and effect;
- (e) No Parent Material Adverse Effect shall have occurred between the date of this Agreement and the Closing Date that is continuing; and
- (f) The Company must have executed an amendment to the Ionic Agreement pursuant to which the Company and Ionic expand their financing relationship by having the Company issue Ionic (i) additional shares of the Company's series B convertible preferred stock and (ii) a warrant to purchase additional shares of the Company's series B convertible preferred stock.
- 8.3 <u>Additional Conditions to the Obligations of Parent and Merger Sub</u>. The obligation of Parent and Merger Sub to consummate the Merger shall be subject to the fulfillment (or, to the extent permitted by applicable Law, waiver by Parent) at or prior to the Closing of the following additional conditions:
- (a) Each of SpinCo and the Company shall each have performed and complied in all material respects with the obligations, covenants and agreements required by this Agreement to be performed or complied with by it at or prior to the Effective Time;
- (b) all representations and warranties made by the Company set forth in Article IV and Article V (other than the representations and warranties referenced in the second and third sentences of this Section 8.3(b)), without giving effect to materiality, "Company Material Adverse Effect", "SpinCo Material Adverse Effect" or similar qualifications, shall be true and correct in all respects at and as of the date hereof and as of the Closing Date as though such representations and warranties were made at and as of the Closing Date (except in the case of any representation or warranty that by its terms addresses matters only as of another specified date, which shall be so true and correct only as of such specified date), except to the extent the failure of such representations and warranties to be true and correct (without giving effect to materiality, "Company Material Adverse Effect", "SpinCo Material Adverse Effect" or similar qualifications) would not have, individually or in the aggregate, a SpinCo Material Adverse Effect, solely with respect to the representations and warranties set forth in Article V, or Company Material Adverse Effect, solely with respect to the representations and warranties set forth in Article IV. The representations and warranties made by the Company set forth in the first

three (3) sentences of Section 4.1, Section 4.2, Section 4.6, Section 4.7, Section 5.1, Section 5.2, Section 5.3 (other than the first sentence of Section 5.3(a)), and Section 5.19 shall be true and correct in all material respects at and as of the date hereof and as of the Closing Date as though such representations and warranties were made at and as of the Closing Date (except in the case of any representation or warranty that by its terms addresses matters only as of another specified date, which shall be so true and correct only as of such specified date). The representations and warranties made by the Company set forth in the first sentence of Section 5.3(a) and the entirety of each of Section 5.3(b) and Section 5.21 shall be true and correct in all respects at and as of the date hereof and as of the Closing Date as though such representations and warranties were made at and as of the Closing Date (other than for *de minimis* deviations in the case of the representations and warranties set forth in the first sentence of Section 5.3(a), and except in the case of other than in the case of any representation or warranty that by its terms addresses matters only as of another specified date, which shall be so true and correct only as of such specified date);

- (c) No SpinCo Material Adverse Effect or Company Material Adverse Effect shall have occurred between the date of this Agreement and the Closing Date that is continuing;
- (d) The Company shall have delivered to Parent the certificate referenced in <u>Section 2.3(a)(i)</u> dated as of the Closing Date signed by an authorized officer of the Company certifying that each of the conditions set forth in <u>Section 8.1(b)</u> and <u>Section 8.3(a)</u>, (b), (c), (e) and (f) have been satisfied;
- (e) SpinCo and the Company (or such other applicable Subsidiary of the Company) shall have executed and delivered each of the applicable Transaction Documents, and to the extent applicable, performed and complied with the obligations, covenants and agreements to be performed thereunder by them prior to the Effective Time in all material respects, and each such agreement shall be in full force and effect;
- (f) As of the Closing Date and after payment in full of any Unpaid Transaction Expenses, SpinCo must have Net Cash of at least twelve million dollars (\$12,000,000);
- (g) The Company shall have, or shall have caused SpinCo to have, delivered to Parent the certificates, consents and other documents described in Section 2.3(a);
- (h) Each of the Stockholder Support Agreements delivered concurrently with the execution and delivery of this Agreement shall be in full force and effect as of the Closing and shall not have been rescinded or otherwise terminated; and
- (i) Each officer and director of SpinCo shall resign, effective as of the Effective Time, from each of his or her positions as an officer or director of SpinCo.
 - (j) The board of directors of Parent must not have made an Adverse Determination.

(k) The Company must have executed an amendment to the Ionic Agreement (which shall be in full force and effect as of the Closing and shall not have been rescinded or otherwise terminated), pursuant to which the Company and Ionic expand their financing relationship by having the Company issue Ionic (i) additional shares of the Company's series B convertible preferred stock and (ii) a warrant to purchase additional shares of the Company's series B convertible preferred stock.

Article IX

INDEMNIFICATION

- 9.1 Survival of Representations, Warranties and Covenants. The obligations, covenants and agreements that by their terms are to be performed following the Closing pursuant to any Transaction Document, including the Separation and Distribution Agreement, or this Agreement shall survive the Effective Time in accordance with their terms and all other obligations, covenants and agreements herein and therein shall terminate and shall not survive the Closing, except with respect to any such obligations, covenants and agreements made by the Company, which shall survive the Closing until fully performed in accordance with their terms. None of the representations or warranties in this Agreement or in any certificate or instrument delivered pursuant to this Agreement shall survive the Effective Time, except that the representations and warranties of the Company contained in this Agreement, any Transaction Document or in any certificate or instrument delivered pursuant to this Agreement or any Transaction Document shall survive the Closing though and including the date that is twelve (12) months following the Closing Date. The period of time a representation, warranty, covenant, agreement or obligation survives the Closing pursuant to this Section 9.1 shall be the "Survival Period" with respect to such representation, warranty, covenant, agreement or obligation. In the event that notice of any claims for indemnification under this Article 9 shall have been given within the applicable Survival Period and such claim has not been finally resolved by the expiration of such Survival Period, the representations, warranties, covenants, agreement or obligations that are the subject of such claim shall survive for purposes of resolving such claim until such matters are finally resolved.
- 9.2 <u>Indemnification by Company</u>. From and after the Closing, the Company shall indemnify and hold the Parent Indemnitees harmless from and against, and pay to the applicable Parent Indemnitees the amount of any and all Losses asserted against, suffered by, imposed upon, or incurred by such Parent Indemnitee based upon, relating to or arising from:
- (a) any breach or inaccuracy of any representation or warranty in <u>Article IV</u> or <u>Article V</u>, other than any Fundamental Representation, as of the date hereof or as of the Closing as if made at and as of the Closing;
- (b) any breach or inaccuracy of any Fundamental Representation as of the date hereof or as of the Closing as if made at and as of the Closing;
 - (c) any breach or default of any of the covenants, agreements or obligations made herein by the Company;

lvii

- (d) (i) any breach or non-fulfillment of any covenant or agreement made by the Company related to Tax matters, (ii) any inaccuracy in, misrepresentation of, or breach of the representations and warranties of the Company contained in Section 4.9 or Section 5.12, (iii) all Taxes of the Company; (iv) all Taxes imposed upon SpinCo or the ownership or operation of the SpinCo Assets or SpinCo Business with respect to any periods before and through the end of the day of the Effective Time, including, without limitation, in the case of Taxes with respect to any Taxable Period that includes but does not end on the date of the Effective Time, all Taxes that are allocable to the portion of such taxable year or period including the day of the Effective Time, (v) all Taxes of any member of an affiliated, consolidated, combined, or unitary group of which SpinCo is or was a member on or before the day of the Effective Time, including Taxes pursuant to Treasury regulations Section 1.1502-6 (or any corresponding or similar provision of foreign, state, or local Tax Law, (vi) all Taxes of any Person other than SpinCo that are imposed on SpinCo as a transferee or successor, by Contract or agreement, pursuant to any Law, and (vii) all Losses arising, directly or indirectly, from or in connection with any of the foregoing clauses (i) through (vi);
 - (e) any Jet.AI Retained Liabilities;
- (f) any Taxes related to, or arising from, the Separation and Distribution Agreement and the transactions contemplated therein.
 - 9.3 <u>Claims for Indemnification</u>. The procedures for indemnification under this Agreement will be as follows:
- (a) The Parent Indemnitee seeking indemnification hereunder (a "Claimant") will promptly give notice to the Company of any demand, suit, assertion of liability or claim. If the claim relates to an action, suit or proceeding filed by another Person against the Claimant (a "Third-Party Claim"), then such notice will be given by the Claimant within twenty (20) Business Days after written notice of such action, suit or proceeding is given to the Claimant and will include true, correct and complete copies of all suit, service and claim documents. The failure or delay of the Claimant to provide any such notice or deliver such copies will not release the Company from any of its obligations under this Article IX unless (and then solely to the extent that) the Company is materially prejudiced thereby.
- (b) With respect to claims solely between a Parent Indemnitee and the Company, the notice by the Claimant shall describe the claim in reasonable detail, shall include copies of all material written evidence thereof and shall indicate the estimated amount, if reasonably practicable, of the Loss that has been or may be sustained by the Claimant. Following receipt of notice from the Claimant of a claim, the Company will have thirty (30) days to make such investigation of the claim as the Company reasonably deems necessary or desirable, and the Claimant agrees to make available to the Company and its authorized Representatives all information relevant and necessary to substantiate the claim, except to the extent any attorney-client privilege would thereby be vitiated. If the Claimant and the Company agree at or prior to the expiration of such thirty (30)-day period to the validity and amount of such claim, then the Company will promptly pay to the Claimant the full amount of the claim, subject to the terms and limitations hereof. If the Claimant and the Company do not reach any

such agreement within such thirty (30)-day period, then the Claimant may seek an appropriate remedy at law or in equity, as applicable, subject to the terms and limitations hereof. If the Company fails to respond to the Claimant within such thirty (30)-day period then the Company shall be deemed to have agreed to the full amount of the claim.

(c) Third-Party Claims.

- (i) With respect to any Third-Party Claim, the Company will be entitled to assume and maintain control of the defense and settlement of such Third-Party Claim if each of the following conditions (collected referred to as the "*Conditions*") are satisfied: (A) the Third-Party Claim seeks solely monetary damages, (B) the Company reasonably demonstrates that it has the financial resources necessary to defend the matter and fulfill its indemnity obligations, (C) the Third-Party Claim does not involve an Action by a Governmental Authority against Claimant and (D) the Company covenants to and will defend the Third-Party Claim actively, diligently and in good faith.
- (ii) If the Company satisfies the Conditions, the Claimant will notify the Company that it has the right to assume and maintain control of the defense and settlement of such Third-Party Claim. The Company may exercise its right within thirty (30) days after receipt of such notice. If the Company exercises its right to assume and maintain control of the defense and settlement of the Third-Party Claim, it must use counsel reasonably satisfactory to the Claimant, which counsel will be at the sole expense of the Company. Additionally, if the Company exercises such right to assume such control, then the Claimant will be entitled to participate in the defense, compromise or settlement of such Third-Party Claim and to employ counsel at its own expense to assist in the handling of such Third-Party Claim. In such event, counsel selected by the Company will be required to reasonably cooperate with such counsel of the Claimant in such defense, compromise or settlement.
- (iii) If the Company does not assume the defense of the Third-Party Claim, does not comply with the foregoing provisions of this Section 9.3(c), or is not entitled to assume such defense, then the Claimant will be entitled to assume and control such defense.
- (iv) If the Company is permitted and has assumed and is maintaining the defense of any such claim or demand as set forth above, then it will have the power and authority to settle or consent to the entry of judgment of such Third-Party Claim without the consent of the Claimant only if (i) the judgment or settlement results solely in the payment by the Company of the full amount of money damages, and (ii) the judgment or settlement includes a full and complete release of the Claimant from any and all liability thereunder. In all other events, the Company will not consent to the entry of judgment or enter into any settlement without the prior written consent of the Claimant. A Parent Indemnitee will not, without the prior written consent of the Company (which consent shall not be unreasonably withheld, conditioned or delayed), settle or compromise or consent to the entry of any judgment with respect to any pending or threatened Action for which indemnification is sought hereunder.

9.4 <u>Limitations on Indemnification</u>.

- (a) Except for Losses arising as a result of Actual Fraud or Willful Breach, as to which the limitations of this Section 9.4(a) shall not apply, the Company shall not have any indemnification obligations under Section 9.2(a) unless and until the claims asserted against the applicable Indemnitor exceed \$50,000.00 in the aggregate (the "Indemnification Threshold"), at which time the applicable Parent Indemnitee shall be entitled to recover the full amount of Losses from the first dollar.
- (b) Except for Losses arising as a result of Actual Fraud or Willful Breach, as to which the limitations of this <u>Section 9.4(b)</u> shall not apply, the Company shall not have any indemnification obligations under <u>Section 9.2(a)</u> in excess of 20% of the Initial Purchase Price.
- (c) No Party shall be liable for any exemplary or punitive Losses (except to the extent actually awarded to a third party as part of a third-party claim).
- 9.5 <u>Materiality</u>. For purposes of determining the existence of any breach or inaccuracy of any representation, warranty, covenant or agreement set forth in this Agreement, or the amount of any Losses in respect thereof, any materiality, Company Material Adverse Effect or SpinCo Material Adverse Effect standard contained therein shall be disregarded.

Article X

TERMINATION

- 10.1 <u>Termination</u>. This Agreement may be terminated and the Transactions may be abandoned at any time prior to the Effective Time:
 - (a) by mutual written agreement of the Company and Parent;
- (b) by the Company or Parent, if the Closing shall not have occurred on or prior to June 30, 2025 (the "Outside Date"); provided, that the right to terminate this Agreement pursuant to this Section 10.1(b) shall not be available to any Party whose action or failure to comply with its obligations under this Agreement or any of the other Transaction Documents has been the primary cause of, or has primarily resulted in, the failure of the Closing to occur on or prior to such date;
- (c) by the Company or Parent, if any Law shall have been promulgated, entered, enforced, enacted or issued and in effect or shall have been deemed to be applicable to the Merger or the other Transactions, including the Distribution, by any Governmental Authority of competent jurisdiction which permanently prohibits, restrains or makes illegal the consummation of the Merger or the other Transactions, and such Law shall have become final and non-appealable; <u>provided</u>, that the right to terminate this Agreement pursuant to this <u>Section 10.1(c)</u> shall not be available to any Party whose action or failure to perform any of its obligations under this Agreement or any of the Transaction Documents is the primary cause of, or primarily resulted in, the enactment or issuance of any such Law;

- (d) by Parent upon written notice to the Company, in the event of a breach of any representation, warranty, covenant or agreement on the part of the Company or SpinCo, such that the conditions specified in Section 8.3(a) or Section 8.3(b) would not be satisfied at the Closing, and which, (i) with respect to any such breach that is capable of being cured, is not cured by the Company or SpinCo by the earlier of: (x) fifteen (15) days after receipt of written notice thereof; or (y) the Outside Date, or (ii) is incapable of being cured prior to the Outside Date; provided, that Parent shall not have the right to terminate this Agreement pursuant to this Section 10.1(d) if Parent or Merger Sub is then in breach of any of its representations, warranties, covenants or agreements set forth in this Agreement to the extent such breach or breaches would give rise to the failure of a condition set forth in Section 8.2(a) or Section 8.2(b);
- (e) by the Company upon written notice to Parent, in the event of a breach of any representation, warranty, covenant or agreement contained in this Agreement on the part of Parent or Merger Sub such that the conditions specified in Section 8.2(a) or Section 8.2(b) would not be satisfied at the Closing, and which, (i) with respect to any such breach that is capable of being cured, is not cured by Parent by the earlier of: (x) fifteen (15) days after receipt of written notice thereof; or (y) the Outside Date, or (ii) is incapable of being cured prior to the Outside Date; provided, that the Company shall not have the right to terminate this Agreement pursuant to this Section 10.1(e) if the Company or SpinCo is then in breach of any of its representations, warranties, covenants or agreements set forth in this Agreement to the extent such breach or breaches would give rise to the failure of a condition set forth in Section 8.3(a) or Section 8.3(b);
- (f) by the Company or Parent, if the Company Shareholder Approval shall not have been obtained, or by the Parent if the SpinCo Shareholder Approval shall not have been obtained; and
- (g) By the Company prior to the Registration Statement being declared effective by the SEC, in order to accept a Superior Proposal and enter into, immediately following such termination, a binding and definitive written Contract with respect to such Superior Proposal, provided, that (x) the Company has complied with its covenants and agreements under Section 7.10 and (y) the Company pays the Parent Expense Reimbursement Amount to Parent in accordance with Section 10.3.
- 10.2 <u>Effect of Termination</u>. In the event of termination of this Agreement pursuant to <u>Section 10.1</u>, this Agreement shall forthwith become null and void and have no effect, without any Liability on the part of any Party except as set forth in <u>Section 10.3</u>; <u>provided</u>, that no such termination shall relieve any Party of any liability or damages resulting from Actual Fraud or Willful Breach; <u>provided</u>, <u>further</u>, that this <u>Section 10.3</u>, <u>Section 10.4</u> and <u>Article XI</u> hereof shall survive any termination of this Agreement.

10.3 Parent Expense Reimbursement Amount.

(a) In the event that this Agreement is terminated (i) by Parent pursuant to <u>Section 10.1(d)</u>, or (ii) by Parent or the Company pursuant to <u>Section 10.1(f)</u> or <u>Section 10.1(g)</u>, the Company shall pay to Parent, the Parent Expenses (the "*Parent Expenses Reimbursement Amount*") by wire transfer of immediately available funds to an account designated by Parent in

writing no later than one (1) Business Day after the date on which Parent delivers to the Company a written invoice for the Parent Expenses. As used herein, "*Parent Expenses*" means the amount of all reasonable and documented out-of-pocket fees and expenses incurred or paid by Parent and its Affiliates in connection with this Agreement and the Transactions, including fees and expenses of law firms, accounting firms, financial advisors, outside experts, third-party services providers, and consultants.

- (b) In addition to payment of the Parent Expenses Reimbursement Amount, in the event that this Agreement is terminated by the Company pursuant to <u>Section 10.1(g)</u>, concurrently with such termination, the Company shall also be obligated to pay Parent a termination fee of \$650,000 (the "*Parent Termination Fee*"), by wire transfer of immediately available funds to an account designated by Parent in writing.
- (c) If the Company fails to pay promptly the Parent Expenses Reimbursement Amount or the Parent Termination Fee, as applicable, and in order to obtain such payment, the Parent commences an Action that results in a judgment against the Company for any amount owed thereby under this Section 10.3, the Company shall reimburse Parent for its reasonable and documented costs and expenses (including reasonable and documented attorneys' fees) in connection with such Action, together with interest on such amount at a rate equal to (x) the prime rate as published in *The Wall Street Journal* in effect on the date such payment was required to be made through the date such payment was actually received, plus (y) five percent (5%).
- (d) Each of the Parties acknowledges that (i) the agreements contained in this <u>Section 10.3</u> are an integral part of the Transactions, (ii) without these agreements, the Parties would not enter into this Agreement and (iii) the payment of the Parent Expenses Reimbursement Amount and Parent Termination Fee does not constitute a penalty, but rather is liquidated damages in a reasonable amount that will compensate Parent for the efforts and resources expended and opportunities foregone while negotiating this Agreement and in reliance on this Agreement and on the expectation of the consummation of the Transactions, which amount would otherwise be impossible to calculate with precision.
- 10.4 <u>Fees and Expenses</u>. Except as otherwise provided in the Separation and Distribution Agreement or this Agreement, including this <u>Section 10.4</u>, and, if applicable, payment of the Parent Expenses Reimbursement Amount pursuant to <u>Section 10.3</u>, all fees and expenses incurred by the Parties shall be borne solely by the Party that has incurred such fees and expenses, whether or not the Merger is consummated.

Article XI

MISCELLANEOUS

11.1 <u>Governing Law; Jurisdiction</u>. This Agreement, and all claims, disputes, controversies or causes of action (whether in contract, tort, equity or otherwise) that may be based upon, arise out of or relate to this Agreement (including any schedule or exhibit hereto) or the negotiation, execution or performance of this Agreement (including any claim, dispute,

lxii

controversy or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), shall be governed by and construed in accordance with the Laws of the State of Delaware, without regard to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than the State of Delaware. Each of the Parties agrees that any Action related to this agreement shall be brought exclusively in the Court of Chancery of the State of Delaware or, if under applicable Law, exclusive iurisdiction over such matter is vested in the federal courts, any federal court in the State of Delaware and any appellate court from any thereof (the "Chosen Courts"). By executing and delivering this Agreement, each of the Parties irrevocably: (i) accepts generally and unconditionally submits to the exclusive jurisdiction of the Chosen Courts for any Action relating to this Agreement, including any Action brought for any remedy contemplated by Section 11.7; (ii) waives any objections which such party may now or hereafter have to the laying of venue of any such Action contemplated by this Section 11.1 and hereby further irrevocably waives and agrees not to plead or claim that any such Action has been brought in an inconvenient forum; (iii) agrees that it will not attempt to deny or defeat the personal jurisdiction of the Chosen Courts by motion or other request for leave from any such court; (iv) agrees that it will not bring any Action contemplated by this Section 11.1 in any court other than the Chosen Courts; (v) agrees that service of all process, including the summons and complaint, in any Action may be made by registered or certified mail, return receipt requested, to such party at their respective addresses provided in accordance with Section 11.2 or in any other manner permitted by Law; and (vi) agrees that service as provided in the preceding clause (v) is sufficient to confer personal jurisdiction over such party in the Action, and otherwise constitutes effective and binding service in every respect. Each of the parties hereto agrees that a final judgment in any Action in a Chosen Court as provided above may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law, and each party further agrees to the non-exclusive jurisdiction of the Chosen Courts for the enforcement or execution of any such judgment.

11.2 <u>Notices</u>. All notices and other communications among the Parties shall be in writing and shall be deemed to have been duly given (a) when delivered in person, (b) when delivered after posting in the national mail having been sent registered or certified mail return receipt requested, postage prepaid, (c) when delivered by FedEx or other internationally recognized overnight delivery service or (d) when delivered by facsimile (solely if receipt is confirmed) or email (so long as the sender of such email does not receive an automatic reply from the recipient's email server indicating that the recipient did not receive such email), addressed as follows:

if to the Company or SpinCo, to:

Jet.AI Inc. 10845 Griffith Peak Dr. – Suite 200 Las Vegas, NV 89135 Attn: Mike Winston, Executive Chairman Telephone: []

```
Email: [ ]
       with a copy (which shall not constitute notice) to:
       Dykema Gossett PLLC
       111 E. Kilbourn Ave. - Suite 1050
       Milwaukee, WI 53202
       Attn: Kate Bechen and Andrew Frost
       Telephone: [ ] and [ ]
       Email: [ ] and [ ]
if to Parent, to:
       flyExclusive, Inc.
       2860 Jetport Road
       Kinston, NC 28504
       Attn: Brad Garner, Chief Financial Officer
       Telephone: []
       Email: []
       with a copy (which shall not constitute notice) to:
```

Wyrick Robbins Yates and Ponton LLP 4101 Lake Boone Trail, Suite 300 Raleigh, NC 27607

Attn: Larry E. Robbins and David P. Creekman

Telephone: [] Email: [] and []

or to such other address or addresses as the Parties may from time to time designate in writing by like notice.

- 11.3 <u>Headings</u>. The headings contained in this Agreement are inserted for convenience only and shall not be considered in interpreting or construing any of the provisions contained in this Agreement.
- 11.4 <u>Entire Agreement</u>. This Agreement (including the Exhibits and Schedules hereto) and the Transaction Documents constitute the entire agreement between the Parties with respect to the subject matter hereof and supersede all prior agreements and understandings between the Parties with respect to such subject matter.

11.5 Amendments and Waivers.

(a) Any Party may, at any time prior to the Closing, by action taken by its board of directors, or officers thereunto duly authorized, waive any of the terms or conditions of this Agreement or (without limiting Section 11.5(b)) agree to an amendment or modification to

lxiv

this Agreement by an agreement in writing executed in the same manner (but not necessarily by the same Persons) as this Agreement. No waiver by any of the Parties of any breach hereunder shall be deemed to extend to any prior or subsequent breach hereunder or affect in any way any rights arising by virtue of any prior or subsequent such occurrence. No waiver by any of the Parties of any of the provisions hereof shall be effective unless explicitly set forth in writing and executed by the Party sought to be charged with such waiver.

(b) This Agreement may be amended or modified, in whole or in part, only by a duly authorized agreement in writing executed by the Parties in the same manner (but not necessarily by the same Persons) as this Agreement, and which makes reference to this Agreement.

11.6 <u>Assignment; Parties in Interest; Non-Parties</u>.

- (a) No Party may assign its rights or delegate its duties under this Agreement without the prior written consent of the other Parties. Any attempted assignment or delegation in breach of this <u>Section 11.6</u> shall be null and void. This Agreement shall be binding upon and inure to the benefit of the Parties and their respective permitted successors and assigns. Nothing expressed or implied in this Agreement is intended or shall be construed to confer upon or give any Person, other than the Parties, any rights or remedies under or by reason of this Agreement, except as provided in <u>Section 11.6(b)</u> (which is intended to be for the benefit of the Persons covered thereby and may be enforced by such Persons).
- (b) Notwithstanding anything to the contrary in this Agreement, it is hereby agreed and acknowledged that this Agreement may only be enforced against, and any claims of action that may be based upon, arise out of or relate to this Agreement or the negotiation, execution or performance of this Agreement may only be made against, the Parties hereto, and no former, current or future Affiliates, officers, directors, managers, employees, equityholders, lenders, financing sources, managers, members, partners, agents or representatives of any Party, in each case, who is not a Party to this Agreement, shall have any liability for any obligations of the Parties hereto or for any claim based on, in respect of, or by reason of, the Transactions.

11.7 Specific Performance.

(a) The Parties agree and acknowledge that the failure to perform under this Agreement will cause an actual, immediate and irreparable harm and injury and that the Parties would not have any adequate remedy at law in the event that any of the provisions of this Agreement were not performed in accordance with their specific terms or were otherwise breached. Accordingly, it is agreed that, (i) each of the Parties shall be entitled to an injunction or injunctions to prevent breaches or threatened breaches of this Agreement by any other Party and to specifically enforce the terms and provisions of this Agreement, and (ii) prior to the Closing or any termination of this Agreement in accordance with Section 10.1, damages shall be awarded only in a case where a court of competent jurisdiction shall have determined that, notwithstanding the Parties' intention for specific performance to be the applicable remedy prior to termination or the Closing, such specific performance is not available or otherwise will not be granted as a remedy.

lxv

- (b) The Parties further agree that (i) by seeking the remedies provided for in this Section 11.7, a Party shall not in any respect waive its right to seek any other form of relief that may be available to a party under this Agreement, including monetary damages, subject to the terms hereof, (ii) nothing contained in this Section 11.7 shall require any Party to institute any proceeding for (or limit any Party's right to institute any proceeding for) specific performance under this Section 11.7 before exercising any termination right under Section 10.1 (and pursuing damages after such termination), nor shall the commencement of any Action pursuant to this Section 11.7 or anything contained in this Section 11.7 restrict or limit any Party's right to terminate this Agreement in accordance with the terms of Section 10.1 or to pursue any other remedies under this Agreement that may be available then or thereafter and (iii) no Person shall be required to obtain, furnish or post any bond or similar instrument in connection with or as a condition to obtaining any remedy referred to in this Section 11.7, and each Party irrevocably waives any right it may have to require the obtaining, furnishing or posting of any such bond or similar instrument.
- (c) To the extent either party hereto brings any Action to enforce specifically the performance of the terms and provisions of this Agreement in accordance with this <u>Section 11.7</u>, the Outside Date shall automatically be extended by (i) the amount of time during which such Action is pending, plus twenty (20) Business Days, or (ii) such other time period established by the court presiding over such Action.
- 11.8 WAIVER OF JURY TRIAL. THE PARTIES HEREBY UNCONDITIONALLY AND IRREVOCABLY WAIVE THEIR RIGHT TO TRIAL BY JURY IN ANY JUDICIAL PROCEEDING IN ANY COURT RELATING TO ANY DISPUTE, CONTROVERSY OR CLAIM ARISING OUT OF, RELATING TO OR IN CONNECTION WITH THIS AGREEMENT OR ANY TRANSACTION DOCUMENT (INCLUDING ANY SCHEDULE OR EXHIBIT HERETO AND THERETO) OR THE BREACH, TERMINATION OR VALIDITY OF SUCH AGREEMENTS OR THE NEGOTIATION, EXECUTION OR PERFORMANCE OF SUCH AGREEMENTS. NO PARTY TO THIS AGREEMENT SHALL SEEK A JURY TRIAL IN ANY LAWSUIT, PROCEEDING, COUNTERCLAIM OR ANY OTHER LITIGATION PROCEDURE BASED UPON, OR ARISING OUT OF, THIS AGREEMENT OR ANY RELATED INSTRUMENTS. NO PARTY WILL SEEK TO CONSOLIDATE ANY SUCH ACTION IN WHICH A JURY TRIAL HAS BEEN WAIVED WITH ANY OTHER ACTION IN WHICH A JURY TRIAL CANNOT BE OR HAS NOT BEEN WAIVED. EACH PARTY TO THIS AGREEMENT CERTIFIES THAT IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT OR INSTRUMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS SET FORTH ABOVE IN THIS SECTION 11.8. NO PARTY HAS IN ANY WAY AGREED WITH OR REPRESENTED TO ANY OTHER PARTY THAT THE PROVISIONS OF THIS SECTION 11.8 WILL NOT BE FULLY ENFORCED IN ALL INSTANCES.
- 11.9 <u>Severability</u>. If any provision of this Agreement or any Transaction Document, or the application of any such provision to any Person or circumstance, shall be held invalid, illegal or unenforceable in any respect by a court of competent jurisdiction, such invalidity, illegality or unenforceability shall not affect any other provision hereof. The Parties further agree that if any

lxvi

provision contained herein is, to any extent, held invalid or unenforceable in any respect under the Laws governing this Agreement, they shall take any actions necessary to render the remaining provisions of this Agreement valid and enforceable to the fullest extent permitted by Law and, to the extent necessary, shall amend or otherwise modify this Agreement to replace any provision contained herein that is held invalid or unenforceable with a valid and enforceable provision giving effect to the intent of the Parties.

- 11.10 <u>Counterparts</u>. This Agreement may be executed in two or more counterparts (including by electronic or .pdf transmission), each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Delivery of any signature page by facsimile, electronic or .pdf transmission shall be binding to the same extent as an original signature page.
- 11.11 <u>Disclosure Schedule</u>. The Company Disclosure Schedule, the SpinCo Disclosure Schedule and the Parent Disclosure Schedule (each, a "*Disclosure Schedule*" and, collectively, the "*Disclosure Schedules*") (including, in each case, any section thereof) referenced herein are a part of this Agreement as if fully set forth herein. All references herein to the Company Disclosure Schedule, SpinCo Disclosure Schedule and Parent Disclosure Schedule (including, in each case, any section thereof) shall be deemed references to such parts of this Agreement, unless the context shall otherwise require. Certain information set forth in the Disclosure Schedules is included solely for informational purposes and may not be required to be disclosed pursuant to this Agreement. The disclosure of any information in any Disclosure Schedule shall not be deemed to constitute in itself an acknowledgment that such information is required to be disclosed in connection with this Agreement, nor shall such information be deemed to establish a standard of materiality.

[Signature page follows.]

1	٠	•
XV	1	1

IN WITNESS WHEREOF, the Parties have caused this Agreement and Plan of Merger and Reorganization to be duly executed by their respective authorized officers as of the day and year first above written.

PARENT:

FLYEXCLUSIVE, INC.

By: /s/ Thomas James Segrave, Jr.
Name: Thomas James Segrave, Jr.
Title: Chief Executive Officer

MERGER SUB:

FLYX MERGER SUB, INC.

By: /s/ Thomas James Segrave, Jr.
Name: Thomas James Segrave, Jr.
Title: Chief Executive Officer

[Signature Page to Agreement and Plan of Merger and Reorganization]

IN WITNESS WHEREOF, the Parties have caused this Agreement and Plan of Merger and Reorganization to be duly executed by their respective authorized officers as of the day and year first above written.

COMPANY:

JET.AI INC.

By: /s/ Michael Winston
Name: Michael Winston
Title: Executive Chairman

SPINCO:

JET.AI SPINCO, INC.

By: /s/ Michael Winston
Name: Michael Winston
Title: Executive Chairman

[Signature Page to Agreement and Plan of Merger and Reorganization]

ANNEX A

Defined Terms

Terms with initial capitalized letters used in the Agreement to which this Annex A is attached will have the following respective meanings, and all references to Sections, Schedules or Annexes in the following definitions will refer to Sections, Schedules or Annexes of or to such Agreement:

- "Action" means any claim, action, suit, litigation, arbitration, mediation, inquiry, investigation or other proceeding, in each case, by any Person or Governmental Authority, in each case, before, heard by or otherwise involving any Governmental Authority.
- "Actual Fraud" means, with respect to a Party, an actual and intentional fraud with respect to the making of the representations and warranties pursuant to Article IV, Article V or Article VI (as applicable); provided, that such actual and intentional fraud shall only be deemed to exist if the applicable representations and warranties were, to the Knowledge of the Company or SpinCo, in the case of Article IV or Article \underline{V} , or to the Knowledge of Parent, in the case of Article VI, actually breached when made.
 - "Additional Merger Consideration Shares" has the meaning set forth in Section 3.3(e).
- "Adverse Determination" means a determination by the board of directors of Parent that the Transactions are not in the best interests of the stockholders of Parent.
- "Affiliate" means, with respect to any Person, any other Person that, directly or indirectly, controls, is controlled by, or is under common control with, such Person, through one or more intermediaries or otherwise. For the avoidance of doubt, following the Effective Time, Affiliates of Parent shall include the SpinCo.
- "Agreement" means this Agreement and Plan of Merger and Reorganization, including all Annexes, Exhibits and Schedules hereto, as it may be amended, restated, modified or supplemented from time to time in accordance with its terms.
- "Antitrust Laws" means the Sherman Act, as amended, the Clayton Act, as amended, the Federal Trade Commission Act, as amended, the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended, and the rules and regulations promulgated thereunder, and all other applicable Laws issued by a Governmental Authority that are designed or intended to prohibit, restrict or regulate actions having the purpose or effect of monopolization or restraint of trade or lessening of competition through merger or acquisition.
- "Applicable Premium Percentage" means (i) 110%, if Net Cash (as adjusted by the Adjustment Amount, as applicable) is less than or equal to \$11,999.999.99; (ii) 115%, if Net Cash is equal to or greater than \$12,000,000.00, but less than or equal to \$14,624,999.99; (iii)

117.5%, if Net Cash is equal to or greater than \$14,625,000.00, but less than or equal to \$17,249,999.99, and (iv) 120%, if Net Cash is equal to or greater than \$17,250,000.00.

"Balance Sheet Date" has the meaning set forth in Section 5.8(a).

"Benefit Plan" means any pension, retirement, profit-sharing, deferred compensation, stock option, equity incentive, employee stock ownership, share purchase, severance pay, vacation, bonus, retention, change in control, or other incentive plan, medical, vision, dental or other health plan, any life insurance plan, flexible spending account, cafeteria plan, vacation, holiday, disability or any other benefit or compensation plan or fringe benefit plan, including any "employee benefit plan," as that term is defined in Section 3(3) of Employee Retirement Income Security Act of 1974, as amended (whether or not subject thereto), and any other plan, fund, policy, program, practice or arrangement providing compensation or other benefits, in each case, (i) that is maintained by a professional employer organization for the benefit of current or former employees of SpinCo, (ii) that is sponsored, maintained, contributed to, required to be contributed to or participated in by SpinCo or to which SpinCo is a party or (iii) under or with respect to which SpinCo has any current or contingent liability or obligation.

"Business Day" means any day that is not a Saturday, a Sunday or other day on which banking institutions in New York, New York are authorized or obligated by Law to be closed.

"Cash" means all cash and cash equivalents of the Company as of 11:59 p.m. on the date of the Effective Time, as computed in accordance with GAAP, other than restricted cash. For the avoidance of doubt, Cash will be calculated net of issued but uncleared checks and drafts and inclusive of deposits in transit.

"Code" means the Internal Revenue Code of 1986, as amended.

"Company Adverse Recommendation Change" means the company Board: (i) failing to make, withdrawing, amending, modifying, or materially qualifying, in a manner adverse to Parent, its recommendation that the stockholders of the Company vote in favor of adoption of the Transactions contemplated this Agreement and the Transaction Documents; (b) recommending a Takeover Proposal; (c) failing to recommend against acceptance of any tender offer or exchange offer for the shares of Company Common Stock within ten (10) Business Days after the commencement of such offer; or (d) resolving or agreeing to take any of the foregoing actions.

"Company Common Stock" means the common stock, par value \$0.001 per share, of the Company.

"Company Disclosure Schedule" means the Disclosure Schedule delivered by the Company to Parent on the date hereof and identified as such.

"Company Material Adverse Effect" means any Effect that has, or would reasonably be expected to, individually or in the aggregate, materially impair, materially delay or otherwise have a material adverse effect on the ability of the Company to perform its obligations hereunder or under the other Transaction Documents or to consummate the Transactions, including the

Separation and the Distribution, or prevent it from performing such obligations or consummating such Transactions.

- "Company SEC Documents" means all forms, reports, Schedules, statements and other documents required to be filed or furnished by the Company or SpinCo with the SEC since January 1, 2022.
- "Company Shareholder Approval" means the approval of the Transactions contemplated hereunder, including the Merger and the Separation and Distribution, by an affirmative vote of the number of holders of Company Common Stock required to approve such Transactions under applicable Law and the Governing Documents of the Company.
- "Consent" means any consent, clearance, expiration or termination of a waiting period, approval, exemption, waiver, authorization, filing, registration or notification.
- "Contract" means any legally binding contract, agreement, understanding, arrangement, loan or credit agreement, note, bond, indenture, lease, warranty, purchase order, sublicense or license or other instrument.
 - "DGCL" means Delaware General Corporation Law, as amended.
 - "Distribution Date" has the meaning set forth in the Separation and Distribution Agreement.
 - "Distribution Time" has the meaning set forth in the Separation and Distribution Agreement.
 - "Effect" means any change, event, development, condition, occurrence or effect.
- "Entity" means any Person that is a legal entity; <u>provided</u>, that when used in reference to Parent or SpinCo, "Entity" means Parent or SpinCo, in each case together with its respective Subsidiaries, taken as a whole.
- "Environmental Laws" means any Law relating to pollution or protection of the environment, natural resources, or human health and safety (to the extent related to exposure to Hazardous Materials or hazardous conditions).
 - "Exchange Act" means the Securities Exchange Act of 1934, as amended.
- "Exchange Ratio" means an amount equal to the quotient of (i) the Merger Consideration Shares, divided by (ii) the total amount of shares of SpinCo Common Stock issued and outstanding on a fully-diluted basis as of the day immediately prior to the Effective Date.
- "Fundamental Representation" means any representation or warrant set forth in Section 4.1 (Organization of the Company), Section 4.2 (Due Authorization), Section 4.6 (Brokers' Fees), Section 4.9 (Tax Matters), Section 5.1 (Organization of SpinCo), Section 5.2 (Due Authorization), Section 5.12 (Tax Matters), and Section 5.19 (Brokers' Fees).

"GAAP" means generally accepted accounting principles in the United States.

"Governing Documents" means, with respect to any Person, its certificate of incorporation and bylaws, operating or limited liability company agreement, limited partnership agreement, memorandum and articles of association or similar Governing documents, in each case, as amended.

"Governmental Authority" means any federal, state, local, transnational, supranational or foreign government, any Person exercising executive, legislative, judicial, regulatory or administrative function of or pertaining to government or Law, including any regulatory, self-regulatory or quasi-regulatory authority, agency, commission, body, department or other instrumentality, and any court, arbitral body or tribunal of competent jurisdiction.

"Government Official" means any officer or employee of a Governmental Authority or any department, agency, or instrumentality thereof, including any political subdivision thereof or any corporation or other Person owned or controlled in whole or in part by any Governmental Authority or any sovereign wealth fund, or of a public international organization, or any Person acting in an official capacity for or on behalf of any such government or department, agency, or instrumentality, or for or on behalf of any such public international organization, or any political party, party official, or candidate thereof.

"Hazardous Material" means any toxic, reactive, corrosive, ignitable or flammable chemical or chemical compound, or hazardous or toxic substance, material or waste, or any pollutant or contaminant, whether solid, liquid or gas, or any other substance, material or waste that is subject to regulation, control or remediation or for which liability or standards of care are imposed under any Environmental Law, including petroleum (including crude oil or any fraction thereof), radon, asbestos, radioactive materials, per- and polyfluoroalkyl substances and polychlorinated biphenyls.

"Initial Purchase Price" means an amount equal to the product of (i) the Applicable Premium Percentage, multiplied by (ii) Estimated Net Cash.

"Intellectual Property" means all U.S. and foreign rights, title and interest (whether statutory, common law or otherwise) in or relating to any intellectual property, including all: (i) trademarks, trade dress, service marks, certification marks, logos, slogans, design rights, names, corporate names, trade names, brand names and other similar designations of source or origin, together with the goodwill symbolized by any of the foregoing, and all applications, registrations, renewals and extensions of any of the foregoing (collectively, "Trademarks"); (ii) patents and patent applications, and all related national or international counterparts thereto, including any divisionals, continuations, continuations-in-part, reissues, reexaminations, substitutions provisionals, renewals, extensions, patents of addition, supplementary protection certificates, utility models, inventors' certificates, or the like, and any foreign equivalents of any of the foregoing (including certificates of invention and any applications therefor) and all rights to claim priority from any of the foregoing (collectively, "Patents"); (iii) copyrights and copyrightable subject matter, whether or not registered or published, and all applications, registrations, reversions, extensions and renewals of any of the foregoing, and all moral rights,

however denominated; (iv) trade secrets, and all other confidential or proprietary information, ideas, technology, Software, compositions, discoveries, improvements, know-how, inventions, designs, processes, techniques, formulae, models, and methodologies, in each case, whether or not patentable or copyrightable, but excluding issued Patents (collectively, "*Know-How*," and trade secrets together with confidential Know-How, "*Trade Secrets*"); (v) Internet domain names and social media accounts and addresses, and all registrations for any of the foregoing (collectively, "*Domain Names*"); and (vi) rights and remedies with respect to any past, present, or future infringement, misappropriation, or other violation of any of the foregoing in clauses (i) through (v).

"Interests" means shares, partnership interests, limited liability company interests or any other equity interest in any Person.

"IRS" means the United States Internal Revenue Service.

"IT Assets" means all information technology, Software, computers, computer systems, communication systems, telecommunications equipment, databases, internet protocol addresses, data rights and documentation, reference, resource and training materials relating to any of the foregoing, and all Contracts (including Contract rights) relating to any of the foregoing (including Software license agreements, source code escrow agreements, support and maintenance agreements, electronic database access contracts, Domain Name registration agreements, website hosting agreements, Software or website development agreements, outsourcing agreements, service provider agreements, interconnection agreements, governmental permits, radio licenses and telecommunications agreements).

"Jet.AI Retained Business" has the meaning given to such term in the Separation and Distribution Agreement.

"Jet.AI Retained Liabilities" has the meaning given to such term in the Separation and Distribution Agreement.

"Knowledge" means (a) with respect to the Company, the actual knowledge of the persons set forth in Section 1.1(a) of the Company Disclosure Schedule, after reasonable investigation, (b) with respect to SpinCo, the knowledge of the persons set forth in Section 1.1(a) of the SpinCo Disclosure Schedule, after reasonable investigation and (c) with respect to Parent, the actual knowledge of the persons set forth in Section 1.1(a) of the Parent Disclosure Schedule, after reasonable investigation.

"Law" means, with respect to any Person, any law, statute, code, ordinance, order, decree, award, directive, judgment, ruling, rule, regulation or similar requirement issued, promulgated, enforced or enacted by or under the authority of a Governmental Authority that is binding upon or applicable to such Person.

"Liability" means any liability (whether known or unknown, whether asserted or unasserted, whether absolute or contingent, whether accrued or unaccrued, whether liquidated or unliquidated, whether direct or indirect, and whether due or to become due).

- "Lien" means any mortgage, deed of trust, pledge, hypothecation, encumbrance, easement, encroachment, license, option, right of first refusal, security interest or other lien or restriction of any kind.
- "Losses" means any and all Liabilities, claims, losses, damages, costs, expenses, interest, awards, judgments and penalties (including reasonable attorneys' and consultants' fees and expenses).
- "Merger Consideration Shares" means a number of shares of Parent Common Stock equal to the quotient of (i) the Initial Purchase Price, divided by (ii) Parent Trading Price.
- "Net Cash" means, without double counting, (i) all assets of SpinCo, as of the Measurement Time, including Cash, accounts receivable, other current assets, right-of-use lease assets and deposits (including but not limited to the Textron Deposit, security deposits and collateral related to the SpinCo Assets), net of (ii) SpinCo liabilities, including SpinCo Payables, SpinCo Transaction Expenses, accrued liabilities, deferred revenue, current operating lease liability and lease liability, net of current portion.
 - "Parent Common Stock" means the Class A common stock, par value \$0.0001 per share, of Parent.
- "Parent Disclosure Schedule" means the Disclosure Schedule delivered by Parent to the Company and SpinCo on the date hereof and identified as such.
 - "Parent Expenses" has the meaning set forth in Section 10.3.
 - "Parent Indemnitees" means Parent, the Surviving Corporation, and their respective Affiliates.
- "Parent Material Adverse Effect" means any Effect that has, or would reasonably be expected to, individually or in the aggregate, materially impair, materially delay or otherwise have a material adverse effect on the ability of Parent to perform its obligations hereunder or under the other Transaction Documents or to consummate the Transactions, or prevent it from performing such obligations or consummating such Transactions.
- "Parent SEC Documents" means all forms, reports, schedules, statements and other documents required to be filed or furnished by Parent with the SEC since December 27, 2023.
 - "Parent Share Issuance" means the issuance of shares of Parent Common Stock pursuant to the Merger.
- "Parent Subsidiaries" means all direct and indirect Subsidiaries of Parent. For the avoidance of doubt, following the Effective Time, the Parent Subsidiaries shall include SpinCo.
- "Parent Trading Price" means the share price equal to the volume weighted average closing sale price of one share of Parent Common Stock as reported on NYSE American for a period of 30 consecutive trading days ending on the trading days three trading days prior to the

day of the Effective Date (as adjusted as appropriate to reflect any stock splits, reverse stock splits, stock dividends (including any dividend or distribution of securities convertible into Parent Common Stock), extraordinary cash dividend, reorganization, recapitalization, reclassification, combination, exchange of shares or other like change or transaction with respect to Parent Common Stock).

- "Patent" has the meaning set forth in the definition of "Intellectual Property."
- "Permits" means licenses, franchises, permits, certificates, approvals and authorizations from Governmental Authorities.
- "Permitted Liens" means (a) Liens in respect of liabilities shown or reflected in the balance sheets included in the Financial Statements, and (b) Liens arising by operation of Law for Taxes or other governmental charges not yet due and payable or due but not delinquent or being contested in good faith by appropriate proceedings or for which adequate reserves are maintained in the Financial Statements as of the Closing Date.
- "*Person*" means any individual, firm, corporation, partnership, limited liability company, incorporated or unincorporated association, joint venture, joint stock company, Governmental Authority or other organization or entity of any kind.
- "Personal Information" means all information in any form or media that identifies or could be used to identify an individual person (including any current, prospective, or former customer, end user or employee), in addition to any definition for "personal information" or any similar term provided by applicable Law or by the Company or SpinCo in any of its privacy policies, notices or contracts (e.g., "personal data," "personally identifiable information" or "PII").
- "Pre-Closing Tax" means any Tax for a Tax Period or portion thereof ending on or before the date of the Effective Time, including the portion of any Straddle Period through the end of the day of the date of the Effective Time.
- "Pre-Closing Tax Period" means any Tax Period ending on or before the Closing Date and, with respect to any Straddle Period, the portion of such Straddle Period through the end of the Closing Date.
- "Privacy Laws" means applicable Laws, and self-regulatory guidelines (including of any applicable foreign jurisdiction) relating to the receipt, collection, compilation, use, storage, processing, sharing, safeguarding, security (technical, physical or administrative), disposal, destruction, disclosure or transfer (including cross-border) of any Personal Information, including the Federal Trade Commission Act, California Consumer Privacy Act (CCPA), Payment Card Industry Data Security Standard (PCI-DSS), EU General Data Protection Regulation (GDPR), any and all applicable Laws relating to breach notification and the use of Personal Information for marketing purposes.

- "*Privacy Requirements*" means all applicable Privacy Laws and all of the Company's and SpinCo's policies, notices, and contractual obligations relating to the receipt, collection, compilation, use, storage, processing, sharing, safeguarding, security (technical, physical and administrative), disposal, destruction, disclosure, or transfer (including cross-border) of Personal Information.
- "*Proxy Statement*" means the proxy statement to be mailed to the shareholders of the Company relating to the Company Shareholders Meeting, including any amendments or supplements thereto.
 - "Purchase Price" means an amount equal to the product of (i) the Applicable Premium Percentage, multiplied by (ii) Net Cash.
 - "Purchase Price Adjustment Threshold" means \$50,000.
- "Record Date" means 5:00 p.m. Eastern time on the date determined by the Company Board as the record date for determining stockholders of the Company entitled to receive shares of SpinCo Common Stock in the Distribution.
- "Registration Statement" means the registration statement on Form S-4 to be filed by Parent with the SEC (as amended and supplemented from time to time) to effect the registration under the Securities Act of the issuance of the shares of Parent Common Stock that will be issued to SpinCo Holders as Merger Consideration Shares pursuant to the Merger, as well as Parent's reasonable estimate of the full number shares of Parent Common Stock that might be issued to SpinCo Holders as Additional Merger Consideration Shares pursuant to Section 3.3(e).
 - "Regulatory Authority" means the FAA or any other comparable Governmental Authority.
- "Regulatory Authorizations" means any approvals, clearances, authorizations, registrations, certifications, licenses, exemptions and permits granted by any Regulatory Authority.
- "Release" means any spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, depositing, escaping, leaching, disposing or dumping into the environment.
- "Representative" means, with respect to any Person, such Person's directors, managers, members, officers, employees, agents, partners, attorneys, financial advisors, financing sources, consultants, advisors or other Persons acting on behalf of such Person.
- "Requisite Regulatory Approvals" means all regulatory authorizations, consents, clearances, orders, approvals or expirations of applicable waiting periods set forth on Section 1.1-A of the Parent Disclosure Schedule.
- "Restricted Portion" means, in respect of each SpinCo Holder, an amount equal to the product of (i) such SpinCo Holder's aggregate portion of the Merger Consideration Shares

actually issued to such SpinCo Holder, multiplied by (ii) 0.20, rounded up to the nearest whole share.

- "Restricted Shares" means shares of Parent Common Stock that are subject to forfeiture pursuant to Section 3.3.
- "Restricted Shares Value" means the aggregate value of the Restricted Shares based on the Parent Trading Price.
- "Securities Act" means the Securities Act of 1933, as amended.
- "Separation and Distribution Agreement" means that Separation and Distribution Agreement, dated as of the date hereof, by and among the Company, Parent and SpinCo, attached as Exhibit A to this Agreement.
- "Software" means all: (i) computer programs, including all software implementations of algorithms, models and methodologies, whether in source code, object code, human readable form or other form; (ii) databases and compilations, including all data and collections of data, whether machine readable or otherwise; (iii) descriptions, flow charts and other work products used to design, plan, organize and develop any of the foregoing, screens, user interfaces, report formats, firmware, development tools, templates, menus, buttons, icons, web content and links; and (iv) documentation relating to any of the foregoing, including user manuals and other training documentation.
- "SpinCo Affiliate Contract" means any Contract, whether or not in writing, (a) between SpinCo, on the one hand, and any present or former officer or director of SpinCo or "immediate family member" thereof (as defined in Rule 16a-1 under the Exchange Act), on the other hand, or (b) between SpinCo, on the one hand, and the Company and/or any of its Subsidiaries (other than SpinCo), on the other hand.
 - "SpinCo Assets" has the meaning set forth in the Separation and Distribution Agreement.
 - "SpinCo Business" has the meaning set forth in the Separation and Distribution Agreement.
 - "SpinCo Common Stock" means the common stock, par value \$0.001 per share, of SpinCo.
 - "SpinCo Contract" has the meaning set forth in the Separation and Distribution Agreement.
 - "SpinCo Datasite" means the datasite established by the Company for purposes of due diligence of SpinCo and the SpinCo Business.
- "SpinCo Disclosure Schedule" means the Disclosure Schedule delivered by the Company and SpinCo to Parent on the date hereof and identified as such.

- "SpinCo Holder" means each holder of shares of SpinCo Common Stock as of immediately prior to the Effective Time.
- "SpinCo IT Systems" means all IT Assets owned, leased or used by or on behalf of (i) as of the date hereof, the Company or any of its Subsidiaries with respect to the SpinCo Business, and (ii) as of the Separation, SpinCo, including, in each case, all IT Assets included in the SpinCo Assets.
- "SpinCo Licensed Intellectual Property" means all Intellectual Property licensed from a third party pursuant to a written Contract to, (a) as of the date hereof, the Company or any of its Subsidiaries with respect to the SpinCo Business, and (b) as of the Separation, SpinCo, that, in each case of clauses (a) and (b), is used, practiced or held for use or practice by or on behalf of (i) as of the date hereof, the Company or any of its Subsidiaries with respect to the SpinCo Business, and (ii) as of the Separation, SpinCo.
- "SpinCo Material Adverse Effect" means any Effect that has, or would reasonably be expected to, individually or in the aggregate, materially impair, materially delay or otherwise have a material adverse effect on the ability of SpinCo to perform its obligations hereunder or under the other Transaction Documents or to consummate the Transactions, including the Separation and the Distribution, or prevent it from performing such obligations or consummating such Transactions.
- "SpinCo Owned Intellectual Property" means all Intellectual Property owned or purported to be owned by SpinCo as of the Separation, including the SpinCo Intellectual Property (as defined in the Separation and Distribution Agreement).
- "SpinCo Payables" means all accrued costs or payables of SpinCo as of the Measurement Time, excluding payments related to the Jet.AI Retained Business. "SpinCo Payables" includes the Card Fee Balance, as of the Measurement Time, of each Jet Card Agreement included among the SpinCo Assets.
- "SpinCo Transaction Expenses" means the aggregate amount of all (a) out-of-pocket expenses and fees incurred by SpinCo in connection with the preparation of this Agreement and the consummation of the transactions contemplated hereby, including all brokers' or finders' fees and fees and expenses of counsel, investment bankers, accountants and other advisers, (b) any change-in-control, transaction bonus, retention, severance or other similar payments to any current or former employee, consultant, manager, trustee, independent contractor, officer, or director of the SpinCo arising from or incurred in connection with this Agreement or the transactions contemplated hereby, (c) the employer portion of any Taxes arising from or incurred in connection with the payments described in clause (b) or otherwise payable with respect to or in connection with the consummation of the transactions contemplated hereby, and (d) all obligations of the type referred to in clauses (a) through (c) of any Person for the payment of which SpinCo is responsible or liable, directly or indirectly.
- "Subsidiary" means, with respect to any Person, a corporation or other entity of which more than 50% of the voting power of the equity securities or Interests that by their terms have

ordinary voting power to elect a majority of the board of directors or other similar body is owned or controlled, directly or indirectly, by such Person, or any organization of which such Person or any of its Subsidiaries is, directly or indirectly, a general partner or managing member or holds a similar role.

"Superior Proposal" means a bona fide written Takeover Proposal (except that, for purposes of this definition, each reference in the definition of "Takeover Proposal" to "15%" shall be "50%") that the Company Board determines in good faith is more favorable from a financial point of view to the holders of Company Common Stock than the transactions contemplated by this Agreement and is reasonably likely to be consummated in accordance with its terms, taking into account the terms and conditions and prospects for completion of such Takeover Proposal and of this Agreement (including any revisions to the terms of this Agreement and the Merger proposed by Parent during the Superior Proposal Notice Period set forth in Section 7.10(d)).

"Surplus" means the amount, if any, by which the Purchase Price exceeds the Initial Purchase Price.

"Takeover Proposal" means a bona fide written proposal or offer by any Person or group (other than Parent and its Subsidiaries, including Merger Sub), relating to any transaction or series of related transactions (other than the transactions contemplated by this Agreement), involving any: (a) direct or indirect acquisition of assets of the Company or its Subsidiaries (including any voting equity interests of Subsidiaries, but excluding sales of assets in the ordinary course of business) equal to 15% or more of the fair market value of the Company's and its Subsidiaries' consolidated assets or to which 15% or more of the Company's and its Subsidiaries' net revenues or net income on a consolidated basis are attributable; (b) direct or indirect acquisition of 15% or more of the voting equity interests of the Company or any of its Subsidiaries whose business constitutes 15% or more of the consolidated net revenues, net income, or assets of the Company and its Subsidiaries, taken as a whole; (c) tender offer or exchange offer that if consummated would result in any Person or group (as defined in Section 13(d) of the Exchange Act) beneficially owning (within the meaning of Section 13(d) of the Exchange Act) 15% or more of the voting power of the Company; or (d) merger, consolidation, other business combination, or similar transaction involving the Company or any of its Subsidiaries, pursuant to which such Person or group (as defined in Section 13(d) of the Exchange Act) would own 15% or more of the consolidated net revenues, net income, or assets of the Company, and its Subsidiaries, taken as a whole.

"Tax" or "Taxes" mean (a) any and all taxes, charges, fees, duties, levies, imposts or other assessments, including federal, state, local, or municipal, or non-U.S. income, gross receipts, capital stock, net proceeds, ad valorem, turnover, commercial activity, real, personal and other property (tangible and intangible), sales, use, franchise, excise, value added, stamp, leasing, lease, user, transfer, fuel, excess profits, occupational, interest equalization, windfall or other profits, unitary, severance, withholding, unemployment, Social Security, estimated, alternative minimum, add-on minimum, real property gains, registration, unclaimed property or abandoned property or escheat, natural resources, net investment income, environmental,

customs, duties, disability, payroll, license, employee, premium, intangible, recording, proceeds, goods and services, healthcare, branch, capital gains, capital, net worth, production, employer health, government pension plan premiums and contributions, workers' compensation premiums, employment/unemployment insurance or compensation premiums and contributions, or other tax, duty, charge, fee, levy, imposts or other governmental charge or assessment in the nature of a tax (including the recapture, cancellation or clawback of any tax items such as investment tax credits or credits or payments in respect of claims for relief from Tax in relation to research and development expenditure) or deficiencies thereof, including any interest, penalties or additions to tax related thereto imposed by any Governmental Authority, in each case whether disputed or not, and (b) any Liability for the payment of any amounts of the type described in clause (a) of this definition as a result of being a member of an affiliated, consolidated, combined or unitary group for any period, as a result of any tax sharing agreement or group payment arrangement, or as a result of being liable for another Person's taxes as a transferee or successor, by contract or otherwise.

"Tax Period" or "Taxable Period" means any period prescribed by any Governmental Authority for which a Tax Return is required to be filed or a Tax is required to be paid.

"Tax Return" means any report, declaration, return, information return, claim for refund, estimate, declaration, statement, election, or other document filed or required to be filed with any Governmental Authority or submitted to another Person in connection with the determination, assessment, deposit, or collection of any Tax or the administration of any Laws related to any Tax, and including any related or supporting schedules, attachments, appendices, addenda, or amendments thereof.

"Trade Secret" has the meaning set forth in definition of "Intellectual Property".

"Trademark" has the meaning set forth in definition of "Intellectual Property".

"Transaction Documents" means the Separation and Distribution Agreement and Support Agreements, in each case, including all annexes, Exhibits, Schedules, attachments and appendices thereto, and any certificate or other instrument delivered by any Party to any other Party pursuant to this Agreement or any of the foregoing.

"Transaction Process" means all matters relating to the separation, disposition or sale of the SpinCo Business and the review of strategic alternatives with respect to the SpinCo Business (including the potential spin-off of the SpinCo Business), including matters relating to (a) the solicitation of proposals from and negotiations with third parties in connection with the disposition or sale of the SpinCo Business or SpinCo Assets or (b) the drafting, negotiation or interpretation of any of the provisions of this Agreement or the other Transaction Documents, or the determination of the allocation of any assets or Liabilities pursuant to the foregoing agreements or the transactions contemplated thereby.

"Transactions" means the Separation, the Distribution, the Merger and the other transactions contemplated by this Agreement and the Transaction Documents.

<u>Cross References</u>. Each of the following terms is defined in the Section set forth opposite such term:

Aircraft Section 5.23(b) Aviation Laws Section 5.23(a) Certificate of Merger Section 5.23(e) Chosen Courts Section 11.1 Claimant Section 9.3(a) Closing Section 2.2 Closing Date Section 2.2 Conditions Section 9.3(e)(i) Company Preamble Company Acquisition Agreement Section 7.10(a) Company Board Recitals DGCL Recitals Distribution Recitals Distribution Documents Section 5.20 DOT Section 5.23(a) Effective Time Section 5.23(a) Effective Time Section 2.3(e) Equity Adjustment Section 2.3(c) Estimated Net Cash Section 2.3(c) Estimated Net Cash Statement Section 3.2(b) Exchange Agent Agreement Section 3.2(b) Exchange Fund Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d) </th <th>Term</th> <th>Section</th>	Term	Section
Certificate of Merger Section 2.3(e) Chosen Courts Section 11.1 Claimant Section 9.3(a) Closing Section 2.2 Closing Date Section 2.2 Conditions Section 9.3(c)(i) Company Preamble Company Acquisition Agreement Section 7.10(a) Company Board Recitals Distribution Recitals Distribution Documents Section 5.20 DOT Section 5.20 DOT Section 5.23(a) Effective Time Section 3.1(a)(iv) Estimated Net Cash Section 3.1(a)(iv) Estimated Net Cash Statement Section 2.3(c) Exchange Agent Section 3.2(b) Exchange Agent Agreement Section 3.2(b) Exchange Fund Section 5.23(a) Interim Period Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Aircraft	Section 5.23(b)
Chosen Courts Section 11.1 Claimant Section 9.3(a) Closing Section 2.2 Closing Date Section 2.2 Conditions Section 9.3(c)(i) Company Preamble Company Acquisition Agreement Section 7.10(a) Company Board Recitals DGCL Recitals Distribution Recitals Distribution Documents Section 5.20 DOT Section 5.23(a) Effective Time Section 2.3(e) Equity Adjustment Section 3.1(a)(iv) Estimated Net Cash Section 2.3(e) Estimated Net Cash Statement Section 3.2(b) Exchange Agent Section 3.2(b) Exchange Fund Section 3.2(b) FAA Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Aviation Laws	Section 5.23(a)
ClaimantSection 9.3(a)ClosingSection 2.2Closing DateSection 2.2ConditionsSection 9.3(e)(i)CompanyPreambleCompany Acquisition AgreementSection 7.10(a)Company BoardRecitalsDGCLRecitalsDistributionRecitalsDistribution DocumentsSection 5.20DOTSection 5.23(a)Effective TimeSection 5.23(e)Equity AdjustmentSection 3.1(a)(iv)Estimated Net CashSection 2.3(c)Estimated Net Cash StatementSection 3.2(b)Exchange Agent AgreementSection 3.2(b)Exchange FundSection 3.2(b)FAASection 5.23(a)Interim PeriodSection 5.23(a)Interim PeriodSection 4.7(d)Ionic AgreementSection 4.7(d)Section 4.7(d)	Certificate of Merger	Section 2.3(e)
Closing Section 2.2 Closing Date Section 9.3(c)(i) Company Preamble Company Acquisition Agreement Section 7.10(a) Company Board Recitals DGCL Recitals Distribution Recitals Distribution Documents Section 5.20 DOT Section 5.23(a) Effective Time Section 2.3(e) Equity Adjustment Section 3.1(a)(iv) Estimated Net Cash Section 2.3(e) Estimated Net Cash Statement Section 3.2(b) Exchange Agent Section 3.2(b) Exchange Agent Agreement Section 3.2(b) Exchange Fund Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Chosen Courts	Section 11.1
Closing Date Section 2.2 Conditions Section 9.3(c)(i) Company Preamble Company Acquisition Agreement Section 7.10(a) Company Board Recitals DGCL Recitals Distribution Recitals Distribution Documents Section 5.20 DOT Section 5.23(a) Effective Time Section 2.3(e) Equity Adjustment Section 3.1(a)(iv) Estimated Net Cash Section 2.3(c) Estimated Net Cash Statement Section 3.2(b) Exchange Agent Section 3.2(b) Exchange Fund Section 3.2(b) FAA Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Claimant	Section 9.3(a)
Conditions Section 9.3(c)(i) Company Preamble Company Acquisition Agreement Section 7.10(a) Company Board Recitals DGCL Recitals Distribution Recitals Distribution Documents Section 5.20 DOT Section 5.23(a) Effective Time Section 2.3(e) Equity Adjustment Section 2.3(e) Estimated Net Cash Section 2.3(e) Estimated Net Cash Statement Section 2.3(e) Exchange Agent Section 3.2(b) Exchange Fund Section 3.2(b) Exchange Fund Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Closing	Section 2.2
Company Acquisition Agreement Section 7.10(a) Company Board Recitals DGCL Recitals Distribution Documents Section 5.20 DOT Section 5.23(a) Effective Time Section 2.3(e) Equity Adjustment Section 3.1(a)(iv) Estimated Net Cash Statement Section 2.3(c) Estimated Net Cash Statement Section 3.2(b) Exchange Agent Agreement Section 3.2(b) Exchange Fund Section 3.2(b) FAA Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Closing Date	Section 2.2
Company Acquisition AgreementSection 7.10(a)Company BoardRecitalsDGCLRecitalsDistributionRecitalsDistribution DocumentsSection 5.20DOTSection 5.23(a)Effective TimeSection 2.3(e)Equity AdjustmentSection 3.1(a)(iv)Estimated Net CashSection 2.3(c)Estimated Net Cash StatementSection 2.3(c)Exchange AgentSection 3.2(b)Exchange Agent AgreementSection 3.2(b)Exchange FundSection 3.2(b)FAASection 5.23(a)Interim PeriodSection 7.1IonicSection 4.7(d)Ionic AgreementSection 4.7(d)	Conditions	Section 9.3(c)(i)
Company BoardRecitalsDGCLRecitalsDistributionRecitalsDistribution DocumentsSection 5.20DOTSection 5.23(a)Effective TimeSection 2.3(e)Equity AdjustmentSection 3.1(a)(iv)Estimated Net CashSection 2.3(c)Estimated Net Cash StatementSection 2.3(c)Exchange AgentSection 3.2(b)Exchange Agent AgreementSection 3.2(b)Exchange FundSection 3.2(b)FAASection 5.23(a)Interim PeriodSection 7.1IonicSection 4.7(d)Ionic AgreementSection 4.7(d)	Company	Preamble
DGCL Distribution Recitals Distribution Documents Section 5.20 DOT Section 5.23(a) Effective Time Section 2.3(e) Equity Adjustment Setimated Net Cash Estimated Net Cash Statement Section 2.3(c) Estimated Net Cash Statement Section 2.3(c) Exchange Agent Section 3.2(b) Exchange Agent Agreement Section 3.2(b) Exchange Fund Section 3.2(b) FAA Section 5.23(a) Interim Period Indic Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Company Acquisition Agreement	Section 7.10(a)
DistributionRecitalsDistribution DocumentsSection 5.20DOTSection 5.23(a)Effective TimeSection 2.3(e)Equity AdjustmentSection 3.1(a)(iv)Estimated Net CashSection 2.3(c)Estimated Net Cash StatementSection 2.3(c)Exchange AgentSection 3.2(b)Exchange Agent AgreementSection 3.2(b)Exchange FundSection 3.2(b)FAASection 5.23(a)Interim PeriodSection 7.1IonicSection 4.7(d)Ionic AgreementSection 4.7(d)	Company Board	Recitals
Distribution DocumentsSection 5.20DOTSection 5.23(a)Effective TimeSection 2.3(e)Equity AdjustmentSection 3.1(a)(iv)Estimated Net CashSection 2.3(c)Estimated Net Cash StatementSection 2.3(c)Exchange AgentSection 3.2(b)Exchange Agent AgreementSection 3.2(b)Exchange FundSection 3.2(b)FAASection 5.23(a)Interim PeriodSection 7.1IonicSection 4.7(d)Ionic AgreementSection 4.7(d)	DGCL	Recitals
DOTSection 5.23(a)Effective TimeSection 2.3(e)Equity AdjustmentSection 3.1(a)(iv)Estimated Net CashSection 2.3(c)Estimated Net Cash StatementSection 2.3(c)Exchange AgentSection 3.2(b)Exchange Agent AgreementSection 3.2(b)Exchange FundSection 3.2(b)FAASection 5.23(a)Interim PeriodSection 7.1IonicSection 4.7(d)Ionic AgreementSection 4.7(d)		Recitals
Effective Time Equity Adjustment Section 2.3(e) Equity Adjustment Section 3.1(a)(iv) Estimated Net Cash Section 2.3(c) Estimated Net Cash Statement Section 2.3(c) Exchange Agent Section 3.2(b) Exchange Agent Agreement Section 3.2(b) Exchange Fund Section 3.2(b) FAA Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Distribution Documents	Section 5.20
Equity Adjustment Estimated Net Cash Estimated Net Cash Statement Exchange Agent Exchange Agent Agreement Exchange Fund FAA Interim Period Ionic Agreement Section 3.1(a)(iv) Section 2.3(c) Section 2.3(c) Section 2.3(c) Section 3.2(b) Section 3.2(b) Section 3.2(b) Section 3.2(b) Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Section 4.7(d)	DOT	Section 5.23(a)
Estimated Net Cash Estimated Net Cash Statement Section 2.3(c) Exchange Agent Section 3.2(b) Exchange Agent Agreement Section 3.2(b) Exchange Fund Section 3.2(b) FAA Section 3.2(b) FAA Section 5.23(a) Interim Period Ionic Section 7.1 Ionic Section 4.7(d) Section 4.7(d)	Effective Time	Section 2.3(e)
Estimated Net Cash Statement Exchange Agent Exchange Agent Agreement Exchange Fund Exchange Agent Agreement Exchange Fund Exchange Fund	Equity Adjustment	Section 3.1(a)(iv)
Exchange Agent Agreement Section 3.2(b) Exchange Fund Section 3.2(b) Exchange Fund Section 3.2(b) FAA Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Estimated Net Cash	Section 2.3(c)
Exchange Agent Agreement Exchange Fund Section 3.2(b) Section 3.2(b) FAA Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Estimated Net Cash Statement	Section 2.3(c)
Exchange Fund Section 3.2(b) FAA Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Exchange Agent	Section 3.2(b)
FAA Section 5.23(a) Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Exchange Agent Agreement	Section 3.2(b)
Interim Period Section 7.1 Ionic Section 4.7(d) Ionic Agreement Section 4.7(d)	Exchange Fund	Section 3.2(b)
IonicSection 4.7(d)Ionic AgreementSection 4.7(d)	FAA	Section 5.23(a)
Ionic Agreement Section 4.7(d)	Interim Period	Section 7.1
-	Ionic	Section 4.7(d)
	Ionic Agreement	Section 4.7(d)
Measurement Time Section 2.3(c)	Measurement Time	Section 2.3(c)
Merger Recitals	Merger	Recitals

[&]quot;Treasury Regulations" means the final or temporary regulations promulgated by the U.S. Treasury Department under the Code.

[&]quot;U.S." or "United States" means and refers to the United States of America.

[&]quot;Willful Breach" means, with respect to any obligation, covenant or agreement of a Party in this Agreement, any material breach of or material failure to perform such obligation, covenant or agreement that such Party intentionally takes (or intentionally fails to take or perform) with actual knowledge that such action or omission or failure to perform would, or would reasonably be expected to, cause or result in a material breach of this Agreement.

Merger Sub Preamble

Merger Sub Common Stock

Outside Date

Parent

Preamble

Parent Expenses Reimbursement Amount

Parent SEC Filings

Parent Securities

Parent Securities

Section 6.3(a)

Parent Termination Fee

Section 10.3(b)

Party and Parties Preamble
Per Share Merger Consideration Section 3.1(a)(i)
Remedies Exception Section 4.2
Separation Recitals
SpinCo Preamble
SpinCo Material Contracts Section 5.14(a)

SpinCo Registered Intellectual Property
Section 5.16(a)
SpinCo Shareholder Approval
Section 5.21
Stockholder Support Agreement
Recitals
Superior Proposal Notice Period
Section 7.10(d)

Surviving Corporation Section 2.1
Third-Party Claim Section 9.3(a)
Unpaid Transaction Expenses Section 2.3(d)

Exhibit A

Separation and Distribution Agreement

See attached.

SEPARATION AND DISTRIBUTION AGREEMENT DATED AS OF FEBRUARY 13, 2025

by and among

JET.AI INC.,

JET.AI SPINCO, INC.

and

FLYEXCLUSIVE, INC.

TABLE OF CONTENTS

Page

<u>1.1</u>	General 5
<u>1.2</u>	References; Interpretation 5
<u>Article II – </u>	THE CONTRIBUTION 6
<u>2.1</u>	Restructuring; Transfer of Assets; Assumption of Liabilities 6
<u>2.2</u>	Non-Transferable Assets 7
<u>2.3</u>	Conveyancing and Assumption Instruments 8
<u>2.4</u>	Further Assurances; Ancillary Agreements 8
<u>2.5</u>	Consideration 8
Article III –	COMPLETION OF THE DISTRIBUTION 8
3.1	Actions Prior to the Distribution 9
3.2	Effecting the Distribution 9
3.3	Conditions to the Distribution 9
Article IV -	REPRESENTATIONS AND WARRANTIES OF JET.AI RELATING TO JET.AI 10
<u>4.1</u>	Organization of the Jet.AI Group 10
<u>4.2</u>	<u>Due Authorization</u> 10
<u>4.3</u>	Governmental Consents 11
<u>4.4</u>	No Conflict 11
<u>4.5</u>	<u>Litigation and Proceedings</u> 11
<u>4.6</u>	Brokers' Fees 11
<u>4.7</u>	Sufficiency of the SpinCo Assets 12
Article V –	CERTAIN COVENANTS 12
5.1	Cooperation 12
Article VI -	MISCELLANEOUS 12
<u>6.1</u>	Entire Agreement; Construction 12
<u>6.2</u>	Ancillary Agreements 13
<u>6.3</u>	Counterparts 13
<u>6.4</u>	Survival of Agreements 13
<u>6.5</u>	Expenses 13
<u>6.6</u>	Notices 13
<u>6.7</u>	Consents 14
<u>6.8</u>	Assignment 14
<u>6.9</u>	Successors and Assigns 15
<u>6.10</u>	
<u>6.11</u>	
<u>6.12</u>	
<u>6.13</u>	
<u>6.14</u>	Schedules 15

- 6.15 Governing Law and Venue; Submission to Jurisdiction; Selection of Forum; Waiver of Trial by Jury 15
- 6.16 Specific Performance 16
 6.17 Severability 16
 6.18 Interpretation 17

- <u>6.19</u> No Waiver 17

LIST OF SCHEDULES

- 2.1(a)(i) SpinCo Assets
- 2.1(a)(ii) SpinCo Contracts
- 2.1(a)(iii) SpinCo Permits

SEPARATION AND DISTRIBUTION AGREEMENT

This SEPARATION AND DISTRIBUTION AGREEMENT (this "Agreement"), dated as of February 13, 2025, is entered into by and among Jet.AI Inc., a Delaware corporation ("Jet.AI"), Jet.AI SpinCo, Inc., a Delaware corporation and a wholly owned subsidiary of Jet.AI ("SpinCo"), and flyExclusive, Inc., a Delaware corporation ("flyExclusive"). "Party" or "Parties" means Jet.AI and SpinCo, individually or collectively, as the case may be. Terms with initial capitalized letters used herein (including in the immediately preceding sentence) and not otherwise defined herein have the meanings set forth in Section 1.1 hereof.

RECITALS

- A. Jet.AI owns 100% of the common stock, par value \$0.001 per share, of SpinCo (the "SpinCo Stock").
- B. Jet.AI, acting through its direct and indirect Subsidiaries, currently conducts the Jet.AI Retained Business and the SpinCo Business.
- C. The Board of Directors of Jet.AI (the "Jet.AI Board") has determined that it is appropriate, desirable and in the best interests of Jet.AI and its stockholders to separate Jet.AI into two separate companies, one for each of (i) the Jet.AI Retained Business, which shall be retained, owned and conducted, directly or indirectly, by Jet.AI and its Subsidiaries (other than SpinCo and its Subsidiaries) and (ii) the SpinCo Business, which shall be owned and conducted by SpinCo (the "Separation").
- D. In order to effect the Separation, the Jet.AI Board has determined that it is appropriate, desirable and in the best interests of Jet.AI and its stockholders for Jet.AI and the other members of the Jet.AI Group to effect the Contribution to SpinCo.
- E. The Jet.AI Board has further determined that it is appropriate and desirable, on the terms and conditions contemplated hereby, immediately following such separation to make a distribution of the SpinCo Business to the holders of common stock, par value \$0.001 per share, of Jet.AI (the "Jet.AI Stock") on the Record Date through the distribution of all of the outstanding shares of SpinCo Stock to holders of Jet.AI Stock on the Record Date on a pro rata basis in accordance with a distribution ratio to be determined by the Jet.AI Board (the "Distribution"), in each case, on the terms and conditions set forth in this Agreement.
 - F. Immediately following the Distribution, Jet.AI will hold none of the outstanding shares of SpinCo Stock.
- G. The Jet.AI Board has further determined that it is appropriate and desirable, on the terms and conditions contemplated in the Agreement and Plan of Merger and Reorganization, dated as of the date hereof (the "*Merger Agreement*"), among Jet.AI, SpinCo, flyExclusive and FlyX Merger Sub, Inc., a Delaware corporation and wholly owned subsidiary of flyExclusive

("Merger Sub"), that following the Separation and Distribution, Merger Sub will merge with and into SpinCo, with SpinCo continuing as the surviving corporation (the "Merger").

- H. Pursuant to the Merger, shares of SpinCo Stock will be exchanged for shares of Parent Common Stock, on the terms and conditions set forth in the Merger Agreement.
- I. Jet.AI and flyExclusive will jointly prepare, and flyExclusive will file with the SEC, the Registration Statement, which will set forth certain disclosure concerning the Separation, the Distribution and the Merger.
- J. The Jet.AI Board has (x) determined that the transactions contemplated by this Agreement, the Merger Agreement and the Ancillary Agreements have a valid business purpose, are in furtherance of and consistent with its business strategy and are in the best interests of Jet.AI and its stockholders and (y) approved this Agreement, the Merger Agreement and each of the Ancillary Agreements. The Board of Directors of SpinCo (the "SpinCo Board") has approved this Agreement, the Merger Agreement and each of the Ancillary Agreements (to the extent SpinCo is a party thereto).
 - K. The board of directors of flyExclusive has approved this Agreement and the Merger Agreement.
- L. The Parties and flyExclusive desire to set forth the principal corporate transactions required to effect the Separation and the Distribution, and certain other agreements relating to the relationship of Jet.AI and SpinCo and their respective Subsidiaries following the Distribution.

THEREFORE, in consideration of the foregoing and the mutual agreements, provisions and covenants contained in this Agreement, the Parties and flyExclusive hereby agree as follows:

Article I

DEFINITIONS AND INTERPRETATION

- 1.1 <u>General</u>. Unless otherwise provided herein, terms with initial capital letters used in this Agreement will have the meanings ascribed to such terms as set forth in <u>Annex A</u> attached hereto, which is incorporated herein and made a part hereof.
- 1.2 <u>References; Interpretation.</u> References in this Agreement to any gender include references to all genders, and references to the singular include references to the plural and vice versa. Unless the context otherwise requires, the words "include", "includes" and "including" when used in this Agreement shall be deemed to be followed by the phrase "without limitation". Unless the context otherwise requires, references in this Agreement to Articles, Sections, Annexes, Exhibits and Schedules shall be deemed references to Articles and Sections of, and Annexes, Exhibits and Schedules to, this Agreement. Unless the context otherwise requires, the words "hereof", "hereby" and "herein" and words of similar meaning when used in this Agreement refer to this Agreement in its entirety and not to any particular Article, Section or

provision of this Agreement. Unless the context otherwise requires, the word "or" shall be disjunctive but not exclusive. The words "written request" when used in this Agreement shall include email. Reference in this Agreement to any time shall be to New York City, New York time unless otherwise expressly provided herein. Unless the context requires otherwise, references in this Agreement to "Jet.AF" shall also be deemed to refer to the applicable member of the Jet.AI Group and, in connection therewith, any references to actions or omissions to be taken, or refrained from being taken, as the case may be, by Jet.AI shall be deemed to require Jet.AI to cause the applicable members of the Jet.AI Group to take, or refrain from taking, any such action. In the event of any inconsistency or conflict which may arise in the application or interpretation of any of the definitions set forth in Section 1.1, for the purpose of determining what is and is not included in such definitions, any item explicitly included on a Schedule referred to in any such definition shall take priority over any provision of the text thereof.

Article II

THE CONTRIBUTION

2.1 Restructuring; Transfer of Assets; Assumption of Liabilities.

- (a) <u>Transfer of Assets</u>. Effective as of immediately before the Distribution Time, pursuant to the Conveyancing and Assumption Instruments and in connection with the Contribution, Jet.AI shall, and shall cause the applicable Asset Transferors to, transfer, contribute, distribute, assign or convey or cause to be transferred, contributed, distributed, assigned or conveyed ("*Transfer*") to SpinCo, all of its and the applicable Asset Transferors' right, title and interest in and to the SpinCo Assets, and SpinCo shall accept from Jet.AI and the applicable members of the Jet.AI Group all of Jet.AI's and the other members of the Jet.AI Group's respective direct or indirect rights, title and interest in and to the applicable SpinCo Assets, free and clear of all Liens. "*SpinCo Assets*" means all of Jet.AI's or the applicable Asset Transferor's right, title, and interest in and to the following assets, properties and rights:
 - (i) the Assets set forth on Schedule $2.1(a)(i)^{\frac{1}{2}}$;
 - (ii) all Contracts set forth on Schedule 2.1(a)(ii) (the "SpinCo Contracts");
- (iii) the licenses, permits, registrations, approvals and authorizations which have been issued by any Governmental Entity set forth on Schedule 2.1(a)(iii) (the "SpinCo Permits");
 - (iv) all Information exclusively related to, or exclusively used in, the SpinCo Business;
 - (v) an amount of cash equal to the Available Cash Amount; and
 - (vi) all goodwill exclusively related to the SpinCo Business.

6

¹ **NTD**: To include the rights to the Textron Deposits.

- (b) In connection with the Contribution and Transfer of the SpinCo Assets to SpinCo, and pursuant to this Agreement or the applicable Conveyancing and Assumption Instruments, SpinCo shall assume, as of immediately before the Distribution Time, and shall pay, perform and discharge when due the following obligations and liabilities arising out of or relating to the SpinCo Business and the SpinCo Assets (the "SpinCo Liabilities"):
- (i) all liabilities and obligations under the SpinCo Contracts to be discharged or performed following the Distribution Time, but only to the extent that such liabilities and obligations do not relate to any failure to perform, improper performance, warranty, or other breach, default, or violation by Jet.AI on or prior to the Contribution; and
- (ii) all liabilities and obligations under each SpinCo Permit to be discharged or performed following the Distribution Time, but only to the extent that such liabilities and obligations do not relate to any failure to perform, improper performance, warranty, or other breach, default, or violation by Jet.AI on or prior to the Contribution.
- (c) Other than the SpinCo Liabilities, SpinCo does not assume any Liabilities of Jet.AI, any member of the Jet.AI Group, or any Affiliate of Jet.AI or any member of the Jet.AI Group (collectively, the "Jet.AI Retained Liabilities"), including (a) liabilities arising out of the Jet.AI Retained Assets, and (b) liabilities for Taxes of the Jet.AI Group (or any Affiliate of the Jet.AI Group) and any liabilities for Taxes (i) of the Jet.AI Group for any Tax Period or (ii) with respect to ownership and operation of the SpinCo Business, the ownership of the SpinCo Assets or the SpinCo Liabilities for any Pre-Closing Period.
- (d) <u>Consents</u>. The Parties shall use their commercially reasonable efforts to obtain the Consents required to Transfer any SpinCo Assets as contemplated by this Agreement.
- Non-Transferable Assets. Nothing in this Agreement shall be determined to Transfer to SpinCo any Contract or other asset that would otherwise be included in the SpinCo Assets if an attempted Transfer (a) without the consent of any Person would constitute a breach or default of such Contract or asset or create a termination right under such Contract, and such Person does not give its consent, or (b) would otherwise be unlawful, ineffective or would materially impair SpinCo's rights under the Contract or asset in question so that SpinCo would in effect not acquire the benefit of such rights (each a "Non-Transferable Asset"). With respect to each such Non-Transferable Asset, (i) Jet.AI will cooperate in any reasonable and lawful arrangements designed to provide to SpinCo, and its successors, all of the benefits, economic and otherwise, of such Non-Transferable Asset (including promptly remitting to SpinCo any revenue received by Jet.AI on account of any such Non-Transferable Asset related to the Post-Closing Period), (ii) Jet.AI will not terminate, alter or amend any such Non-Transferable Asset except with SpinCo's express written consent, and (iii) until such Non-Transferable Asset is Transferred to SpinCo, Jet.AI will use its commercially reasonable efforts to enforce, at the reasonable direction and for the account, at the sole expense, of SpinCo, any right of Jet.AI arising under any such Non-Transferable Asset against any counterparty thereto or other Person (including the right to amend or terminate any such Non-Transferable Asset in accordance with the terms thereof). Jet.AI will provide SpinCo with notice of any claim asserted by any Person against or with respect to any Non-Transferable Asset after the Closing as soon as practicable after

becoming aware of any such claim. If any such claim is asserted, SpinCo will have the right to direct the defense of such claim at SpinCo's sole cost and expense, whether or not litigation, arbitration or other dispute resolution proceedings have been instituted and even if SpinCo is not named as a party to such claim.

2.3 <u>Conveyancing and Assumption Instruments.</u> In connection with, and in furtherance of, the Transfer of the SpinCo Assets and the assumption of the SpinCo Liabilities contemplated by this Agreement, the Parties shall execute or cause to be executed, on or after the date hereof by the appropriate member of the Jet.AI Group, any Conveyancing and Assumption Instruments necessary to evidence the valid Transfer to SpinCo of all right, title and interest in and to the SpinCo Assets and the assumption by SpinCo of the SpinCo Liabilities for Transfers and assumptions to be effected pursuant to Delaware Law or the Laws of one of the other states of the United States, in such form as the Parties and flyExclusive shall reasonably agree.

2.4 Further Assurances; Ancillary Agreements.

- (a) In addition to and without limiting the actions specifically provided for elsewhere in this Agreement and subject to the limitations expressly set forth in this Agreement, each of the Parties shall cooperate with each other and use (and shall cause its respective Subsidiaries and Affiliates to use) commercially reasonable efforts, at and after the Distribution Time, to take, or to cause to be taken, all actions, and to do, or to cause to be done, all things reasonably necessary on its part under applicable Law or contractual obligations to consummate and make effective the transactions contemplated by this Agreement and the Ancillary Agreements.
- (b) Without limiting the foregoing, at and after the Distribution Time, each Party shall cooperate with the other Party, and without any further consideration, from and after the Distribution Time, to execute and deliver, or use commercially reasonable efforts to cause to be executed and delivered, all instruments, including instruments of Transfer or title, and to make all filings with, and to obtain all Consents or Governmental Approvals, any permit, license, Contract, indenture or other instrument (including any Consents or Governmental Approvals), and to take all such other actions as such Party may reasonably be requested to take by any other Party from time to time, consistent with the terms of this Agreement and the Ancillary Agreements, in order to effectuate the provisions and purposes of this Agreement and the Ancillary Agreements and the Transfer of the SpinCo Assets and the assumption of the SpinCo Liabilities and the other transactions contemplated hereby and thereby. Without limiting the foregoing, Jet.AI, at its sole cost and expense and at the reasonable request of SpinCo, will take such other actions as may be reasonably necessary to vest in SpinCo such title and such rights as possessed by any member of the Jet.AI Group to the SpinCo Assets under this Agreement or any of the Ancillary Agreements, free and clear of any Security Interest.
- 2.5 <u>Consideration</u>. In exchange for the Contribution, SpinCo agrees, on or prior to the Distribution Date, to issue to Jet.AI a number of newly issued, fully paid and nonassessable shares of SpinCo Stock as is necessary to effect the Distribution and such that immediately following the Distribution Jet.AI will hold none of the outstanding shares of SpinCo Stock.

Article III

COMPLETION OF THE DISTRIBUTION

- 3.1 <u>Actions Prior to the Distribution</u>. Prior to the Distribution Time, subject to the terms and conditions set forth herein, the Parties shall take, or cause to be taken, the following actions in connection with the Distribution:
- (a) <u>Notice to Nasdaq</u>. Jet.AI shall give Nasdaq not less than ten (10) days' advance notice of the Record Date in compliance with Rule 10b-17 under the Exchange Act.
- (b) <u>Securities Law Matters</u>. Jet.AI and SpinCo shall take all such action as may be necessary or advisable under the securities or "blue sky" Laws of the United States (and any comparable Laws under any non-U.S. jurisdiction) in connection with the transactions contemplated by this Agreement.
- (c) <u>Authorized Number of Shares</u>. Prior to the Distribution, the Parties shall take all necessary action required to file an amendment to the Certificate of Incorporation of SpinCo with the Secretary of State of the State of Delaware to increase the number of authorized shares of SpinCo Stock so that SpinCo Stock then authorized shall be equal to the number of shares of SpinCo Stock necessary to effect the Distribution.
- (d) <u>Distribution Agent</u>. Jet.AI shall enter into a distribution agent agreement with the Exchange Agent or otherwise provide instructions to the Exchange Agent regarding the Distribution.
- (e) <u>Satisfying Conditions to the Distribution</u>. Each of Jet.AI and SpinCo shall cooperate to cause the conditions to the Distribution set forth in Section 3.3 to be satisfied and to effect the Distribution at the Distribution Time.
 - (f) <u>Contribution</u>. On or prior to the Distribution Date, the Contribution shall have been made.

3.2 Effecting the Distribution.

(a) <u>Delivery of SpinCo Stock</u>. Upon consummation of the Distribution, Jet.AI shall deliver to the Exchange Agent a bookentry authorization representing the SpinCo Stock being distributed in the Distribution for the account of the Record Holders and shall take all such other actions (including delivering any other instruments of transfer required by applicable law) as may be necessary to effect the Distribution. The Exchange Agent shall hold such book-entry shares for the account of the Record Holders pending the Merger, as provided in Section 3.2 of the Merger Agreement. Immediately after the Distribution Time and prior to the Effective Time, the shares of SpinCo Stock shall not be transferable and the transfer agent for the SpinCo Stock shall not transfer any shares of SpinCo Stock; <u>provided</u>, for the avoidance of doubt, that the exchange of such shares of SpinCo Stock for shares of Parent Common Stock pursuant to the Merger shall not be deemed a transfer subject to the foregoing restrictions. The Distribution shall

be deemed to be effective at the Distribution Time upon written authorization from Jet.AI to the Exchange Agent to proceed with the Distribution.

- 3.3 <u>Conditions to the Distribution</u>. The consummation of the Distribution shall be subject to the satisfaction or waiver by Jet.AI in its sole and absolute discretion, of the following conditions (other than the condition set forth in <u>Section 3.3(a)</u>, which prior to the termination of the Merger Agreement may not be waived without flyExclusive's written consent):
 - (a) <u>Completion of the Contribution</u>. The Contribution shall have been completed in accordance with this Agreement.
- (b) <u>Execution of Ancillary Agreements</u>. Each of the Ancillary Agreements shall have been duly executed and delivered by the parties thereto.
- (c) <u>Conditions to Jet.AI Obligations in Merger Agreement</u>. Each of the conditions in Section 8.1 (other than Section 8.1(b)) and Section 8.2) of the Merger Agreement to Jet.AI's obligations to consummate the Merger shall have been satisfied or waived (other than those conditions that by their nature are to be satisfied contemporaneously with the Distribution or the Merger; <u>provided</u>, that such conditions are capable of being satisfied at such time).
- (d) Confirmation of Satisfaction of flyExclusive Conditions in Merger Agreement flyExclusive shall have confirmed to Jet.AI that each condition in Section 8.1 (other than Section 8.1(b)) and Section 8.3 of the Merger Agreement to flyExclusive's obligations to consummate the Merger (i) has been satisfied, (ii) will be satisfied at the time of the Distribution or the Merger, or (iii) subject to applicable Law, is or has been waived by flyExclusive.

Article IV

REPRESENTATIONS AND WARRANTIES OF JET.AI RELATING TO JET.AI

Jet.AI hereby represents and warrants to SpinCo and flyExclusive as follows:

4.1 Organization of the Jet.AI Group. Each member of the Jet.AI Group has been duly incorporated or organized and is validly existing and in good standing under the Laws of its jurisdiction of organization or incorporation, as applicable. Jet.AI and each member of the Jet.AI Group has all requisite entity power and authority to own, lease and operate its properties and assets in the manner in which such assets and properties are now owned, leased and operated and to conduct its business as it is now being conducted. Each member of the Jet.AI Group is duly licensed or qualified and in good standing (or equivalent status as applicable) in each jurisdiction in which the assets owned or leased by it or the character of its activities require it to be so licensed or qualified or in good standing (or equivalent status as applicable), except as would not, individually or in the aggregate, have a Jet.AI Material Adverse Effect.

- Due Authorization. Subject to the receipt of the Company Shareholder Approval, each member of the Jet.AI Group has all requisite entity power and authority to execute and deliver this Agreement and the Ancillary Agreements, in each case to which it is or will be a party, and to consummate the Transactions, except for such further action of Jet.AI Board required, if applicable, to establish the Record Date and the Distribution Date, and the declaration of the Distribution by Jet.AI Board (the effectiveness of which is subject to the satisfaction or, to the extent permitted by applicable Law, waiver, of the conditions set forth in this Agreement). The execution and delivery by each member of Jet.AI Group of this Agreement or the Ancillary Agreements to which it is or will be a party as of the Distribution Time and the consummation of the Distribution have been duly authorized by all necessary and proper entity action on its part, and no other entity action on the part of such Jet.AI Group member is necessary to authorize this Agreement or the Ancillary Agreements to which it is or will be a party as of the Distribution Time or, subject to such further action of Jet.AI Board required, if applicable, to establish the Record Date and the Distribution Date, and the declaration of the Distribution by Jet.AI Board (the effectiveness of which is subject to the satisfaction or, to the extent permitted by applicable Law, waiver of the conditions set forth in this Agreement), and consummate the Transactions. Each of this Agreement and the Ancillary Agreements to which Jet.AI is or will be a party as of the Effective Time has been or will be duly and validly executed and delivered by it and (assuming that this Agreement constitutes a legal, valid and binding obligation of flyExclusive) constitutes or, when executed and delivered, will constitute the legal, valid and binding obligation of Jet.AI or the applicable Jet.AI Group member, enforceable against it in accordance with its terms, subject to applicable bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium and similar Laws affecting creditors' rights generally and subject, as to enforceability, to general principles of equity.
- 4.3 <u>Governmental Consents</u>. No Consent of, with or to any Governmental Entity is required to be obtained or made by any member of the Jet.AI Group in connection with the execution or delivery by Jet.AI of this Agreement or the Ancillary Agreements to which it is or will be a party or the consummation by Jet.AI of the Transactions.
- 4.4 No Conflict. Subject to the receipt of the Consents set forth in Section 4.3, the execution and delivery by Jet.AI and each member of the Jet.AI Group of this Agreement or any Ancillary Agreement to which it is or will be a party as of the Effective Time and the consummation by Jet.AI and each member of the Jet.AI Group of the Transactions do not and will not as of the Effective Time, (a) violate any provision of, or result in the breach of, any Law applicable to any member of the Jet.AI Group or by which any of its assets or properties is bound; (b) with or without lapse of time or the giving of notice or both, require a consent or approval under, conflict with, result in a violation or breach of, or constitute a default under, result in the acceleration of, create in any party the right to accelerate, terminate or cancel any Contract, including any SpinCo Contract; or (c) breach or violate any provision of the Governing Documents of any member of the Jet.AI Group, except, in the case of each of clauses (a) (b), as would not, individually or in the aggregate, reasonably be expected to have a Jet.AI Material Adverse Effect.

- 4.5 <u>Litigation and Proceedings.</u> (a) There are no Actions pending or threatened before or by any Governmental Entity against any member of the Jet.AI Group that, individually or in the aggregate, would reasonably be expected to result in a Jet.AI Material Adverse Effect, and (b) no member of the Jet.AI Group is subject to any judgment, decree, injunction or order of any Governmental Entity that, in each case, would reasonably be expected to result, individually or in the aggregate, in a Jet.AI Material Adverse Effect.
- 4.6 <u>Brokers' Fees</u>. No broker, investment banker, or other Person is entitled to any brokerage fee, finders' fee or other similar commission for which SpinCo would be liable in connection with the transactions contemplated by this Agreement based on arrangements made on behalf of Jet.AI or any of its Affiliates.

4.7 <u>Sufficiency of the SpinCo Assets</u>.

- (a) As of the Distribution Time (assuming receipt of all consents, approvals and authorizations relating to the matters set forth in Section 4.3), after giving effect to the Separation, the SpinCo Assets owned by SpinCo will constitute all of the assets, properties and rights reasonably necessary for SpinCo to conduct the SpinCo Business immediately following the Closing in all material respects in the same manner as it is conducted on the date hereof.
- (b) Except for Permitted Liens or Liens created by or through SpinCo after the Distribution Time, Jet.AI collectively has, and as of the Distribution Time (after giving effect to the Transactions) SpinCo shall have, good, valid and marketable title to, or a valid leasehold in, license to or other legal right to use, as the case may be, all of the SpinCo Assets, free and clear of any Liens.
- (c) The SpinCo Assets will include a deposit of at least \$3,500,000 held by Textron Aviation (the "*Textron Deposit*"). Following the Distribution Time, SpinCo will able to use the Textron Deposit in substantially the same manner and for substantially the same purposes as Jet.AI prior to the Separation.

Article V

CERTAIN COVENANTS

5.1 <u>Cooperation</u>. From and after the Distribution Time, and subject to the terms of and limitations contained in this Agreement, the Merger Agreement and the Ancillary Agreements, each Party shall, and shall cause each of its respective Affiliates and employees to provide reasonable cooperation and assistance to the other Party (and any member of its respective Group) in connection with the completion of the transactions contemplated herein and in each Ancillary Agreement, except as otherwise set forth in this Agreement or the Merger Agreement or may otherwise be agreed to by the Parties in writing, at no additional cost to the Party requesting such assistance other than for the actual out-of-pocket costs incurred by any such Party, if applicable.

Article VI

MISCELLANEOUS

- 6.1 Entire Agreement; Construction. This Agreement, including the Schedules, and the Ancillary Agreements and Merger Agreement shall constitute the entire agreement among the Parties and flyExclusive with respect to the subject matter hereof and shall supersede all previous negotiations, commitments, course of dealings and writings with respect to such subject matter. In the event of any inconsistency between this Agreement and any Schedule hereto, the Schedule shall prevail. In the event and to the extent that there shall be a conflict between the provisions of (a) this Agreement and the provisions of any Ancillary Agreement or Continuing Arrangement, such Ancillary Agreement or Continuing Arrangement shall control (except with respect to any Conveyancing and Assumption Instruments, in which case this Agreement shall control) and (b) this Agreement and any agreement which is not an Ancillary Agreement, this Agreement shall control unless specifically stated otherwise in such agreement. For the avoidance of doubt, the Conveyancing and Assumption Instruments are intended to be ministerial in nature and only to effect the transactions contemplated by this Agreement with respect to the applicable local jurisdiction and shall not expand or modify the rights and obligations of the Parties, flyExclusive or their respective Affiliates under this Agreement or any of the Ancillary Agreements that are not Conveyancing and Assumption Instruments.
- 6.2 <u>Representations and Warranties; Covenants</u>. Effective as of the Closing, each of the representations, warranties and covenants made by the Parties herein, will be deemed to be made for purposes of the Merger Agreement and shall be subject to the applicable survival and indemnification provisions set forth therein.
- 6.3 <u>Counterparts</u>. This Agreement may be executed in more than one counterpart, all of which shall be considered one and the same agreement, and shall become effective when one or more such counterparts have been signed by each of the Parties and flyExclusive and delivered to each of the Parties and flyExclusive.
- 6.4 <u>Survival of Agreements</u>. Except as otherwise contemplated by this Agreement or any Ancillary Agreement, all covenants and agreements of the Parties and flyExclusive contained in this Agreement and each Ancillary Agreement shall survive the Distribution Time and remain in full force and effect in accordance with their applicable terms.
- 6.5 Expenses. Except as otherwise specified in this Agreement, the Ancillary Agreements or the Merger Agreement, or as otherwise agreed in writing among Jet.AI, SpinCo and flyExclusive, Jet.AI and SpinCo shall each be responsible for its own fees, costs and expenses paid or incurred in connection with this Agreement, any Ancillary Agreement and the consummation of the Contribution, the Distribution and the Merger (the "*Transaction-related Expenses*").
- 6.6 Notices. All notices, requests, claims, demands and other communications under this Agreement and, to the extent applicable and unless otherwise provided therein, under each of the Ancillary Agreements shall be in English, shall be in writing and shall be given or made (and shall be deemed to have been duly given or made upon receipt) by delivery in person, by overnight courier service, by registered or certified mail (return receipt requested), or by email

(provided confirmation of transmission is electronically generated and kept on file by the sending party and so long as the sender of such email does not receive an automatic reply from the recipient's email server indicating that the recipient did not receive such email), to the respective Parties or flyExclusive at the following addresses (or at such other address for a Party or flyExclusive as shall be specified in a notice given in accordance with this <u>Section 6.6</u>):

```
if to Jet.AI or SpinCo, to:
              Jet.AI Inc.
              10845 Griffith Peak Dr. - Suite 200
              Las Vegas, NV 89135
              Attn: Mike Winston, Executive Chairman
              Telephone: []
              Email: []
              with a copy (which shall not constitute notice) to:
              Dykema Gossett PLLC
              111 E. Kilbourn Ave. - Suite 1050
              Milwaukee, WI 53202
              Attn: Kate Bechen and Andrew Frost
              Telephone: [ ] and [ ]
              Email: [ ] and [ ]
       if to flyExclusive, to:
              flyExclusive, Inc.
              2860 Jetport Road
              Kinston, NC 28504
              Attn: Brad Garner, Chief Financial Officer
              Telephone: []
              Email: []
              with a copy (which shall not constitute notice) to:
              Wyrick Robbins Yates and Ponton LLP
              4101 Lake Boone Trail, Suite 300
              Raleigh, NC 27607
              Attn: Larry E. Robbins and David P. Creekman
              Telephone: []
              Email: [] and []
or to such other address or addresses as the Parties may from time to time designate in writing by like notice.
```

- 6.7 <u>Consents</u>. Any consent required or permitted to be given by any Party or flyExclusive to the other Party or flyExclusive under this Agreement shall be in writing and signed by the Party or flyExclusive, as the case may be, giving such consent and shall be effective only against such Party (and its Group) or flyExclusive (and its Affiliates), as the case may be.
- 6.8 <u>Assignment</u>. This Agreement shall not be assignable, in whole or in part, directly or indirectly, by any Party or flyExclusive without the prior written consent of the other Party and flyExclusive, and any attempt to assign any rights or obligations arising under this Agreement without such consents shall be void. Notwithstanding the foregoing, this Agreement shall be assignable to a bona fide third party in connection with a merger, reorganization, consolidation or the sale of all or substantially all the assets of a party hereto so long as the resulting, surviving or transferee entity assumes all the obligations of the relevant party hereto by operation of law or pursuant to an agreement in form and substance reasonably satisfactory to the other parties to this Agreement; <u>provided, however</u>, that no assignment permitted by this <u>Section 6.8</u> shall release the assigning Party or flyExclusive, as the case may be, from liability for the full performance of its obligations under this Agreement.
- 6.9 <u>Successors and Assigns</u>. The provisions of this Agreement and the obligations and rights hereunder shall be binding upon, inure to the benefit of and be enforceable by (and against) the Parties, flyExclusive and their respective successors and permitted assigns.
- 6.10 <u>Termination and Amendment</u>. This Agreement may not be terminated, modified or amended except by an agreement in writing signed by each of Jet.AI, SpinCo and flyExclusive; <u>provided</u>, <u>however</u>, that this Agreement shall terminate immediately upon termination of the Merger Agreement if the Merger Agreement is terminated in accordance with its terms prior to the Distribution.
- 6.11 <u>Subsidiaries</u>. JetAI shall cause to be performed, and hereby guarantees the performance of, all actions, agreements and obligations set forth herein to be performed by any Subsidiary of Jet.AI, or by any entity that becomes a Subsidiary of Jet.AI at and after the Distribution Time, to the extent such Subsidiary remains a Subsidiary of Jet.AI.
- 6.12 <u>Third Party Beneficiaries</u>. Except as specifically provided in any Ancillary Agreement, this Agreement is solely for the benefit of the Parties and flyExclusive and should not be deemed to confer upon third parties any remedy, claim, liability, reimbursement, claim of Action or other right in excess of those existing without reference to this Agreement. For the avoidance of doubt, no stockholder of Jet.AI, SpinCo or flyExclusive shall be third-party beneficiaries for any purpose prior to the Distribution.
- 6.13 <u>Titles and Headings</u>. Titles and headings to Sections herein are inserted for the convenience of reference only and are not intended to be a part of or to affect the meaning or interpretation of this Agreement.
- 6.14 <u>Schedules</u>. The Schedules shall be construed with and as an integral part of this Agreement to the same extent as if the same had been set forth verbatim herein. Nothing in the

Exhibits or Schedules constitutes an admission of any liability or obligation of any member of the Jet.AI Group or the SpinCo Group or any of their respective Affiliates to any third party, nor, with respect to any third party, an admission against the interests of any member of the Jet.AI Group or the SpinCo Group or any of their respective Affiliates. The inclusion of any item or liability or category of item or liability on any Schedule is made solely for purposes of allocating potential liabilities among the Parties and shall not be deemed as or construed to be an admission that any such liability exists.

- 6.15 Governing Law and Venue; Submission to Jurisdiction; Selection of Forum; Waiver of Trial by Jury. This Agreement, and all claims, disputes, controversies or causes of action (whether in contract, tort, equity or otherwise) that may be based upon, arise out of or relate to this Agreement (including any schedule or exhibit hereto) or the negotiation, execution or performance of this Agreement (including any claim, dispute, controversy or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), shall be governed by and construed in accordance with the Laws of the State of Delaware, without regard to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than the State of Delaware. Each of the Parties and flyExclusive agrees that any Action related to this agreement shall be brought exclusively in the Court of Chancery of the State of Delaware or, if under applicable Law, exclusive jurisdiction over such matter is vested in the federal courts, any federal court in the State of Delaware and any appellate court from any thereof (the "Chosen Courts"). By executing and delivering this Agreement, each of the Parties and flyExclusive irrevocably: (i) accepts generally and unconditionally submits to the exclusive jurisdiction of the Chosen Courts for any Action relating to this Agreement, including any Action brought for any remedy contemplated by Section 6.16; (ii) waives any objections which such party may now or hereafter have to the laying of venue of any such Action contemplated by this Section 6.15 and hereby further irrevocably waives and agrees not to plead or claim that any such Action has been brought in an inconvenient forum; (iii) agrees that it will not attempt to deny or defeat the personal jurisdiction of the Chosen Courts by motion or other request for leave from any such court; (iv) agrees that it will not bring any Action contemplated by this Section 6.15 in any court other than the Chosen Courts; (v) agrees that service of all process, including the summons and complaint, in any Action may be made by registered or certified mail, return receipt requested, to such party at their respective addresses provided in accordance with this Agreement or in any other manner permitted by Law; and (vi) agrees that service as provided in the preceding clause (v) is sufficient to confer personal jurisdiction over such party in the Action, and otherwise constitutes effective and binding service in every respect. Each of the parties hereto agrees that a final judgment in any Action in a Chosen Court as provided above may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law, and each party further agrees to the non-exclusive jurisdiction of the Chosen Courts for the enforcement or execution of any such judgment.
- 6.16 Specific Performance. From and after the Distribution Time, in the event of any actual or threatened default in, or breach of, any of the terms, conditions and provisions of this Agreement or any Ancillary Agreement, the Parties and flyExclusive agree that flyExclusive and

the Party or Parties to this Agreement or such Ancillary Agreement who are or are to be thereby aggrieved shall have the right to specific performance and injunctive or other equitable relief of its or their rights under this Agreement or such Ancillary Agreement, in addition to any and all other rights and remedies at law or in equity, and all such rights and remedies shall be cumulative. The Parties and flyExclusive agree that, from and after the Distribution Time, the remedies at law for any breach or threatened breach of this Agreement or any Ancillary Agreement, including monetary damages, are inadequate compensation for any Indemnifiable Loss, that any defense in any action for specific performance that a remedy at law would be adequate is hereby waived, and that any requirements for the securing or posting of any bond with such remedy are hereby waived.

- 6.17 <u>Severability</u>. In the event any one or more of the provisions contained in this Agreement should be held invalid, illegal or unenforceable in any respect, the validity, legality and enforceability of the remaining provisions contained herein and therein shall not in any way be affected or impaired thereby. The Parties and flyExclusive shall endeavor in good-faith negotiations to replace the invalid, illegal or unenforceable provisions with valid provisions, the economic effect of which comes as close as possible to that of the invalid, illegal or unenforceable provisions.
- 6.18 <u>Interpretation</u>. The Parties and flyExclusive have participated jointly in the negotiation and drafting of this Agreement. This Agreement shall be construed without regard to any presumption or rule requiring construction or interpretation against the Party or flyExclusive, as the case may be, drafting or causing any instrument to be drafted.
- 6.19 No Waiver. No failure to exercise and no delay in exercising, on the part of any Party, any right, remedy, power or privilege hereunder or under the other Ancillary Agreements shall operate as a waiver hereof or thereof; nor shall any single or partial exercise of any right, remedy, power or privilege hereunder or thereunder preclude any other or further exercise thereof or the exercise of any other right, remedy, power or privilege.

[Signature page follows.]

IN WIT	NESS WHEREOF,	the Parties have cause	d this Agreement to	be duly executed b	y their respective	authorized offi	cers as of the
day and year fi	rst above written.		-	•	_		

JET.AI:

Jet.AI Inc.

By: /s/ Michael Winston
Name: Michael Winston
Title: Executive Chairman

SPINCO:

Jet.AI SpinCo, Inc.

By: /s/ Michael Winston
Name: Michael Winston
Title: Executive Chairman

[Signature Page to Separation and Distribution Agreement]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be duly executed by their respective authorized officers as of the
day and year first above written.

FLYEXCLUSIVE:

flyExclusive, Inc.

By: /s/ Thomas James Segrave, Jr.
Name: Thomas James Segrave, Jr.
Title: Chief Executive Officer

[Signature Page to Separation and Distribution Agreement]

Annex A

Defined Terms

Terms with initial capitalized letters used in the Agreement to which this Annex A is attached will have the following respective meanings, and all references to Sections, Schedules or Annexes in the following definitions will refer to Sections, Schedules or Annexes of or to such Agreement:

- "Action" means any demand, action, claim, suit, countersuit, arbitration, inquiry, subpoena, case, litigation, proceeding or investigation (whether civil, criminal, administrative or investigative) by or before any court or grand jury, any Governmental Entity or any arbitration or mediation tribunal.
- "Affiliate" means, when used with respect to a specified Person and at a point in, or with respect to a period of, time, a Person that directly or indirectly, through one or more intermediaries, controls, is controlled by, or is under common control with, such specified Person at such point in or during such period of time. For the purposes of this definition, "control", when used with respect to any specified Person means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of such Person, whether through the ownership of voting securities or other interests, by Contract or otherwise. It is expressly agreed that, from and after the Distribution Time, solely for purposes of this Agreement, (i) no member of the SpinCo Group shall be deemed an Affiliate of any member of the SpinCo Group.
 - "Agreement" has the meaning set forth in the Preamble.
- "Ancillary Agreements" means the Conveyancing and Assumption Instruments, and any other agreements to be entered into by and between any member of the Jet.AI Group, on one hand, and SpinCo, on the other hand, at, prior to or after the Distribution Time in connection with the Separation, the Distribution and the other transactions contemplated by this Agreement.
- "Asset Transferors" means the entities transferring SpinCo Assets to SpinCo in order to consummate the transactions contemplated hereby.
- "Assets" means any rights (including Intellectual Property), title and ownership interests in and to any properties, claims, Contracts, businesses, or assets (including goodwill), wherever located (including in the possession of vendors or other third parties or elsewhere), of every kind, character and description, whether real, personal or mixed, tangible or intangible, whether accrued, contingent or otherwise, in each case, whether or not recorded or reflected on the books and records or financial statements of any Person.
 - "Available Cash Amount" means an amount of cash equal to at least \$12,000,000.
 - "Business" means the Jet.AI Retained Business or the SpinCo Business, as applicable.

- "Business Day" means any day other than Saturday or Sunday and any other day on which commercial banking institutions located in New York, New York are required, or authorized by Law, to remain closed.
- "Business Entity" means any corporation, partnership, limited liability company, joint venture or other entity that may legally hold title to Assets.
 - "Chosen Courts" means the meaning set forth in Section 6.14.
 - "Closing" has the meaning set forth in the Merger Agreement.
 - "Company Shareholder Approval" has the meaning set forth in the Merger Agreement.
- "Consents" means any consents, waivers, notices, reports or other filings to be obtained from or made, including with respect to any Contract, or any registrations, licenses, permits, authorizations to be obtained from, or approvals from, or notification requirements to, any third parties, including any third party to a Contract and any Governmental Entity.
- "Continuing Arrangement" means this Agreement and the Ancillary Agreements (and each other Contract expressly contemplated by this Agreement or any Ancillary Agreement to be entered into or continued by any of the Parties or any of the members of their respective Groups).
- "Contract" means any agreement, contract, subcontract, obligation, binding understanding, note, indenture, instrument, option, lease, promise, arrangement, release, warranty, license, sublicense, insurance policy, benefit plan, purchase order or legally binding commitment or undertaking of any nature (whether written or oral and whether express or implied).
- "Contribution" means (i) the Transfer, directly or indirectly, of the SpinCo Assets from Jet.AI or its Subsidiaries to SpinCo and (ii) the assumption of the SpinCo Liabilities by SpinCo pursuant to the terms of this Agreement.
- "Conveyancing and Assumption Instruments" means, collectively, the various Contracts, including the related local asset transfer agreements and local stock transfer agreements, and other documents entered into prior to the Distribution Time and to be entered into to effect the Transfer of the SpinCo Assets and the assumption of the SpinCo Liabilities in the manner contemplated by this Agreement, or otherwise relating to, arising out of or resulting from the transactions contemplated by this Agreement, in such form or forms as the applicable Parties thereto agree.
 - "Distribution" has the meaning set forth in the Recitals.
- "Distribution Date" means the date on which Jet.AI, through the Agent, distributes all of the issued and outstanding shares of SpinCo Stock to holders of Jet.AI Stock in the Distribution, and "Distribution Time" means the time at which the Distribution occurs on the Distribution Date, which shall be deemed to be 12:01 a.m., New York time on the Distribution Date.

- "Effective Time" has the meaning set forth in the Merger Agreement.
- "Exchange Act" means the United States Securities Exchange Act of 1934, as amended, together with the rules and regulations promulgated thereunder.
 - "Exchange Agent" has the meaning set forth in the Merger Agreement.
 - "flyExclusive" has the meaning set forth in the Preamble.
 - "Governing Documents" has the meaning set forth in the Merger Agreement.
- "Governmental Approvals" means any notices or reports to be submitted to, or other registrations or filings to be made with, or any consents, approvals, licenses, permits or authorizations to be obtained from, any Governmental Entity.
- "Governmental Entity" means any nation or government, any state, municipality or other political subdivision thereof and any entity, body, agency, commission, department, board, bureau or court, whether domestic, foreign, multinational, or supranational exercising executive, legislative, judicial, regulatory, self-regulatory or administrative functions of or pertaining to government and any executive official thereof.
 - "Group" means (i) with respect to Jet.AI, the Jet.AI Group and (ii) with respect to SpinCo, the SpinCo Group.
- "Indemnifiable Loss" and "Indemnifiable Losses" means any and all damages, losses, deficiencies, Liabilities, obligations, penalties, judgments, settlements, claims, payments, fines, interest, costs and expenses (including the costs and expenses of any and all Actions and demands, assessments, judgments, settlements and compromises relating thereto and the costs and expenses of attorneys', accountants', consultants' and other professionals' fees and expenses incurred in the investigation or defense thereof or the enforcement of rights hereunder).
- "Information" means information, content and data (including personal information) in written, oral, electronic, computerized, digital or other tangible or intangible media, including (i) books and records, whether accounting, legal or otherwise, ledgers, studies, reports, surveys, designs, specifications, drawings, blueprints, diagrams, models, prototypes, samples, flow charts, marketing plans, customer names and information (including prospects), technical information relating to the design, operation, testing, test results, development, and manufacture of any Party's or its Group's, or any of their partners', products, technologies or materials or facilities (including specifications and documentation; engineering, design and manufacturing drawings, diagrams, layouts, maps and illustrations; formulations and material specifications; quality assurance policies, procedures and specifications; evaluation and validation studies; process control and/or shop-floor control strategy, logic or algorithms; assembly code, Software, firmware, programming data, databases, and all information referred to in the same); costs, margins and pricing; as well as product marketing studies and strategies; all other methodologies, procedures, techniques and know-how related to discovery, research, engineering, development and manufacturing; communications, correspondence, materials, product literature, artwork,

files, documents; and (ii) financial and business information, including earnings reports and forecasts, macro-economic reports and forecasts, all cost information (including partner and supplier records and lists), sales and pricing data, business plans, market evaluations, surveys, credit-related information, and other such information as may be needed for reasonable compliance with reporting, disclosure, filing or other requirements, including under applicable securities laws or regulations of securities exchanges.

- "Intellectual Property" has the meaning set forth in the Merger Agreement.
- "Jet.AI" has the meaning set forth in the Preamble.
- "Jet.AI Board" has the meaning set forth in the Recitals.
- "Jet.AI Group" means (i) Jet.AI, the Jet.AI Retained Business and each Person that is a direct or indirect Subsidiary of Jet.AI as of immediately following the Distribution Time and (ii) each Business Entity that becomes a Subsidiary of Jet.AI after the Distribution Time.
- "Jet.AI Material Adverse Effect" means any change, event, development, condition, occurrence or effect that has, or would reasonably be expected to, individually or in the aggregate, materially impair, materially delay or otherwise have a material adverse effect on the ability of Jet.AI to perform its obligations hereunder or under the Ancillary Agreements.
 - "Jet.AI Retained Assets" means any and all Assets of Jet.AI and any other member of the Jet.AI Group, other than the SpinCo Assets.
- "Jet.AI Retained Business" means (i) those businesses operated by the Jet.AI Group prior to the Distribution Time other than the SpinCo Business, and (ii) those Business Entities or businesses acquired or established by or for any member of the Jet.AI Group after the Distribution Time.
 - "Jet.AI Retained Liabilities" has the meaning set forth in Section 2.1(c).
 - "Jet.AI Stock" has the meaning set forth in the Recitals.
- "Law" means any applicable U.S. or non-U.S. federal, national, supranational, state, provincial, local or similar statute, law, ordinance, regulation, rule, code, income tax treaty, order, requirement or rule of law (including common law) or other binding directives promulgated, issued, entered into or taken by any Governmental Entity.
- "Liabilities" means any and all indebtedness, liabilities, costs, expenses, interest and obligations, whether accrued or fixed, absolute or contingent, matured or unmatured, known or unknown, reserved or unreserved, or determined or determinable, including those arising under any Law, Action, whether asserted or unasserted, or order, writ, judgment, injunction, decree, stipulation, determination or award entered by or with any Governmental Entity and those arising under any Contract or any fines, damages or equitable relief which may be imposed and including all costs and expenses related thereto.

- "Lien" has the meaning set forth in the Merger Agreement.
- "Merger" has the meaning set forth in the Recitals.
- "Merger Agreement" has the meaning set forth in the Recitals.
- "Merger Sub" has the meaning set forth in the Recitals.
- "Non-Transferable Asset" has the meaning set forth in Section 2.2.
- "Parent Common Stock" has the meaning set forth in the Merger Agreement.
- "Party" and "Parties" has the meanings set forth in the Preamble.
- "*Person*" means any natural person, firm, individual, corporation, business trust, joint venture, association, bank, land trust, trust company, company, limited liability company, partnership, or other organization or entity, whether incorporated or unincorporated, or any Governmental Entity.
 - "Permits" has the meaning set forth in the Merger Agreement.
 - "Permitted Liens" has the meaning set forth in the Merger Agreement.
 - "Pre-Closing Period" means the time period from the date hereof to the Effective Time.
- "Record Date" means 5:00 p.m. New York time on the date to be determined by the Jet.AI Board as the record date for determining stockholders of Jet.AI entitled to receive shares of SpinCo Stock in the Distribution.
 - "Record Holders" means the holders of record of Jet.AI Stock as of the Record Date.
 - "SEC" means the United States Securities and Exchange Commission.
- "Security Interest" means any mortgage, security interest, pledge, lien, charge, claim, option, right to acquire, voting or other restriction, right-of-entry, covenant, condition, easement, encroachment, restriction on transfer, or other encumbrance of any nature whatsoever, excluding restrictions on transfer under securities Laws.
 - "Separation" has the meaning set forth in the Recitals.
- "Software" means all: (i) computer programs, including all software implementations of algorithms, models and methodologies, whether in source code, object code, human readable form or other form; (ii) databases and compilations, including all data and collections of data, whether machine readable or otherwise; (iii) descriptions, flow charts and other work products used to design, plan, organize and develop any of the foregoing, screens, user interfaces, report formats, firmware, development tools, templates, menus, buttons, icons, web content and links;

and (iv) documentation relating to any of the foregoing, including user manuals and other training documentation.

- "SpinCo" has the meaning set forth in the Preamble.
- "SpinCo Assets" has the meaning set forth in Section 2.1(a).
- "SpinCo Board" has the meaning set forth in the Recitals.
- "SpinCo Business" means the business, operations, services and activities of the jet charter business of Jet.AI and its Subsidiaries as conducted with the SpinCo Assets as of immediately prior to the Closing by Jet.AI and its Subsidiaries.
 - "SpinCo Contracts" has the meaning set forth in Section 2.1(a)(ii).
- "SpinCo Group" means (i) SpinCo, the SpinCo Business and each Person that is a direct or indirect Subsidiary of SpinCo as of immediately following the Distribution Time and (ii) each Business Entity that becomes a Subsidiary of SpinCo after the Distribution Time.
 - "SpinCo Permits" has the meaning set forth in Section 2.1(a)(iii).
 - "SpinCo Stock" has the meaning set forth in the Recitals.
- "Subsidiary" means with respect to any Person (i) a corporation, fifty percent (50%) or more of the voting or capital stock of which is, as of the time in question, directly or indirectly owned by such Person and (ii) any other Person in which such Person, directly or indirectly, owns fifty percent (50%) or more of the equity or economic interest thereof or has the power to elect or direct the election of fifty percent (50%) or more of the members of the governing body of such entity.
 - "Tax" or "Taxes" has the meaning set forth in the Merger Agreement.
 - "Tax Period" has the meaning set forth in the Merger Agreement.
 - "Textron Deposit" has the meaning set forth in Section 4.7(c).
 - "Transaction Expenses" has the meaning set forth in Section 6.5.
- "*Transactions*" means, collectively, the Separation, Contribution, and Distribution, and other transactions contemplated by this Agreement.
 - "Transfer" has the meaning set forth in Section 2.1(a); and the term "Transferred" shall have its correlative meaning.
 - "US" means the United States of America.

FORM OF SUPPORT AGREEMENT

This SUPPORT AGREEMENT (this "Agreement"), dated as of [], 2025, is by and among flyExclusive, Inc., a Delaware corporation ("Parent"), FlyX Merger Sub, Inc., a Delaware corporation and wholly owned subsidiary of Parent ("Merger Sub"), and each of the persons set forth on Schedule I attached hereto (each, a "Equityholder", and collectively, the "Equityholders").

RECITALS

- A. Each Equityholder is, as of the date hereof, the holder of record or the "beneficial owner" (within the meaning of Rule 13d-3 under the Exchange Act) of the number of shares of securities of Jet.AI, Inc., a Delaware corporation (the "Company"), as set forth opposite the name of each Equityholder on Schedule I hereto (the "Owned Securities"). The Owned Securities, together with any shares of capital stock of the Company or other securities of the Company that each Equityholder purchases or with respect to which each Equityholder otherwise acquires beneficial ownership (as defined in Rule 13d-3 under the Exchange Act) after the date hereof, are referred to herein as the "Covered Securities".
- B. Parent, the Company, Merger Sub, and Jet.AI SpinCo, Inc., a Delaware corporation and wholly owned subsidiary of Company ("SpinCo") have negotiated (i) an Agreement and Plan of Merger, as may be amended, supplemented or otherwise modified from time to time (substantially in the form attached hereto as Exhibit A, the "Merger Agreement"), pursuant to which Merger Sub will merge with and into SpinCo (the "Merger") and (ii) a Separation and Distribution Agreement, as may be amended, supplemented or otherwise modified from time to time (substantially in the form attached hereto as Exhibit B, the "Separation Agreement"), pursuant to which the SpinCo Assets will be transferred from the Company and its Subsidiaries to SpinCo (the "Contribution") and the outstanding shares of SpinCo Stock (as defined in the Separation Agreement) will be distributed to holders of Jet.AI Stock (as defined in the Separation Agreement) on a pro rata basis (the "Distribution" and, together with the Merger, Contribution and all other transactions contemplated under the Merger Agreement and Separation Agreement, the "Transactions"), with SpinCo surviving the Merger as a wholly owned subsidiary of Parent, upon the terms and subject to the conditions set forth in the Merger Agreement and Separation Agreement. Capitalized terms used herein without definition shall have the respective meanings specified in the Merger Agreement.
- C. As a condition to the willingness of Parent and Merger Sub to enter into the Merger Agreement and consummate the Transactions, and as a material inducement and in consideration therefor, each Equityholder has agreed to enter into this Agreement.

AGREEMENT

In consideration of the foregoing and the mutual covenants and agreements set forth herein and in the Merger Agreement, and intending to be legally bound hereby, the parties hereto agree as follows:

- SECTION 1. Representations and Warranties of the Equityholders. Each Equityholder hereby represents and warrants to Parent and Merger Sub as follows:
- (a) Such Equityholder, if it is an entity, has all requisite power and authority or, if such Equityholder is an individual, has the legal capacity, to execute and deliver this Agreement and to consummate the transactions contemplated hereby. If such Equityholder is a corporation, limited partnership, or limited liability company, such Equityholder is an entity duly organized, validly existing and in good standing under the laws of the jurisdiction in which it is incorporated, organized, or constituted.
- (b) This Agreement has been duly executed and delivered by such Equityholder and constitutes a legal, valid and binding obligation of such Equityholder, enforceable against such Equityholder in accordance with its terms, subject to bankruptcy, insolvency (including all Laws relating to fraudulent transfers), reorganization, moratorium and similar Laws of general applicability relating to or affecting creditors' rights.
- (c) Such Equityholder has full voting power, with respect to the Owned Securities, and full power of disposition, full power to issue instructions with respect to the matters set forth herein, and full power to agree to all of the matters set forth in this Agreement, in each case with respect to all of its Owned Securities, with no restrictions on such Equityholder's right and powers of voting or disposition pertaining thereto and no person not a signatory to this Agreement has a beneficial interest in or a right to acquire or vote any of the Owned Securities (other than, if such Equityholder is a married individual and resides in a state with community property laws, the community property interest of his or her spouse to the extent applicable under such community property laws, which spouse hereby consents to this Agreement by executing the spousal consent attached hereto as **Exhibit B**). The Owned Securities held by such Equityholder are not subject to any proxy, voting trust or other agreement, arrangement or restriction with respect to the voting of such Owned Securities.
- (d) There is no Action pending or, to the knowledge of such Equityholder, threatened against such Equityholder at law or equity before or by any Governmental Authority that could reasonably be expected to impair or materially delay the performance by such Equityholder of its obligations under this Agreement or otherwise adversely impact such Equityholder's ability to perform its obligations hereunder.
- (e) Such Equityholder has received and reviewed a copy of the Merger Agreement and Separation Agreement. Such Equityholder understands and acknowledges that Parent and Merger Sub are entering into the Merger Agreement and Separation Agreement in reliance upon such Equityholder's execution, delivery and performance of this Agreement.

SECTION 2. Transfer of the Securities; Other Actions.

(a) Each Equityholder agrees not to (i) directly or indirectly sell, transfer, pledge, encumber, assign, or otherwise dispose of or hypothecate (including by gift or by contribution or distribution to any trust or similar instrument or to any beneficiaries of such Equityholder) (collectively, "*Transfer*"), or enter into any contract, option, or other arrangement or understanding (including any profit sharing arrangement) with respect to the Transfer of, any of the Covered Securities held by such Equityholder other than pursuant to the terms hereof and the Merger Agreement, or (ii) enter into any voting arrangement or understanding with respect to the Covered Securities held by such Equityholder, whether by proxy, voting agreement or otherwise, other than pursuant to this Agreement. Any attempted Transfer of Covered Securities in violation of this Section 2(a) shall be null and void.

- (b) Each Equityholder hereby irrevocably agrees that it will not commence or join in, and agrees to take all actions necessary to opt out of any class in any class action with respect to, any claim, derivative or otherwise, against Parent, Merger Sub, the Company or any of their respective successors challenging the validity of, or seeking to enjoin the operation of, any provision of this Agreement, the Merger Agreement or the Separation Agreement (including any claim seeking to enjoin or delay the Closing).
- (c) Except as otherwise provided herein, each Equityholder will not, in its capacity as a stockholder of the Company, directly or indirectly, take any action that would make any representation or warranty contained herein untrue or incorrect or that would have the effect of impairing the ability of such Equityholder to perform its obligations under this Agreement, the Merger Agreement or the Separation Agreement, or preventing or delaying the consummation of any of the transactions contemplated hereby or thereby.
- (d) During the term of this Agreement, in furtherance of this Agreement, the Equityholders hereby authorize the Company or its counsel to notify the Company's transfer agent that there is a stop transfer order with respect to all of the Covered Securities (and that this Agreement places limits on the voting and transfer of the Covered Securities), subject to the provisions hereof and provided that any such stop transfer order and notice will immediately be withdrawn and terminated by the Company following the Expiration Time.

SECTION 3. Voting of Securities.

- (a) Until the Expiration Time, at every annual, special or other meeting of the stockholders of the Company called, and at every adjournment or postponement thereof, each Equityholder shall, or shall cause the holder of record on any applicable record date to (i) if applicable, appear at each such meeting or otherwise cause all of such Equityholder's Covered Securities entitled to vote to be counted as present thereat for purposes of calculating a quorum, and (ii) vote all Covered Securities beneficially owned or controlled by Equityholder and entitled to vote (A) in favor of the approval of the Merger Agreement, the Separation Agreement, the approval of the Transactions and any actions required in furtherance thereof and hereof, , (B) in favor of the approval of any proposal to adjourn or postpone the meeting of the Stockholders to a later date, if there are not sufficient votes for the adoption and approval of the Merger Agreement, the Separation Agreement and the Transactions or such other transaction on the date on which such meeting is held, and (C) against (x) any action or agreement which would reasonably be expected to in any material respect impede, interfere with or prevent the Transactions, including, but not limited to, any other extraordinary corporate transaction, including any merger, acquisition, sale, consolidation, reorganization, recapitalization, extraordinary dividend or liquidation involving the Company and any Person (other than Parent, Merger Sub or their Affiliates) to acquire the Company or all or substantially all of the assets thereof, and (y) any action, proposal, transaction or agreement that would reasonably be expected to result in a breach of any covenant, representation or warranty or any other obligation or agreement of such Equityholder under this Agreement.
- (b) Notwithstanding the foregoing, each Equityholder shall retain at all times the right to vote the Covered Securities held by such Equityholder in its sole discretion and without any other limitation on those matters other than those set forth in this Agreement that are at any time or from time to time presented for consideration to the Stockholders generally.
- (c) Each Equityholder hereby appoints Parent and any designee of Parent, and each of them individually, until the Expiration Time (at which time this proxy shall automatically be revoked), as its proxies and attorneys-in-fact, with full power of substitution and resubstitution, to vote or act by written consent during the term of this Agreement with respect to the Covered

Securities in accordance with Section 3(a). This proxy and power of attorney is given to secure the performance of the duties of such Equityholder under this Agreement. Such Equityholder shall take such further action or execute such other instruments as may be necessary to effectuate the intent of this proxy. This proxy and power of attorney granted by such Equityholder shall be irrevocable during the term of this Agreement, shall be deemed to be coupled with an interest sufficient in law to support an irrevocable proxy, and shall revoke any and all prior proxies granted by such Equityholder with respect to the Covered Securities. The power of attorney granted by such Equityholder herein is a durable power of attorney and shall survive the bankruptcy, death, or incapacity of such Equityholder. The proxy and power of attorney granted hereunder shall terminate upon the termination of this Agreement.

SECTION 4. No Solicitation.

Each Equityholder agrees to adhere to, and be bound by, the same restrictions and prohibitions as are applicable to the Company pursuant to Section 7.10 of the Merger Agreement.

SECTION 5. Further Assurances.

Each Equityholder agrees, from time to time, and without additional consideration, to execute and deliver any additional documents, proxies and other instruments and take such further actions as Parent may reasonably request or as may be reasonably necessary or desirable to carry out all of the provisions hereof, including, without limitation, to vest in Parent the power to vote the Covered Securities, as applicable, to the extent contemplated by Section 3 hereof.

SECTION 6. Expenses.

All fees and expenses incurred in connection with the negotiation and execution of this Agreement and the transactions contemplated hereby shall be paid by the party incurring such fees and expenses.

SECTION 7. Termination.

- (a) This Agreement shall terminate upon the earlier to occur of (i) the Effective Time, and (ii) the termination of the Merger Agreement in accordance with its terms (the "*Expiration Time*"). No party hereto shall be relieved from any liability for any breach of this Agreement prior to any such termination.
- (b) This Agreement is conditioned upon the consummation of the Merger as contemplated in the Merger Agreement and shall be null and void and shall have no effect whatsoever, without any action on the part of any person, upon termination of the Merger Agreement for any reason prior to the occurrence of the Effective Time.

SECTION 8. <u>Miscellaneous.</u>

(a) Notices. All notices, waivers, consents and other communications to any party hereunder shall be in writing and shall be deemed to have been duly given (a) when delivered in person, (b) when delivered after posting in the national mail having been sent registered or certified mail return receipt requested, postage prepaid, (c) when delivered by FedEx or other internationally recognized overnight delivery service or (d) when delivered by facsimile (solely if receipt is confirmed) or email (so long as the sender of such email does not receive an automatic reply from the recipient's email server indicating that the recipient did not receive

such email), in each case to the parties at the address, or if applicable, email address following such party's name below or such other address, email address as such party may subsequently designate to the other parties by notice in accordance with this <u>Section 8</u>:

If to Equityholder, to the address set forth on the signature page hereto.

If to Parent or Merger Sub, to:

flyExclusive, Inc.
2860 Jetport Road
Kinston, NC 28504
Attn: Brad Garner, Chief Financial Officer
Telephone: []
Email: []

with a copy (which shall not constitute notice) to:

Wyrick Robbins Yates and Ponton LLP 4101 Lake Boone Trail, Suite 300 Raleigh, NC 27607 Attn: Larry E. Robbins and David P. Creekman Telephone: [] Email: [] and []

- (b) <u>Headings.</u> The headings contained in this Agreement are inserted for convenience only and shall not be considered in interpreting or construing any of the provisions contained in this Agreement.
- (c) <u>Counterparts</u>. This Agreement may be executed in two or more counterparts (including by electronic or .pdf transmission), each of which shall be deemed an original, but all of which together shall constitute one and the same instrument. Delivery of any signature page by facsimile, electronic or .pdf transmission shall be binding to the same extent as an original signature page.
- (d) <u>Entire Agreement</u>. This Agreement (including the Exhibits and Schedules hereto) and other documents contemplated hereby constitute the entire agreement between the parties with respect to the subject matter hereof and supersede all prior agreements and understandings between the parties with respect to such subject matter.
- (e) Governing Law; Jurisdiction; Service of Process. This Agreement, and all claims, disputes, controversies or causes of action (whether in contract, tort, equity or otherwise) that may be based upon, arise out of or relate to this Agreement (including any schedule or exhibit hereto) or the negotiation, execution or performance of this Agreement (including any claim, dispute, controversy or cause of action based upon, arising out of or related to any representation or warranty made in or in connection with this Agreement or as an inducement to enter into this Agreement), shall be governed by and construed in accordance with the Laws of the State of Delaware, without regard to any choice or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the Laws of any jurisdiction other than the State of Delaware. Each of the parties agrees that any Action related to this agreement shall be brought exclusively in the Court of Chancery of the State of Delaware or, if under applicable Law, exclusive jurisdiction over such matter is vested in the federal courts.

any federal court in the State of Delaware and any appellate court from any thereof (the "Chosen Courts"). By executing and delivering this Agreement, each of the parties irrevocably: (i) accepts generally and unconditionally submits to the exclusive jurisdiction of the Chosen Courts for any Action relating to this Agreement; (ii) waives any objections which such party may now or hereafter have to the laying of venue of any such Action contemplated by this Section 8(e) and hereby further irrevocably waives and agrees not to plead or claim that any such Action has been brought in an inconvenient forum; (iii) agrees that it will not attempt to deny or defeat the personal jurisdiction of the Chosen Courts by motion or other request for leave from any such court; (iv) agrees that it will not bring any Action contemplated by this Section 8(e) in any court other than the Chosen Courts; (v) agrees that service of all process, including the summons and complaint, in any Action may be made by registered or certified mail, return receipt requested, to such party at their respective addresses provided in accordance with Section 8(a) or in any other manner permitted by Law; and (vi) agrees that service as provided in the preceding clause (v) is sufficient to confer personal jurisdiction over such party in the Action, and otherwise constitutes effective and binding service in every respect. Each of the parties hereto agrees that a final judgment in any Action in a Chosen Court as provided above may be enforced in other jurisdictions by suit on the judgment or in any other manner provided by Law, and each party further agrees to the non-exclusive jurisdiction of the Chosen Courts for the enforcement or execution of any such judgment.

- (f) Waiver of Jury Trial. THE PARTIES HEREBY UNCONDITIONALLY AND IRREVOCABLY WAIVE THEIR RIGHT TO TRIAL BY JURY IN ANY JUDICIAL PROCEEDING IN ANY COURT RELATING TO ANY DISPUTE, CONTROVERSY OR CLAIM ARISING OUT OF, RELATING TO OR IN CONNECTION WITH THIS AGREEMENT OR ANY DOCUMENT CONTEMPLATED HEREBY (INCLUDING ANY SCHEDULE OR EXHIBIT HERETO AND THERETO) OR THE BREACH, TERMINATION OR VALIDITY OF SUCH AGREEMENTS OR THE NEGOTIATION, EXECUTION OR PERFORMANCE OF SUCH AGREEMENTS. NO PARTY TO THIS AGREEMENT SHALL SEEK A JURY TRIAL IN ANY LAWSUIT, PROCEEDING, COUNTERCLAIM OR ANY OTHER LITIGATION PROCEDURE BASED UPON, OR ARISING OUT OF, THIS AGREEMENT OR ANY RELATED INSTRUMENTS. NO PARTY WILL SEEK TO CONSOLIDATE ANY SUCH ACTION IN WHICH A JURY TRIAL HAS BEEN WAIVED WITH ANY OTHER ACTION IN WHICH A JURY TRIAL CANNOT BE OR HAS NOT BEEN WAIVED. EACH PARTY TO THIS AGREEMENT CERTIFIES THAT IT HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT OR INSTRUMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS SET FORTH ABOVE IN THIS SECTION 8(f). WILL NOT BE FULLY ENFORCED IN ALL INSTANCES.
- (g) Assignment. This Agreement shall be binding upon, and shall be enforceable by and inure solely to the benefit of, the parties hereto and their respective successors and permitted assigns; provided, that neither this Agreement nor any of the rights, interests or obligations under this Agreement shall be assigned, in whole or in part, by operation of Law or otherwise by any of the parties hereto without the prior written consent of the other parties, except that Parent and Merger Sub may assign, in their sole discretion and without the consent of any other party, any or all of their rights hereunder to each other or to one or more direct or indirect wholly owned Subsidiaries of Parent. Any assignment in violation of the preceding sentence shall be void. Subject to the preceding two sentences, this Agreement will be binding upon, inure to the benefit of, and be enforceable by, the parties and their respective successors and assigns.
- (h) <u>Severability of Provisions</u>. If any provision of this Agreement or any document contemplated hereby, or the application of any such provision to any Person or circumstance, shall be held invalid, illegal or unenforceable in any respect by a court of competent jurisdiction,

such invalidity, illegality or unenforceability shall not affect any other provision hereof. The parties further agree that if any provision contained herein is, to any extent, held invalid or unenforceable in any respect under the Laws governing this Agreement, they shall take any actions necessary to render the remaining provisions of this Agreement valid and enforceable to the fullest extent permitted by Law and, to the extent necessary, shall amend or otherwise modify this Agreement to replace any provision contained herein that is held invalid or unenforceable with a valid and enforceable provision giving effect to the intent of the parties.

- Remedies; Specific Performance. Except as otherwise expressly provided herein, any and all remedies provided herein will be deemed cumulative with and not exclusive of any other remedy conferred hereby, or by law or equity upon such party, and the exercise by a party hereto of any one remedy will not preclude the exercise of any other remedy. The parties hereby agree that irreparable damage would occur in the event that any provision of this Agreement was not performed in accordance with its specific terms or were otherwise breached, and that money damages or other legal remedies would not be an adequate remedy for any such damages. Except as otherwise expressly provided herein, the parties agree that: (i) in the event of any breach or threatened breach by any party of any covenant, obligation or other provision set forth in this Agreement, the other parties shall be entitled (in addition to any other remedy that may be available to it) to (A) a decree or order of specific performance or mandamus to enforce the observance and performance of such covenant, obligation or other provision, and (B) an injunction restraining such breach or threatened breach; and (ii) no party shall be required to provide any bond or other security in connection with any such decree, order or injunction or in connection with any related action or proceeding. . Each party agrees not to raise any objections to the availability of the equitable remedy of specific performance to prevent or restrain breaches or threatened breaches of this Agreement by such party, and to specifically enforce the terms and provisions of this Agreement to prevent breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of such party under this Agreement. Any party's pursuit of any injunction or specific performance at any time will not be deemed an election of remedies or waiver of the right to pursue any other right or remedy to which such party may be entitled. The parties hereto have specifically bargained for the right to specific performance of the obligations hereunder, in accordance with the terms and conditions of this Section 8(i).
- (j) <u>Amendment</u>. Any amendments, or alternative or supplementary provisions, to this Agreement must be made in writing and duly executed by each of the parties hereto in order to be effective against such party.
- (k) <u>No Presumption</u>. This Agreement shall be construed as if drafted jointly by the parties, and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provisions of this Agreement.

* * *

SECTION 9. [Signature page follows]

SECTION 10.

IN WITNESS WHEREOF, Pare delivered as of the date first written abo	nt, Merger Sub and the following Equityholders have caused this Agreement to be duly executed and ve.
	FLYEXCLUSIVE, INC.
	By: Name: Thomas James Segrave, Jr. Title: Chief Executive Officer and Chairman
	FLYX MERGER SUB, INC.
	By: Name: Thomas James Segrave, Jr. Title: Chief Executive Officer and Chairman

MICHAEL D.	WINSTON
------------	---------

Address:			
_			
Email:			

EHUD TALMOR			
Address:	-		
Email:			

WRENDON TIMOTHY
Address:
Email:

WILLIAM YANKUS			
Address:			
Email:			

LT. COL. RAN DAVID

Address	3:		
			-
Email:			_

DONALD JEFFREY WOODS

Address	:		
Email:			_

GEORGE MURN	ANE
Address:	
Email:	_
[Signature Page	to Support Agreement]

SCHEDULE I

Name	Common Stock	Preferred Stock	Options
Michael D. Winston	29,740	-	740
George Murnane	-	-	5,682
Ehud Talmor	97	-	640
Wrendon Timothy	1,206	-	-
William Yankus	97	-	-
Lt. Col. Ran David	97	-	777
Donald Jeffrey Woods	97	-	-

EXHIBIT A -FORM OF AGREEMENT AND PLAN OF MERGER

[Attached]

EXHIBIT B -SPOUSAL CONSENT

I, spouse of, have read and approve the Support Agreement (the "Agreem	, having the legal capacity, power and authority to do so, hereby confirm that I nent") to which this Spousal Consent is attached. I understand the contents of the
Agreement. I am aware that my spouse is a party to the Agree	ement and the Agreement contains provisions regarding the voting and transfer of spouse may own, including any interest I might have therein In consideration of
the terms and conditions as set forth in the Agreement, I her rights and obligations under the Agreement, and agree to be	reby appoint my spouse as my attorney in fact with respect to the exercise of any e bound by the provisions of the Agreement insofar as I may have any rights or y laws of the State of Nevada or similar laws relating to marital or community
Date:	
Signature of Spouse:	
Printed Name of Spouse:	

flyExclusive Reaches Definitive Agreement to Acquire Jet.Al Aviation Business in an All-Stock Transaction

Proposed transaction provides flyExclusive with additional growth capital and will add trading liquidity through incremental equity ownership

Business combination transforms Jet.Al into a pure-play Al solutions company

KINSTON, NC and LAS VEGAS, NV — flyExclusive Inc. (NYSE American: FLYX), a publicly traded provider of premium jet charter experiences, has entered a definitive agreement to acquire the aviation business of Jet.Al Inc. (NASDAQ: JTAI), an innovative private aviation and artificial intelligence company.

flyExclusive will acquire the business in conjunction with Jet.Al's focus on being a pure-play AI solutions company. The transaction will occur following Jet.Al's spin of the business into a new company ("SpinCo"), which would then be acquired by flyExclusive (collectively, the "Business Combination"). Jet.Al shareholders will retain their Jet.Al stock and receive new Class A common shares in flyExclusive as part of the transaction. The Business Combination is contingent upon the closing conditions outlined in the agreement.

This strategic Business Combination is a natural fit for flyExclusive to acquire Jet.Al's aviation business, as both companies operate planes from Textron Aviation and HondaJet and the acquired assets will help support flyExclusive's 2025 growth plans. As the one of the largest and most well-established companies in private aviation, flyExclusive brings a robust fleet and leading operational expertise. This deal is mutually beneficial as it is expected to provide flyExclusive with additional growth capital and enhanced shareholder liquidity and to allow Jet.Al to focus as a pure-play Al solutions company.

flyExclusive's Founder and Chief Executive Officer, Jim Segrave noted "The proposed transaction with Jet.AI is the latest example of the value flyExclusive's vertically integrated private aviation platform provides to industry participants. The transaction benefits flyExclusive investors and will augment our continued growth and market share expansion as an industry leader. Additional growth capital and new shareholders provide an early tailwind in 2025 as we onboard additional high-performance aircraft and serve strong demand for our differentiated customer service.'

"This Business Combination with flyExclusive offers our shareholders the opportunity to benefit from growth in both private aviation and AI," said Jet.AI Founder and Executive Chairman Mike Winston. "flyExclusive, the fifth-largest operator in the industry by hours flown, is a natural fit, with clear synergies given the common aircraft we operate."

Transaction Details

The Business Combination is structured as an all-stock transaction, spinning off Jet.Al's charter business assets into the SpinCo, which in turn will be acquired by flyExclusive. Key details of the transaction include:

• Ownership: The Business Combination consideration will be in the form of a number of fully paid and non-assessable shares of Class A common stock of flyExclusive equal to the quotient of

the purchase price, divided by the average volume-weighted average price of flyExclusive's stock for the thirty trading days immediately prior to the effective date. Jet.Al shareholders will retain their Jet.Al stock while receiving new Class A common shares in flyExclusive to account for the portion of the aviation business being spun-off.

- Purchase Price: The purchase price shall be determined based on Jet.Al's Net Cash multiplied by the Applicable Premium Percentage. Net Cash of at least \$12 million is a condition to closing the Business Combination and will consist of the Company's cash on hand (including any deposits held by Textron Aviation) subtracting any cash net working capital requirement mutually agreed upon, and transaction costs triggered by the closing of the Business Combination. To satisfy closing conditions and to meet the company's ongoing financing requirements, Jet.Al has signed a \$50 million non-binding term sheet with Hexstone Capital LP on economic terms substantially similar to those of its existing \$16.5 million arrangement with Ionic Ventures LLC. The applicable premium percentage paid for the aviation business will be between 115% and 120% equivalent to a dollar value today contemplated in the range of \$12mm to \$22mm, depending on the Net Cash value at the close of transaction.
- **Timing**: The transaction is expected to close in the second quarter of 2025, subject to various closing conditions, including but not limited to Jet.Al financing, regulatory review, and shareholder approval.

About flyExclusive

flyExclusive is a vertically integrated, FAA-certificated air carrier providing private jet experiences by offering customers a choice of ondemand charter, Jet Club, and fractional ownership services to destinations across the globe. flyExclusive has one of the world's largest fleets of Cessna Citation aircraft, and it operates a combined total of approximately 100 jets, ranging from light to large cabin sizes. The company manages all aspects of the customer experience, ensuring that every flight is on a modern, comfortable, and safe aircraft. flyExclusive's in-house repair station, aircraft paint, cabin interior renovation, and avionics installation capabilities, are all provided from its campus headquarters in Kinston, North Carolina. To learn more, visit www.flyexclusive.com.

About Jet.Al

Jet.Al currently operates in two segments, Software and Aviation, respectively. The Software segment features the B2C CharterGPT app, the Ava agentic booking AI, and the B2B Jet.AI Operator platform. The CharterGPT app and Ava both use natural language processing and machine learning to improve the private jet booking experience. The Jet.AI operator platform offers a suite of stand-alone software products to enable FAA Part 135 charter providers to add revenue, maximize efficiency, and reduce environmental impact. The Aviation segment features jet aircraft fractions, jet cards, on-fleet charter, management, and buyer's brokerage. Jet.AI is an official partner of the Las Vegas Golden Knights, 2023 NHL Stanley Cup® champions. The Company was founded in 2018 and is based in Las Vegas, NV and San Francisco, CA.

Additional Information and Where to Find It

In connection with the Business Combination, flyExclusive and Jet.Al intend to file relevant materials with the SEC, including a registration statement on Form S-4, which will include a proxy statement/prospectus. After the registration statement is declared effective by the SEC, the definitive proxy statement/prospectus and other relevant documents will be mailed to the shareholders of Jet.Al as of

the record date established for voting on the Business Combination and will contain important information about the Business Combination and related matters. Shareholders of Jet.Al and other interested persons are advised to read, when available, these materials (including any amendments or supplements thereto) and any other relevant documents in connection with Jet.Al's solicitation of proxies for the meeting of shareholders to be held to approve, among other things, the proposed Business Combination because they will contain important information about Jet.Al, flyExclusive and the Business Combination. Shareholders will also be able to obtain copies of the preliminary proxy statement/prospectus, the definitive proxy statement/prospectus and other relevant materials in connection with the transaction without charge, once available, at the SEC's website at www.sec.gov or by directing a request to: Jet.Al Inc., 10845 Griffith Peak Drive, Suite 200, Las Vegas, NV 89135, Attention: John Yi, email: Jet.Al@gateway-grp.com or Telephone: (949) 574-3860.

Participants in the Solicitation

Jet.Al and its respective directors and executive officers may be deemed participants in the solicitation of proxies from Jet.Al's shareholders in connection with the Business Combination. Jet.Al's shareholders and other interested persons may obtain, without charge, more detailed information regarding the directors and officers of Jet.Al as reflected in the annual report on Form 10-K for the period ended December 31, 2023, filed with the SEC on April 1, 2024, as amended by Form 10-K/A filed on April 29, 2024. Information regarding the persons who may, under SEC rules, be deemed participants in the solicitation of proxies to Jet.Al's shareholders in connection with the Business Combination will be set forth in the proxy statement/prospectus for the Business Combination when available. Additional information regarding the interests of participants in the solicitation of proxies in connection with the Business Combination will be included in the proxy statement/prospectus that flyExclusive and Jet.Al intend to file with the SEC. You may obtain free copies of these documents as described in the preceding paragraph.

flyExclusive and its directors and executive officers may also be deemed to be participants in the solicitation of proxies from the shareholders of Jet.Al in connection with the Business Combination. A list of the names of such directors and executive officers and information regarding their interests in the Business Combination will be included in the proxy statement/prospectus for the Business Combination when available.

No Solicitation or Offer

This communication is for informational purposes only and is neither an offer to purchase, nor a solicitation of an offer to sell, subscribe for or buy any securities or the solicitation of any vote in any jurisdiction pursuant to the Business Combination or otherwise, nor shall there be any sale, issuance or transfer of securities in any jurisdiction in contravention of applicable law. No offer of securities shall be made except by means of a prospectus meeting the requirements of Section 10 of the Securities Act of 1933, as amended, and otherwise in accordance with applicable law.

Forward-Looking Statements

This press release contains certain statements that may be deemed to be "forward-looking statements" within the meaning of the federal securities laws, including the safe harbor provisions under the Private Securities Litigation Reform Act of 1995, including with respect to Business Combination and its potential benefits. Statements that are not historical are forward-looking statements within the meaning of Section 27A of the Securities Act of 1933 and Section 21E of the Securities Exchange Act of 1934.

Forward-looking statements relate to future events or our future performance or future financial condition. These forward-looking statements are not historical facts, but rather are based on current expectations, estimates and projections about our companies, our industry, our beliefs and our assumptions. These forward-looking statements generally are identified by the words "believe," "project," "expect," "anticipate," "estimate," "intend," "strategy," "future," "opportunity," "plan," "may," "should," "will," "would," "will be," "will continue," "will likely result," and similar expressions or the negative of these terms or other similar expressions, but the absence of these words does not mean that a statement is not forward-looking. Forward-looking statements are predictions, projections and other statements about future events that are based on current expectations and assumptions and, as a result, are subject to risks and uncertainties that could cause the actual results to differ materially from the expected results. As a result, caution must be exercised in relying on forward-looking statements, which speak only as of the date they were made. Factors that could cause actual results to differ materially from those expressed or implied in forward-looking statements include the risk that the Business Combination might not be completed in a timely manner or at all, which could adversely affect the price of Jet.Al's common stock or flyExclusive's securities; the failure to satisfy the conditions to the consummation of the Business Combination, including required Jet.Al financing, board and shareholder approvals; potential dilution to Jet.Al shareholders from any financing; the occurrence of any event, change or other circumstance that could give rise to the termination of the definitive agreement for the Business Combination; the effect of the announcement or pendency of the transaction on Jet.Al's or flyExclusive's business generally; risks that the Business Combination disrupts current plans or operations of Jet.Al or flyExclusive; the outcome of any legal proceedings that may be instituted against Jet.Al or flyExclusive related to the Business Combination; the ability to realize any of the benefits anticipated in the Business Combination; risks relating to agreements with third parties; the companies' ability to raise funding in the future, as needed, and the terms of such funding, including potential dilution caused thereby; the companies' ability to maintain the listing of its securities on a national securities exchange; and those risks that can be found in the companies' most recent Annual Report on Form 10-K and subsequent filings with the Securities and Exchange Commission. These filings identify and address other important risks and uncertainties that could cause actual events and results to differ materially from those contained in the forward-looking statements. Readers are cautioned not to put undue reliance on forward-looking statements, and Jet.Al assumes no obligation and does not intend to update or revise these forward-looking statements, whether because of new information, future events, or otherwise, except as provided by law.

Contacts:

flyExclusive

Media: Jillian Wilson, Marketing Specialist
media@flyexclusive.com

Investor Relations: Sloan Bohlen, Solebury Strategic Communications

investors@flyexclusive.com

Jet.Al Inc. Gateway Group, Inc. 949-574-3860 Jet.Al@gateway-grp.com